TG 1164 JULY 1971 Copy No. 8

AD 731662



Technical Memorandum

THE MOTION OF BALLISTIC MISSILES

by L. S. GLOVER and J. C. HAGAN

THE JOHNS HOPKINS UNIVERSITY . APPLIED PHYSICS LABORATORY

Approved for public release; distribution unlimited.

the state of the second st

And the second sec



. n. n

"MARTINE



THIS DOCUMENT IS BEST QUALITY AVAILABLE. THE COPY FURNISHED TO DTIC CONTAINED A SIGNIFICANT NUMBER OF PAGES WHICH DO NOT REPRODUCE LEGIBLY.

> REPRODUCED FROM BEST AVAILABLE COPY

UNCLASSIFIED			
DOCUMEN	NT CONTROL DATA - R	& D	· · · · · · · · · · · · · · · · · · ·
Security classification of title, hady of abstract an Original ting ACTIVITY (Corporate author)	d indexing annotation must be	atered when the overall report is class 20. HEPORT SECURITY CLASSIEG	AliQN
The Johns Hopkins University Appli	ed Physics Lab.	Unclassified	
8621 Georgia Ave. Silven Spring Md 20010		NA	
NEPORT STUE			
The Motion of Ballistic Missiles			
DESCRIPTIVE NOTES (Type of report and inclusive dates)		
Technical Memorandum			· · · · · · · · · · · · · · · · · · ·
L. S. Glover and J. C. Hagen			
REPORT DATE	74. TOTAL NO. C	FPAGES 76, NO. OF REFS	
July 1971	390	20	<u> </u>
$\mathbf{N} = \mathbf{O} \mathbf{O} \mathbf{O} \mathbf{O} \mathbf{O} \mathbf{O} \mathbf{O} \mathbf{O}$		- SERVA) AVMBER(8)	
1,0001 /=02~C=0004	TG 116	1	
e.	S. OTHER HEPO	RT NO(5) (Any other numbers that may	be assigned
	this teport)	i i i i i i i i i i i i i i i i i i i	
<i>d</i> .		· · · · · · · · · · · · · · · · · · ·	
1. SUPPLEMENTARY NOTES	12. SPONSORING Strategic	Systems Project Offic	e
11. SUPPLEMENTARY NOTES 3. ABSTRACT This bandbook on the m	Strategic	Systems Project Offic	e
This handbook on the m primarily for use in training new pe evaluating the performance of reent characteristics of the atmosphere a of the effects on ballistic missile m sphere, and initial body rates. The types of mass and/or aerodynamic solutions are given for the impact of nonstandard meteorological charact jectory simulations and flight test a amples illustrate the material.	12. SPONSORING Strategic Strategic notion of ballistic i ersonnel charged y try bodies. Aeroc are reviewed brief notion of gravity, o e anomalous motic asymmetries is di dispersion resultin teristics at the im analyses are also o	Systems Project Offic nissiles has been prepa with the responsibility of ynamic fundamentals a ly, followed by discuss earth rotation, the atmo n resulting from various scussed in detail. Sim g from asymmetries an pact location. Various liscussed. Numerical	e ared of nd ions o- 2s aple nd tra- ex-
This handbook on the m primarily for use in training new pe evaluating the performance of reent characteristics of the atmosphere a of the effects on ballistic missile m sphere, and initial body rates. The types of mass and/or aerodynamic solutions are given for the impact of nonstandard meteorological charact jectory simulations and flight test a amples illustrate the material.	12. SPONSORING Strategic notion of ballistic n ersonnel charged of try bodies. Aeroc are reviewed brief notion of gravity, of e anomalous motic asymmetries is di dispersion resultin teristics at the im analyses are also of	Systems Project Offic nissiles has been prepa with the responsibility of ynamic fundamentals a ly, followed by discuss earth rotation, the atmo- n resulting from various scussed in detail. Sim g from asymmetries an pact location. Various liscussed. Numerical	e ared of nd ions o- is uple nd tra- ex-
This handbook on the m primarily for use in training new pe evaluating the performance of reent characteristics of the atmosphere a of the effects on ballistic missile m sphere, and initial body rates. The types of mass and/or aerodynamic solutions are given for the impact of nonstandard meteorological charact jectory simulations and flight test a amples illustrate the material.	12. SPONSORING Strategic notion of ballistic r ersonnel charged try bodies. Aeroc are reviewed brief notion of gravity, o e anomalous motic asymmetries is di dispersion resultin teristics at the im analyses are also o	Systems Project Offic missiles has been prepa- vith the responsibility of ynamic fundamentals a ly, followed by discuss earth rotation, the atmo- n resulting from various scussed in detail. Sim g from asymmetries an pact location. Various liscussed. Numerical	e ared of nd ions o- is uple nd tra- ex-

なない。同時には

1999 - Y

A CONTRACTOR OF A CONTRACTOR A

....

1

Į.

Ş

UNCLASSIFIED

Security Classification

F,

14.

Ballistic missile Ballistic flight test data Trajectory simulation Aerodynamic characteristics Training manual Impact dispersion Missile trajectory Numerical solution

ah : " ...

n ng tag

UNCLASSIFIED Security Classification

1.56

s 📠

Strand Colors

was a gran constrained against start of the start of the

「日本の日本」

「「「「「「「「「「「」」」」」

1

TG 1164 JULY 1971

Technical Memorandum

THE MOTION OF BALLISTIC MISSILES

by L. S. GLOVER and J. C. HAGAN

THE JOHNS HOPKINS UNIVERSITY

APPLIED PHYSICS LABORATORY

8621 Georgia Avenue, Silver Spring, Maryland 20910

Cperating under Contract N00017- 69-C-0604 with the Department of the Navy

Approved for public release; distribution unlimited.

THE JOHNS HOPKINS UNIVERSITY APPLIED PHYSICS LABORATORY SILVER SPRING MARYLAND

ABSTRACT

This handbook on the motion of ballistic missiles has been prepared primarily for use in training new personnel charged with the responsibility of evaluating the performance of reentry bodies. Aerodynamic fundamentals and characteristics of the atmosphere are reviewed briefly, followed by discussions of the effects on ballistic missile motion of gravity, earth rotation, the atmosphere, and initial body rates. The anomalous motion resulting from various types of mass and/or aerodynamic asymmetries is discussed in detail. Simple solutions are given for the impact dispersion resulting from as_ymmetries and nonstandard meteorological characteristics at the impact location. Various trajectory simulations and flight test analyses are also discussed. Numerical examples illustrate the material. THE JOHNS HOPKILLS UNIVERSITY APPLIED PHYSICS LABORATORY SILVER SPRING MARYLAND

たいないのないでもないないないのないで、「ないない」

ABSTRACT

This handbook on the motion of ballistic missiles has been prepared primarily for use in training new personnel charged with the responsibility of evaluating the performance of reentry bodies. Aerodynamic fundamentals and characteristics of the atmosphere are reviewed briefly, followed by discussions of the effects on ballistic missile motion of gravity, earth rotation, the atmosphere, and initial body rates. The anomalous motion resulting from various types of mass and/or aerodynamic asymmetries is discussed in detail. Simple solutions are given for the impact dispersion resulting from asymmetries and nonstandard meteorological characteristics at the impact location. Various trajectory simulations and flight test analyses are also discussed. Numerical examples illustrate the material.

- **.** - - - -

my same allow is port of

THE JOHNS HOPKING UNIVERSITY APPLIED PHYSICS LABORATORY SILVER SPRING MARYLAND

CONTENTS

	List of Illustrations		•	•	vii
	List of Tables .	•	•	•	xi
1	Introduction .	•	•	•	1
2	Aerodynamics .	•	•	•	5
3	Atmosphere .		•	•	43
4	Equations of Motion	•	•	•	71
4.1	Effects of Gravity	•	•	•	75
4.2	Effects of Earth Rota	tion	•	•	95
4.3	Effects of Atmospher	е.	•	•	107
4.4	Effects of Body Dyna	mics	•	•	12 9
4.5	Effects of Asymmetry	ies .		•	157
4.6	Examples .	•	•	•	235
5	Dispersion .	•	•		26 9
6	Trajectory Simulation	n.	•	•	32 3
7	Flight Dynamics Data	Analysis	•	•	333
8	Constants and Conver	sion Equa	tions	•	365
	Index .	•	•		369

ν -

Preceding page blank

: A

and a second state of the second states of the second states of the second states of the second states of the s

114

Carling and a second second

1.000

đ

····///·······

THE JOHNS HOPKINS UNIVERSITY APPLIED PHYSICS LABORATORY BILVER SPRING MARYLAND

ŀ

ILLUSTRATIONS

•-----

2-1	Variation of Viscosity with Temperature		35
2-2	Variation of Viscosity with Temperature	•	36
2-3	Cone Axial Force Coefficient, Lamina: Flow, Cold Wall, $\delta = 10^{\circ}$.	r •	37
2-4	Cone Axial Force Coefficient, Turbule Flow, Hot Wall, $\delta = 10^\circ$.	nt •	38
2-5	Pitch Damping Coefficient for 10° Semiangle Cone	•	3 9
2-6	Roll Damping Coefficient, Laminar Flo Cold Wall, 6 = 10°	ow, •	40
2-7	Roll Damping Coefficient, Turbulent Flow, Hot Wall, $\delta = 10^{\circ}$.	•	41
3-1	Typical Atmospheric Temperature Profile and Altitude Regions .		65
3-2	Standard Atmosphere Characteristics	•	66
3-3	Effect of Water Vapor on Density	•	67
4-1	Range versus γ_0 and V_0 .	•	89
4-2	Initial $\gamma_{\rm o}$ versus V and Range .	•	90
4-3	Flight Time versus γ_0 , R = 3000 nmi	•	91
4-4	Flight Time versus Range, Minimum Energy Trajectory	•	92
4-5	Range Sensitivity to V_{0}	•	93
4-6	Range Sensitivity to γ_0	•	94
4-7	Variation of $I_1(K)$ with K .	•	125
4-8	Variation of I, (K) with K .	•	126

Preceding page blank

.

S. S. M. .

·. ,

高泉保護地 御みおしく

THE JOHNS HOPKING UNIVERSITY APPLIED PHYSICS LABORATORY SILVER SPRING MARYLAND

H. Atte

ILLUSTRATIONS (cont'd)

4-9	Variation of $I_2(K)$ with K .		127
4-10	g_X/g_X versus K	•	1 28
4-11	Body Axis System	•	155
4-12	Variation of $J_{O}(X)$ with X .		156
4-13	Variation of $ ilde{A}$ and $\Delta\psi$ with λ .	•	227
4-14	Variation of $\lambda_{\rm p}$ with D cot $\emptyset_{\rm p}$.		228
4-15	Variation of G with λ and ϕ_0 , D = 0.2	•	229
4-16	Variation of $F(\lambda, D)$ with λ , $D = 0.2$	•	230
4-17	Roll Lock-In Criteria, Body A .	•	231
4-18	Roll Lock-In Criteria, Body B .	•	232
4-19	Roll Lock-In Criteria, Body C .	•	233
4-20	$\overline{\pmb{\alpha}}$ Decay Characteristics .	•	25 9
4-21	Significant Events	•	260
4-22	$\overline{\alpha}_{0}$ Required for $g_{1} = 20$.	•	261
4-23	p_r and ω_A versus Altitude .	•	262
4-24	Conditions for Roll Resonance at $h = 89 000$ Feet	•	263
4-25	$\overline{\boldsymbol{\alpha}}$ Required for Lock-In and Spin through Zero Roll Rate		264
4-26	Boundaries of $\overline{\alpha}_0$ and $\overline{\alpha}_0$ for Lock-In at First Resonance		265
4-27	Variation of r _{min} with Initial Roll Rate	•	2 66
5-1	Limiting Values of K	•	30 9
5 -2	Limiting Values of $\gamma_{\rm E}$	•	310
5-3	Variations of E with h , 1962 Standard Atmosphere	•	311

ił Ŋ

192

5

and the meaning of a

- viii -

THE JOHNE HOPKINE UNIVERSITY APPLIED PHYSICS LABORATORY SILVER SPRING MARYLAND

Ę i

.į.

ILLUSTRATIONS (cont'd)

.

5-4	Dispersion Resulting from Wind and Deviation in Density	•	312
5-5	Dispersion Resulting from Meteorologi Characteristics at High Altitude .	cal •	313
5 - 6	Probability of Encountering Zero Roll Rate	•	314
5-7	Maximum Altitude for Zero Roll Rate		315
5-8	Maximum Dispersion Resulting from $\Delta \theta$ and r/d	•	316
5-9	Dispersion versus $\phi_{\rm constant}$	•	317
5-10	Dispersion Probability	•	318
7-1	Bit Presentation Scheme, DITAP Flight Records	•	35 3
7-2	Timing Traces, DITAP Flight Records	•	354
7-3	Bit Presentation Scheme, DITAR Fligh Records	t.	355
7-4	Timing Traces, DITAR Flight Records	•	356
7-5	Sample Roll Rate History .	•	357
7-6	Sample Lateral Acceleration History	•	358
7-7	Sample Longitudinal Acceleration History	•	359
7-8	Sample Pitch Rate History .	•	360
7-9	Angle of Attack Work Sheet .	•	361
7-10	Aerodynamic Pitch Frequency (Weathe cock Frequency) Work Sheet	r- ,	362
7-11	Amplification Factor Work Sheet		363
7-12	Rotation of Maneuver Plane Work Sheet	•	364

- ix -

<u>.</u>

THE JOHNS MOPKINS UNIVERSITY APPLIED PHYSICS LABORATORY BILVER SPRING MARYLANS

•

TABLES

2-1	Mass and Aerody	namie C	Characte	eristics	•	42
3-1	1962 Standard Atr	mospher	re	•	•	68
3 -2	Constants for the Atmosphere	196 2 S t	andard	•	•	69
4-1	Reentry Characte	ristics		,	•	267
5-1	Values of I ₃	•	•	•	•	319
5-2	Values of F	•	•	•	•	320
5-3	Values of I ₅	•	•	•	•	321
5-4	Values of F	•1	•	•	•	322

Preceding paga blank

- xi -

THE JOHNS HOPKINS UNIVERSITY APPLIED PHYSICS LABORATORY BILVER BRING, MARVLAND

l

I

I

I

۸,

٠.

1. INTRODUCTION

- 1 -

All and the second

 The term ballistic missile as used herein describes any body in motion that is unpowered and uncontrolled. The motion may occur in either the atmosphere or the exoatmosphere. However, most of the interesting motion phenomena occur within the atmosphere, and most of the discussions concern reentry-type vehicles entering the atmosphere at speeds from 10 000 to 25 000 ft/sec.

The primary purpose of this report is to present material suitable for training new personnel charged with the responsibility of evaluating the performance of reentry bodies; therefore, the subject matter is treated as simply as possible while retaining enough detail to explain the missile behavior. Although the material is written primarily for personnel unfamiliar with the details of missile motion, it provides experienced personnel with readily available equations suitable for estimating order-of-magnitude effects of phenomena frequently encountered in flight test analysis, performance, and preliminary design work.

ţ

Ý.

A brief discussion of the aerodynamic characteristics of missiles and the nature of the atmosphere is followed by a discussion of vehicle motion as affected by gravity, earth rotation, the atmosphere, and initial body rates. Also included is a detailed discussion of the anomalous motion resulting from various types of mass and aerodynamic asymmetries. Simple solutions are given for the impact dispersion resulting from asymmetries and nonstandard meteorological characteristics at the impact location. Simulations required for various types of studies and flight test analysis complete the subjects discussed. Numerical examples illustrate the material presented.

In some sections (particularly parts of Section 4) the material is new and is given in considerable detail, Since some readers may not be interested in details, a summary of the material covered and the major conclusions derived is given at the beginning of each section (each subsection for Section 4).

- 3 -

Preceding page blank

An attempt was made to use symbols and units of measure currently used by specialists in the various fields covered in this report. As a result, a particular symbol may have different definitions in various sections. A complete list of symbols is given at the beginning of each section (each subsection for Section 4), and conversion factors for units of measure are given in Section 8. THE JOHNE HOPKING UNIVERSITY APPLIED PHYSICS LABORATORY SILVER SPRING. MARYLAND

u.

2. AERODYNAMICS

ł

i.

İ

i

۰.

۰.

110411-00

TRAFFIC AND

 2 AERODYNAMICS

10.2.7

1111

The second s

-

ļ

1.

ł

١

.

ļ

i

SYMBOLS FOR SECTION 2

Symbol	Definition	Typical Units
4	axial force	pounds
C 🛓	axial force coefficient	-
с _{Ав}	base drag coefficient	-
с _А _F	friction drag coefficient	-
°A _P	pressure drag coefficient	-
c.g.	refers to center of gravity	-
°, p	rolling moment damping coefficient	-
C	pitching moment coefficient	-
c _m	pitching moment coefficient at $\overline{\alpha}$ = 0	-
C _m	pitching moment damping coefficient	-
C _m	$dC_m/d\overline{\alpha} at \overline{\alpha} = 0$	-
C _N	normal force coefficient	-
C _N	$dC_N/d\overline{\alpha}$ at $\overline{\alpha} = 0$	-
c. p.	refers to center of pressure	-
D	total drag	pounds
D _F	drag resulting from skin friction	pounds
d	reference diameter	feet
к	Knudsen number (see Eq. (2-7))	-
L	lift or	pounds
	body length	feet

i

ALC: UNDA

Preceding page blank

- 7 -

Symbol	Definition	Typical Units
L(p)	rolling moment resulting from roll rate (roll damping moment)	lb-ft
¹ 1	moment arm for asymmetric force	feet
M	Mach number = $V N_{\gamma} RT$ or	-
	pitching moment	lb-ft
M(q)	pitching moment resulting from pitch rate (pitch damping moment)	lb-ft
N	normal force	pounds
N _{asy}	normal force resulting from an asymmetry	pounds
Ng	normal force for a symmetrical vehicle	pounds
Р	ambient pressure	$1b/ft^2$
P	cone surface pressure	$1b/ft^2$
P _F	flap surface pressure	$1b/ft^2$
- q	roll rate	rad/sec
q	pitch rate	rad/sec
q	dynamic pressure, ($\gamma/2$)PM 2 or (1/2) $ ho V^2$	lb/ft ²
R	gas constant = 1716	$ft^2/sec^2 - R$
R _e	Reynolds number, $\rho VL/\mu = VL/\nu$	-
S	reference area	${\rm ft}^2$
S. M.	static margin	-
s _w	surface area excluding the base area	${\rm ft}^2$
т	absolute ambient air temperature	°R
т	reference temperature for viscosity	°R
v	vehicle velocity	ft/sec

- 8 -

1

S. M. Barrette

THE JOHNS HOPKINS UNIVERSITY APPLIED PHYSICS LABORATORY BILVER SPRING MARYLAND

1

1.

.

.

Symbol	Definition	Typical Units
x	distance from nose to a point on the body centerline	feet
X _{c. g.}	distance from vehicle nose to the ${ m c.g.}$	feet
X _{c, p}	distance from vehicle nose to the c.p.	feet
ΔX	X _{c. p.} - X _{c. g.}	feet
Y	distance normal to the body surface	feet
ā	vehicle total angle of attack	radians
α,	local angle of attack	radians
Ŷ	ratio of specific heats - 1,4	-
8	cone half-angle	radians
¢	asymmetry deflection angle	radians
ρ	ambient air density	$slug/ft^3$
μ	dynamic viscosity	slug/ft-sec
μ	dynamic viscosity at T	slug/ft-sec
ν	kinematic viscosity	ft ² /sec
τ	shear stress	$1b/ft^2$
∆ θ	body bend angle	degrees

٠.

1

a shed down

. . .

.

THE JOHNS HOPKING UNIVERSITY APPLIED PHYSICS LABORATORY SILVER SPRING MARYLAND

SUMMARY AND CONCLUSIONS

The aerodynamic forces of interest for reentry body studies are discussed. The significance of the aerodynamic coefficients is reviewed, and equations are presented for estimating the magnitude of the coefficients for a sharp-nosed conical body. Several types of asymmetric cones are also considered. Numerical examples are given for a 10° semiangle cone having a length of 10 feet.

The major conclusions are:

1. The following aerodynamic forces and moments are important in the analysis of motion of a typical reentry body:

- Force perpendicular to the body centerline (normal force) and resulting moment about the body center of gravity (pitching moment)
- Force parallel to the body centerline (axial force)
- Moments resulting by virtue of body pitch, yaw, or roll rotational rate (pitch, yaw, or rol¹ damping moments)
- Forces and/or moments resulting from body asymmetries.

2. For uncontrolled vehicles, it is important that the normal force act through a point (center of pressure) aft of the center of gravity. In this case the vehicle is said to be statically stable. The ratio of the distance between the center of gravity and the center of pressure to some characteristic body dimension (for example, length) is called the stability margin and is usually expressed as a percentage of the characteristic length.

3. The aerodynamic forces and moments are represented by dimensionless coefficients, thereby facilitating evaluation of the forces and moments over a wide variation in environmental parameters (velocity, altitude, etc.).

- 11 -

Preceding page blank

THE JOHNS HOPKINS UNIVERSITY APPLIED PHYSICS LABORATORY SILVER SPRING, MARYAND

> 4. In general, the coefficients are functions of vehicle angle of attack and the similarity factors, Mach number — an indication of the effect of compressibility of the air, Reynolds number — an indication of the nature of the boundary layer on the vehicle, and the Knudsen number — an indication of the size of the vehicle relative to the mean free path of the ambient air particles.

2.1 Aerodynamic Forces

A body moving through air experiences forces – aerodynamic forces – resulting from the motion. The aerodynamic forces of interest for a reentry vehicle are:

- 1. Axial force,
- 2. Normal force,
- 3. Pitch damping force,
- 4. Roll damping force.

The axial force (for a symmetrical vehicle) acts along the body centerline as shown in the sketch for a cone,



This force is produced primarily by relatively high pressure acting on the conical surface (pressure drag) and relatively low pressure acting on the cone base (base drag). In addition, a force produced by air viscosity (skin friction drag) exists on the conical surface.

A normal force (normal to the body centerline) exists when the body centerline is not directed along the

- 13 -

THE JOHNS HOPKING UNIVERSITY APPLIED PHYSICS LABORATORY SILVER SPRING MARYLAND

> velocity vector. The angle between the velocity vector and the body centerline is called the angle of attack, α . The normal force results from high pressure acting on the windward side of the body compared to low pressure acting on the leeward side. The integration of this pressure difference over the cone surface is the normal force. The pressure differential across the body also produces a moment about the body center of gravity. For convenience in analysis, this aerodynamic moment is computed as the product of the normal force and a moment arm (X_c , $X_{c,g}$, where $X_{c,p}$ is the longitudinal location at which the fotal normal force acts and is called the center of pressure and $X_{c, g}$, is the longitudinal location of the center of gravity. Assume that a vehicle has its center of pressure forward of the center of gravity and that some disturbance causes an initial angle of attack. This angle of attack results in a normal force. Since the normal force is forward of the center of gravity. the aerodynamic moment produced is in a direction to increase the angle of attack. This process continues, and the missile eventually tumbles. Similar reasoning will show that, for a center of pressure located aft of the center of gravity, the aerodynamic moment will always tend to cancel a perturbation in $\overline{\alpha}$. Therefore, a vehicle having a center of pressure forward of the center of gravity is said to be statically unstable; a vehicle having a center of prossure aft of the center of gravity is said to be statically stable. The value of the ratio $(X_{c.p.} - X_{c.g.})/L = \Delta X/L$ is called the static margin. The margin must be positive for uncontrolled vehicles and for high performance vehicles is usually 3% to 5%.

The use of the word "normal" to define the force in the plane of α and perpendicular to the body centerline is used here to be consistent with the usual terminology found in the literature. It is hoped that there will be no confusion with telemetry terminology where "normal" force and "lateral" force are the two components of the normal force as defined above. In an attempt to avoid confusion, in some sections of this report we have used

- 14 -

the term "transverse" force to mean the force in the plane of $\overline{\alpha}$ and perpendicular to the body centerline.

The pitch and roll damping forces exist by virtue of the body pitch and roll rates, respectively, and therefore are called dynamic forces. Consider a body pitching about its center of gravity with rotational velocity, q. The pitching motion produces a local velocity at X equal to $(X - X_{0, g})q$. The distribution of velocity along the body is shown in the sketch.



The local velocity distribution produces a local angle of attack, α_{i} , whose magnitude is given by the equation:

$$\tan \alpha_{\ell} = \frac{(X_{c.g.} - X)q}{V} \approx \alpha_{\ell} \quad (2-1)$$

The local angle of attack results in a local normal force which, at all points along the body, tends to resist the pitching motion. The integral of the local forces (the damping force) is too small to be of significance in ballistic missile analyses. However, the damping moment, which is the integral of the local forces multiplied by their moment arms, is very important in some portions of the trajectory as will be shown later.

- 15 -

The example cited above is just one type of damping effect. Any force that exists by virtue of the pitch rate is a pitch damping force, and these forces do not always resist body pitch motion. The forces that do resist the pitch motion are said to be stable forces (or moments), and those that reinforce the pitch motion are said to be unstable forces (or moments).

The only roll damping force on a symmetrical body of revolution is the force produced by the viscosity of the air (skin friction). This force always tends to retard the roll rate.

Ideally, it would be desirable to calculate the serodynamic forces accurately by theoretical means. However, this is not always possible, and it is necessary to measure the forces in ground tests by blowing air past a fixed body (wind tunnel), or by propelling the body through a stationary instrumented region (ballistic range), or sometimes by using a combination of both techniques. Economically it is not possible to obtain measurements on the actual vehicle for actual reentry conditions of ambient pressure, temperature, velocity, etc. To circumvent this difficulty, the aerodynamic ist defines the forces and moments in terms of aerodynamic coefficients which spply over a wide range of environmental conditions. The following coefficients are of interest in the study of reentry body motion:

$$A = C_A \overline{q} S$$
 (2-2)

$$N = C_{N} \overline{q} S$$
 (2-3)

$$M(q) = C_{m_q} \left(\frac{qd}{V}\right) \overline{q}Sd \qquad (2-4)$$

$$L(p) = C_{\ell_p} \left(\frac{pd}{V}\right) \overline{q}Sd \qquad (2-5)$$

- 16 -

1. 1. 1. 1. 1.

where $\overline{q} = (1/2)\rho V^2 = (\gamma/2)PM^2$ and is called dynamic pressure. The bar over the q is deleted in some sections of this report where there is no possible confusion between the symbols for pitch rate and dynamic pressure. The reference dimensions S and d may be any convenient dimensions. For a cone, they are usually taken as the base area and base diameter, and these reference dimensions will be used in the examples.

From dimensional analysis, it may be shown that for a given $\overline{\alpha}$ the aerodynamic coefficients C_A , C_N (and $X_{c. p.}$), C_{mq} , and $C_{\ell p}$ may be obtained from tests using a geometrically similar model (external shape) provided certain model similarity factors are the same as those for the actual vehicle. The major similarity factors of interest for the reentry body are:

> Mach number, Reynolds number, Knudsen number.

The Mach number is the ratio of the vehicle velocity (V) to the velocity of a sound wave $(\sqrt[V_{V}RT)$. There are four significant Mach number regimes. At subsonic speeds (M \approx 0 to 0, 8), the aerodynamic coefficients are weakly dependent upon M. The transonic Mach number regime extends from the Mach number at which a local Mach number over the body first becomes sonic $(M_f = 1)$ to the Mach number at which the flowfield over the body is predominantly supersonic. This Mach number range extends from about 0, 8 to 1, 5. The local supersonic speed results in the appearance of shock waves. The strength and location of the waves are highly sensitive to changes in Mach number. Therefore, in the transonic speed range, the aerodynamic coefficients are usually sensitive to change in M. In the supersonic speed range $(M \approx 1.5 \text{ to } 6)$, the shock wave strength is dependent upon Mach number but the shock location is somewhat insensitive to changes in Mach number so that the aerodynamic coefficients are moderately dependent upon M. The effect of

- 17 -

aerodynamic heating is moderate in the supersonic speed range. At hypersonic speeds (M > 6), the aerodynamic coefficients tend to be relatively insensitive to M and aerodynamic heating is severe.

The Reynolds number, $\rho VL/\mu = VL/\nu$, is indicative of the importance of the dynamic viscosity of the air, μ (or kinematic viscosity ν). As a result of viscous forces, a region of low velocity is produced in the vicinity of the body. In fact, for continuum flow (discussed below) the velocity of the air relative to the vehicle is zero at the vehicle surface. The region of degraded velocity is called the boundary layer. Since the layer of air adjacent to the body "sticks" to the surface, a shear stress, τ , or an element of friction drag, dD, results, tending to reduce the velocity of the vehicle.



The total friction drag is given by:

$$D_{F} = \int \tau \, dS_{W} = \int \mu \left(\frac{dV}{dY}\right)_{Y=0} \, dS_{W} \quad (2-6)$$

Therefore, the viscous drag is proportional to the dynamic viscosity and the velocity gradient at the wall. The viscosity is very nearly a function of air temperature only

- 18 -

Sec. B. She and sec.

and is given by Sutherland's equation plotted in Figs. 2-1 and 2-2. The velocity gradient is a strong function of the Reynolds number. At very high Reynolds numbers, the flow in the boundary layer contains much eddy or vortical motion. As a result of mixing, the high energy air near the outer portion of the flow is transported in toward the surface. Since the layer of air adjacent to the wall sticks to the surface, this type of boundary layer has a very large velocity gradient at the surface and the skin friction is relatively high. This type of boundary layer is called a turbulent boundary layer. At sufficiently low Reynolds numbers, the eddy motion within the boundary layer disappears, and the velocity gradient at the wall and skin friction become much less than those for the turbulent case. This type of boundary layer is called a laminar boundary layer. At low Reynolds numbers the flow tends to be laminar; at high Reynolds numbers the flow tends to be turbulent. The Reynolds number at which transition from laminar to turbulent flow occurs varies with body shape and type of trajectory. A typical value of R_e at which transition beging to occur on a slender conical body is 10^7 .

The surface heating rate is proportional to the skin friction and therefore is greater for turbulent flow than for laminar flow. In fact, the heating rate in turbulent flow may be nearly an order of magnitude greater than that for laminar flow.

The Knudsen number is the ratio of the mean free path of the ambient air particles to a characteristic dimension of the reentry vehicle. This ratio may be expressed in terms of R_p and M by the relation:

$$K = \frac{1.49 M}{R_e}$$
 (2-7)

For the very low values of density that exist early in the reentry phase, the particles of air are spaced at distances that are large compared to, say, the body length

- 19 -

or diameter. An air particle bouncing off the body is not likely to encounter another air particle. This region of flow is called the free molecular flow regime. At a somewhat lower altitude, the air particles have frequent collisions with other air particles in the vicinity of the reentry body. However, there are insufficient particles in the film of air adjacent to the body surface to cause it to stick to the surface. Therefore, although a boundary layer exists, the film of air adjacent to the wall "slides" along the body surface. This flow regime is called the slip flow regime. At still higher densities, the air particles at the surface stick to the surface. This flow regime is called the continuum flow regime.

The characteristics of the aerodynamic forces and surface heating depend upon the type of flow that exists on the body. There are no sharp boundaries between the various types of flow, but approximate boundaries are as follows:

Continuum to slip
$$\frac{M}{\sqrt{R_e}} \approx 10^{-2}$$

Slip to transition

$$\frac{M}{R_e} \approx 10^{-1}$$

Transition to free molecular $\frac{M}{R_{e}} \approx 10$.

The transition regime considered here is a transition in basic flow regimes which occurs at Knudsen numbers of approximately unity. This transition should not be confused with transition from laminar to turbulent flow in the boundary layer.

The Mach number, boundary layer, and basic fluid flow regimes are shown for a typical reentry body in Fig. 4-21, page 260. Note that practically all the reentry

- 20 -

THE JOHNS HOPKINS UNIVERSITY APPLIED PHYSICS LABORATORY BILVET SPRING MARYLAND

> phenomena for which aerodynamic data are required occur in the continuum flow regime; only this regime will be considered in the remainder of this report.

Sometimes the lift and drag components of the aerodynamic force are required for analysis. The relationship between the lift-drag system of forces and the normal force-axial force system is shown in the sketch.

112 N 10

. (-



From this force diagram, the lift and drag coefficients may be obtained from the following equations:

$$C_L = C_N \cos \overline{\alpha} - C_A \sin \overline{\alpha}$$
 (2-8)

$$C_{D} = C_{A} \cos \overline{\alpha} + C_{N} \sin \overline{\alpha}$$
 (2-9)

Differentiating Eq. (2-8) with respect to $\overline{\alpha}$, and noting that the variation of C_A with $\overline{\alpha}$ is small, the slope of the lift curve at $\overline{\alpha} = 0$ is given by the equation:

$$C_{L_{\alpha}} = C_{N_{\alpha}} - C_{A}$$
(2-10)

- 21 -

2.2 General Characteristics

It is beyond the scope of this report to provide proecdures for estimating the aerodynamic coefficients for a body of arbitrary shape. However, a few remarks on the general behavior of the coefficients are given in this section, and approximate equations applicable to a sharp nosed cone are given in Section 2, 3,

Axial Force. For most bodies of interest, the axial force coefficient is weakly dependent upon α for small values of $\overline{\alpha}$ (1° to 2°), strongly dependent upon M, and moderately dependent upon R_e. For a sharp slender cone, the order of magnitude of C_A is 0.1.

Normal Force. The normal force coefficient is strongly dependent upon $\overline{\alpha}$ and M and nearly independent of R_e. For small values of $\overline{\alpha}$, C_N varies linearly with $\overline{\alpha}$ so that C_N = C_{N α} $\overline{\alpha}$. For a sharp slender cone, C_{N α} ≈ 2 per radian.

<u>Center of Pressure</u>. In general, the center of pressure is strongly dependent upon $\overline{\alpha}$ and M and weakly dependent upon R_c . However, for small values of $\overline{\alpha}$, X_c, p , is independent of $\overline{\alpha}$. For a sharp slender cone, $X_c, p, \approx (2/3)L$.

<u>Pitch Damping</u>. The pitch damping coefficient is strongly dependent upon M, weakly dependent upon α for small values of α , and nearly independent of R_e. For a sharp slender cone, a typical value of C_{m_q} is -1 per

radian. The negative value of $\mathbf{C}_{m_{\mathbf{q}}}$ indicates a stable damping coefficient.

Roll Damping. The roll damping coefficient for a symmetrical vehicle is very small, negative, highly dependent upon R_e and M, and weakly dependent upon $\overline{\alpha}$ for small values of $\overline{\alpha}$. For a cone, a typical value of the roll damping coefficient is -0.003 per radian.

- 22 -

2.3 Approximate Equations for a Sharp Cone

For a major portion of the reentry trajectory, the Mach number is in the hypersonic regime. The aerodynamic coefficients are nearly independent of Mach number, and good approximations to the aerodynamic coefficients for sharp nosed cones are given by relatively simple equations.

The skin friction drag coefficient, C_{AF} , is very sensitive to the cone surface temperature, to the type of boundary layer, and to surface roughness. To provide some idea of the order of magnitude of the coefficient, C_{AF} , without becoming involved in details, equations are

given below which apply to laminar flow with a cold wall (typical of early reentry conditions) and to a turbulent flow with a hot wall (typical of late reentry conditions). A smooth wall was assumed.

An estimate of $C_{\ell p}$ is given in which it is assumed that the shear stress τ is directed along the vector $\vec{v} + \vec{pr}$ where \vec{pr} is the circumferential velocity of a point on the cone surface. The circumferential component of τ is proportional to $C_{\ell p}$.

The following equations are applicable to a cone of half angle δ :

$$C_{N_{\alpha}} = 2 \cos^2 \delta \qquad (2-11)$$

$$\mathbf{x}_{c. p.} = \left[\frac{2}{3} \sec^2 \mathbf{\delta}\right] \mathbf{L}$$
 (2-12)

$$C_{A} = C_{A_{P}} + C_{A_{B}} + C_{A_{F}}$$
(2-13)

$$C_{A_{p}} = 2 \sin^2 \delta$$
 (2-14)

$$C_{A_{B}} = \frac{1}{M^{2}}$$
 (2-15)

- 23 -

THE JOHNS HOPKINS UNIVERSITY APPLIED PHYSICS LABORATORY SILVER SPRING MARYLAND

$$C_{A_{F}} = \frac{1.9}{\tan \delta} \left[\frac{(1 + \gamma M^{2} \sin^{2} \delta) \cos \delta}{R_{e}} \right]^{1/2}$$
(2-16)

. . .

ο.

Sec. Charles in

Laminar flow, cold wall

$$C_{A_{F}} = \frac{0.1}{\sqrt[5]{R_{e}}} \frac{(\cos \delta)^{0.8}}{\tan \delta} \frac{(1 + \gamma M^{2} \sin^{2} \delta)^{0.8}}{(1 + 0.17 M^{2})^{0.7}} \quad (2-17)$$

Turbulent flow, hot wall

$$C_{m_{q}} = \frac{1}{\tan^{2} \delta} \left[\frac{2}{3} \frac{X_{c.g.}}{L} - \frac{1}{4} (1 + \tan^{2} \delta) - \frac{1}{2} \cos^{2} \delta \left(\frac{X_{c.g.}}{L} \right)^{2} \right]$$
(2-18)
$$C_{\ell_{p}} = - \left[\frac{1}{8} \cos^{2} \delta \right] C_{A_{F}}$$
(2-19)

2.4 Asymmetric Bodies

Bodies intended to be symmetrical are always asymmetrical to some extent as a result of manufacturing tolerances and various types of in-flight body distortion. The magnitude of the asymmetric force is usually very small. However, these small forces can have disastrous results on vehicle performance as will be shown later.

Consider a body for which the asymmetric force is produced by a small protuberance located near the aft end of the body, as shown in the sketch.



- 24 -

At $\alpha = 0$, the protuberance results in a downward asymmetric normal force, N_{asy}. This force causes the nose to pitch up, producing an angle of attack. The angle of attack produces a normal force, N_s. The vehicle continues to pitch up until the moment about the center of gravity is zero. The zero moment condition is called the trim condition, and the values N and α that exist at this condition are called the trim normal force and trim angle of attack, respectively.

It is convenient to define another dimensionless aerodynamic coefficient called the pitching moment coefficient, C_m , such that the moment about the center of gravity is given by:

$$M = C_{m} \overline{q} Sd . \qquad (2-20)$$

From the above sketch (using a nose-up moment as positive):

$$M = N_{asy} l_{1} - N_{s} (X_{c, p}, -X_{c, g}) . \qquad (2-21)$$

The total normal force is given by:

$$N = N_{g} - N_{gsv}$$
 (2-22)

Writing Eqs. (2-21) and (2-22) in coefficient form:

$$C_{m} = \frac{N_{asy} l_{1}}{\overline{q} sd} - \frac{N_{s}}{\overline{q}s} \frac{(X_{c.p.} - X_{c.g.})}{d} . \qquad (2-23)$$

$$C_{N} = \frac{N_{B}}{\overline{q}S} - \frac{N_{aBY}}{\overline{q}S} \qquad (2-24)$$

The term, $N_g/\overline{q}S$, is the C_N for a symmetrical vehicle. The N_{asy} term is usually negligible compared to $N_g/\overline{q}S$

- 25 -

THE JOHNS NOPKING UNIVERSITY APPLIED PHYSICS LABORATORY SILVER SPRING MARVLAND

and may be ignored in Eq. (2-24). However, the effect of this term on C_m cannot be ignored. Let

$$\frac{N_{asy}l_1}{\overline{q} sd} = C_{m_o}$$

Then

$$C_{m} = C_{m_{o}} - C_{N} = \frac{(x_{c.p.} - x_{c.g.})}{d}, \qquad (2-25)$$

If the body has a linear variation of C_N with $\overline{\alpha}$, then Eq. (2-25) may be written:

$$C_{m} = C_{m_{o}} - C_{N_{o}} \frac{(x_{c.p.} - x_{c.g.})}{d} \overline{\alpha}$$
, (2-26)

If C_{m_0} and $X_{c, p}$, are independent of $\overline{\alpha}$, then C_m varies linearly with $\overline{\alpha}$.

Within the limitations imposed by the assumptions listed above, typical C_m characteristics for a stable symmetrical and a stable asymmetrical vehicle are shown in the sketch, page 27.

Note that C_{m_0} is the value of C_m at $\overline{\alpha} = 0$, and $\overline{\alpha}_{TR}$ is the value of $\overline{\alpha}$ at $C_m = 0$. The value of $\overline{\alpha}_{TR}$ is obtained from Eq. (2-26) as:

$$\overline{\alpha}_{TR} = \frac{C_{m_o}}{C_{N_{\alpha}} \frac{(X_{c.p.} - X_{c.g.})}{d}} = \frac{C_{m_o}}{C_{N_{\alpha}} \frac{\Delta X}{d}} \cdot (2-27)$$

- 26 -

THE JOHNS HOPKINS UNIVERSITY APPLIED PHYRICS LABORATORY Shver Grins, Maryland



In terms of the static margin defined previously:

$$\overline{\alpha}_{TR} = \frac{C_{m_o}}{(L/d)C_{N_a}S.M.}$$
 (2-28)

In some reports the term $C_{m_{\alpha}}$ is used. An equation for $C_{m_{\alpha}}$ may be obtained by differentiating Eq. (2-26) with respect to $\overline{\alpha}$. Therefore:

$$C_{m_{\alpha}} = -C_{N_{\alpha}} \left(\frac{\Delta X}{d}\right)$$
 (2-29)

To provide a feel for the magnitude of C_{m_0} , consider the flap-type protuberance discussed at the beginning of this section.

- 27 -


The pressure acting on the flap and cone surfaces may be estimated using Newtonian theory which, in the simplest form, states that (for zero angle of attack):

$$\frac{P_{F} - P}{q} = 2 \sin^{2} (\mathbf{0} + \mathbf{\epsilon}) \qquad (2-30)$$

$$\frac{P_c - P}{\overline{q}} = 2 \sin^2 \delta . \qquad (2-31)$$

The force resulting from the asymmetry may be divided into normal and axial force components. For small values of δ and ϵ , the axial component is small compared to the normal component. Neglecting the axial component, the total increment in normal force is:

$$N_{asy} = S_F (P_F - P_c)$$
, (2-32)

Then:

$$C_{m_{o}} = \frac{M}{\overline{q}Sd} = \frac{N_{aby}l_{1}}{\overline{q}Sd} = \frac{S_{F}}{S} \frac{l_{1}}{d} \left[\frac{P_{F}}{\overline{q}} - \frac{P_{c}}{\overline{q}} \right]. \quad (2-33)$$

Rain to the Stranger and the sec

- 28 -

THE JOHNS HOPKINE UNIVERSITY APPLIED PHYSICS LABORATORY SILVER SPRING MARYLAND

Using Eqs. (2-30) and (2-31), and assuming δ and ϵ to be small:

$$C_{m_{o}} = 2 \frac{S_{F}}{S} \frac{\ell_{1} \delta^{2}}{d} \left[2 \frac{\epsilon}{\delta} + \left(\frac{\epsilon}{\delta}\right)^{2} \right] . \qquad (2-34)$$

As an example, let $\delta = 10^{\circ}$, $\ell_1/d = 1$, $S_F/S = 0.1$, and $\epsilon = 2^{\circ}$. For this asymmetry, Eq. (2-34) shows that $C_{m_0} = 0.0027$.

Another type of asymmetry for which C_{m_0} can be

readily evaluated is one in which the body centerline is the arc of a civcle. This type of asymmetry, for example, is an approximation to a distortion resulting from lateral loads or heating effects. For this asymmetry the moment is nearly a pure couple and C_{m_o} may be expressed in

turms of the change in slope of the body centerline from nose to base.



At high hypersonic Mach numbers:

$$C_{m_o} \approx 0.008 \Delta \theta$$
 (2-35)

where $\Delta \theta$ is in degrees.

Contraction of the second

- 29 -

2.5 Examples

A typical reentry body will be used in many examples in later sections of this report. To illustrate the material in the preceding subsections, the aerodynamic data required for subsequent examples will be computed.

The typical body is a 10-foot-long cone with a base diameter of 3.5 feet. These dimensions correspond to a cone semiangle of about 10°. The mass characteristics are assumed to be as follows:

> Weight = 1000 pounds X_{c.g.} = 6.27 feet aft of the nose Pitch moment of inertia about the center of gravity = 300 slug-ft²

Roll moment of inertia = 30 $slug-ft^2$

We will use a reference length d of 3.5 fect and a reference area S of 9.65 ft^2 ,

At hypersonic speeds and for angles of attack close to zero, the aerodynamic coefficients are as follows:

$$X_{c.p.} = \left[\frac{2}{3} \sec^2 10^\circ\right] 10 = 6.87 \text{ feet}$$
 (2-12)

$$C_{N_{\alpha}} = 2 \cos^2 \delta = 1.94 \text{ per radian}$$
 (2-11)

It is observed from Eqs. (2-13) to (2-17) that CA is a function of M, R_e, wall temperature, and type of boundary layer. The values of C_A for two combinations of wall temperature and boundary layer type are shown in Figs. 2-3 and 2-4. The laminar flow, cold wall case is typical of conditions that would exist early in the reentry flight, say above 100 000 feet; the turbulent flow, hot wall case is typical of conditions which would exist later in

- 30 -

1

THE JOHNS HOPKINS UNIVERSITY APPLIED PHYSICS LABORATORY BILVER SPRING MARYLAND

flight, say below 100 000 feet. As will be shown later, most of the phenomena of interest occur for values of R_e from about 10⁶ to 10⁸. For these values of R_e :

 $C_A \approx 0.1$,

At M = 10 approximately 60% of C_A results from pressure drag (C_{Ap}), 30% from skin friction (C_{AF}), and about 10% from base drag (C_{AB}). The percentages vary somewhat with R_e , M, and type of boundary layer, but the given values are typical of the relative importance of the three sources of axial force for low-drag reentry bodies. For blunted bodies, the pressure drag may account for nearly 100% of the total drag.

Values of C_{mq} (Eq. (2-18)) are shown as a function of $X_{c, g}$ /L in Fig. 2-5, and values of C_{lp} (Eq. (2-19)) are shown in Figs. 2-6 and 2-7.

In the examples included in subsequent sections we will use typical values of the aerodynamic coefficients. The values of the mass and aerodynamic characteristics that will be used are shown in Table 2-1. THE JOHNE HOPKING UNIVERSITY APPLIED PHYSICS LABORATORY BILVER SPRING MARYLAND

BIBLIOGRAPHY

- 2-1 S. A. Schaaf and L. Talbot, <u>Handbook of Super-sonic Aerodynamics</u>, Section 16, "Mechanics of Rarefled Gases," NAVORDSYSCOM Report 1488 (Vol. 5), 1959.
- 2-2 M. Tobak and W. R. Wehrend, <u>Stability Deriva-</u> tives of Cones at Supersonic Speeds, NACA TN 3788, September 1956.

Preceding page blank

ţ

1

[

- 33 -

THE JOHNS HORKINS UNIVERSITY APPLIED PHYSICS LABORATORY BILVER SPRING MARTAND



Preceding page blank

- 35 -

• 1. •

THE JOHNS HOPKINE UNIVERSITY APPLIED PHYSICS LABORATORY Silver Bring Maryland



Fig. 2-2 VARIATION OF VISCOSITY WITH TEMPERATURE



half all the head and a

خد





u t

4

THE JOHNS HOPKING UNIVERSITY APPLIED PHYSICS LABORATORY SILVER EPRING MARTLAND

ŗ

ł

THE JOHNE HOPKING UNIVERSITY APPLIED PHYSICS LABORATORY Silver Bring Maryland



and Suissing

10 A A

Į

and the second second second



THE JOHNS HOPRING UNIVERSITY APPLIED PHYSICS LABORATORY SILVER SPRING MARYLAND



Fig. 2-5 PITCH DAMPING COEFFICIENT FOR 10° SEMIANGLE CONE

.

and the method of the Alexandra for the second



P

THE JOHNS HOPKING UNIVERSITY APPLIED PHYSICS LABORATORY SILVER SPRING MARYLAND

)_i {-Ì

1







Į.

in sand

and a stopped and in

11

THE JOHNS HOPKINS UNIVERSITY APPLIED PHYSICS LABORATORY SILVER EPRING MARYLAND

Table 2-1

Mass and Aerodynamic Characteristics

Mass Characteristics

L = 10 feet S = 9.65 ft² d = 3.5 feet $X_{c.g.} = 6.27$ feet Weight (W) = 1000 pounds Mass (m) = 31.1 slugs Rell moment of inertia (I_X) = 30 slug-ft² Pitch moment of inertia ($I_Y = I_Z$) = 300 slug-ft²

$$\frac{1}{2} = 0.79$$

$$\frac{1}{2} = 0.079$$

Aerodynamic Characteristics

Combined Parameters

 $1b/ft^2$

Sec. Store Blook American Street State

$$C_{N_{\alpha}} = 2 \text{ per radian} \qquad \beta = \frac{W}{C_{A}S} = 10001$$

$$M_{\alpha} = 0.035 \text{ per degree} \qquad S.M. = \frac{\Delta X}{L} = 4\%$$

$$C_{A} = 0.104 \qquad S.M. = \frac{\Delta X}{L} = 4\%$$

$$C_{m_{q}} = -1 \text{ per radian} \qquad \frac{\Delta X}{d} = 0.114$$

$$C_{I_{p}} = -0.003 \text{ per radian}$$

- 42 -

THE JOHNS HOPKINS UNIVERSITY APPLIED PHYSICS LABORATORY BILVER SPRING MARTLAND

I

1

> > I

I

ſ

-

a pa mar Angela a Tangang Angela Nanger A

in the second second

3. ATMOSPHERE

l.

j,

.

İ

ł

••

,AN

THE JOHNE HOPKING UNIVERSITY APPLIED PHYSICS LABORATORY BILVER BPRING MARYLAND

SYMBOLS FOR SECTION 3

Symbol	Definition	Typical Units
ล	speed of sound = $\sqrt{\gamma RT}$	ft/sec
e	Napierian base	**
F_	Coriolis force	pounds
F _P	horizontal force resulting from a pressure gradient	pounds
g	acceleration of gravity	ft/sec"
H	scale height, RT/g_0	føet
h	geometric altitude	feet
ha	geopotential altitude	feet
P	static pressure of dry or moist air	$1b/ft^2$
Ρ.	partial pressure of dry air	1b/ft ²
A P	saturation pressure at T	1b/ft ²
e P	partial pressure of water vapor	lb/ft ²
w R	gas constant for dry air	ft ² /sec ² - "R
R	radius of earth	feet
R	gas constant for water vapor	ſt ² /sec ² - °R
W RAOB's	weather measurements in which a balloon is used as instrumentation carrier	-
ROCOB's	weather measurements in which a rocket is used as instrumentation carrier	-
S. L.	sea level	-
Т	temperature of dry or moist air	٩R
τ _D	dew point temperature	۰R

Preceding page blank

ارون د مېږي اد د مېښې د د ولې الله. د وله و د لکې رومونه کې ۱۹۰۰ و د و ولې د

Symbol	Definition	Typical Units
t	time	seconds
v	missile velocity	ft/sec
V _w	wind speed	ft/sec
W	weight of an elementary volume of air	pounds
dx, dy, dh	dimensions of an elementary volume of air	feot
Z	refers to time at the zero meridian	-
E	change in density defined by Eq. $(3-22)$	percent
γ	ratio of specific heats	-
ø	relative humidity	-
	or	
	latitude	degrees
λ	mean free path	feet
ν	kinematic viscosity	ft ² /sec
P	density of dry air = P/RT	slug/ft ³
p'	density of moist air	slug/ft ³
ω	earth rotation rate	rad/sec

Subscripts

n refers to a reference altitude or to altitudes at which weather measurements are obtained

o refers to sea level conditions

and the second second second second

.

THE IDENS HOPKINS UNIVERSITY APPLIED PHYSICS LABORATORY SELVER SPRING MARYLAND

SUMMARY AND CONCLUSIONS

The atmosphere from see level to about 400 000 feet plays an important role in the motion of reentry vehicles. In this section the general nature of the atmosphere is considered, and equations required for studies concerning vehicle motion are discussed. The variables of most concern are atmospheric density, temperature, wind, and moisture profiles (variations with altitude). The 1962 Standard Atmosphere is used in many analyses. Some of the characteristics of this atmosphere are presented, and differences from other types of standard days are noted.

The major conclusions are:

1. Most phenomena of interest in the analysis of reentry body motion occur in an altitude region called the homosphere (sea level to approximately 300 000 feet). In this region the atmosphere obeys two laws of considerable importance. The perfect gas law provides a relationship between pressure, temperature, and density; the perfect gas law in conjunction with the hydrostatic equilibrium law provides a relationship between density, temperature, and altitude. The hydrostatic law is frequently used as an interpolation equation for tabular values of density versus altitude, permitting relatively few values of the independent variable for a given accuracy. This consideration is of particular importance in trajectory simulation work.

2. Winds are obtained from direct measurements (balloon observations, or RAOB's, for the lower atmosphere and rocket observations, or ROCOB's, for the upper atmosphere), from forecasts based on observations, or from forecasts based on statistical analyses of many previous observations (climatological data). These sources are listed in the order of decreasing accuracy, and all sources

- 47 -

are frequently employed in some phase of analysis of body motion. A law used extensively in the forecast procedure is the geostrophic wind law, which is derived by equating the Coriolis and pressure forces. However, this law becomes invalid at low latitudes.

3. If the monsture content of the air is high, the perfect gas law must be applied using partial pressures of air and water mixtures with appropriate gas constants. Sometimes moisture is accounted for by using an effective (or virtual) temperature. The effect of water vapor is to decrease the density when it is compared with a computed density in which the moisture content is neglected. Neglect of the moisture effect results in a density error of a few percent at most, and the error decreases rapidly with decreasing altitude and decreasing temperature. For most analyses this effect may be ignored.

my of the Preserver

3.1 General Characteristics

The atmosphere is divided into three broad altitude bands called the barosphere, exosphere, and interplanetary gas (Fig. 3-1). The barosphere exists from sea level up to about 500 km (1.65×10^6 feet or 275 nmi). The exosphere exists from 500 to about 60 000 km (33 000 nmi). In the exosphere, the air density is so low that atoms hardly ever collide with one another. Molecules entering at the base of this region describe ballistic trajectories, and some travel beyond the gravitational field of the planet. In this region, temperature has no significance since there is no Maxwellian distribution of velocity. Interplanetary gas exists at altitudes above about 60 000 km.

The barosphere is subdivided into the troposphere, stratosphere, mesosphere, and thermosphere with the tropopause, stratopause, mesopause, and thermopause located between these regions, as shown in Fig. 3-1. These altitude bands are based upon the temperature variation with altitude. The region from sea level to the top of the mesophere is also called the homosphere.

The behavior of each altitude band varies somewhat with earth latitude. Many types of standard atmospheres are in common use; only three will be considered in this section. The 1962 Standard Atmosphere is typical of middle latitude locations; the Tropical Atmosphere is typical of low latitude locations (0° to 20° N. latitude), and the Polar Atmosphere is typical of high latitude locations (north of 60° N. latitude).

In the troposphere the variation of temperature with altitude is strongly influenced by the earth acting as a heat source. The temperature, in general, decreases with increasing altitude. The top of the troposphere occurs at an altitude of 36 000 feet for the Standard Atmosphere, at 55 000 feet for the Tropical Atmosphere, and at 83 000 feet for the Polar Atmosphere. Essentially all weather phenomena occur in the troposphere. THE JOHNS HOPKING UNIVERSITY APPLIED PHYSICS LABORATORY SILVER SPRING MARYLAND

> The tropopause is the top of the troposphere and is an altitude of minimum temperature. For the Standard Atmosphere the temperature is constant from 36 000 to 66 000 feet, where it begins to rise; for the tropical atmosphere the temperature begins to rise at 55 000 feet; for the polar atmosphere the temperature is constant to 96 000 feet, where it begins to rise. The temperature rise is a result of absorption of ultraviolet solar radiations by ozone. The temperature continues to rise with increasing altitude to about 156 000 feet for the Standard Atmosphere. Characteristics to sufficiently high altitudes are not available for the tropical and polar atmospheres. Above the stratosphere the temperature again decreases with increasing altitude, reaching a minimum temperature at the mesopause. For the Standard Atmosphere, this altitude is 263 000 feet. Above the mesopause the temperature again increases with increasing altitude up to the thermopause, which occurs at approximately 1 650 000 feet (275 nmi). The temperature rise is caused by the absorption of ultraviolet rays with a wavelength of 100Å.

The homosphere (sea level to 300 000 feet) is the altitude band of primary interest for analysis of reentry body motion. In this altitude region, the atmosphere is characterized by constant proportions of N_2 , O_2 , and argon, and the atmosphere obeys the hydrostatic equilibrium and perfect gas equations.

For analysis of reentry vehicles, the atmospheric property of most interest is air density; of secondary interest is air temperature. The ratios of density for the tropical and polar atmospheres to the density for the 1962 Standard Atmosphere are shown in Fig. 3-2. It may be expected that the tropical atmosphere would have a lower density and the polar atmosphere a higher density than those of the 1962 Standard Atmosphere. However, this is true only for altitudes below about 20 000 to 30 000 feet. Therefore, depending upon the vehicle design and trajectory, the tropical day may be equivalent to an atmosphere of greater density than that of the 1962

- 50 -

Standard Day, and the polar day may be equivalent to an atmosphere of lesser density than that of the 1962 Standard Day.

The temperature, in addition to the effect on density, affects the speed of sound, and thus Mach number (= V/a). The speed of sound (= $\sqrt{\gamma}$ RT) is proportional to the square root of the ambient temperature and these characteristics are also shown in Fig. 3-2. Note that the speed of sound at a given altitude does not vary from one standard atmosphere to another by more than about 7.5%. This amount of variation is relatively unimportant. However, extreme variations in temperature from the 1.362 Standard Atmosphere, such as those of the winter months at northern latitudes, may have a significant effect on a reentry body trajectory by affecting the Mach number history.

Some characteristics of the 1962 Standard Atmosphere are given as a function of altitude in Table 3-1. More elaborate tables are given in Ref. 3-1. Similar data for the tropical and polar days are given in Refs. 3-2 and $3\cdot 3$.

3.2 Variation of Density and Temperature with Altitude

Consider an elementary volume of air (page 52) of base area dxdy and height dh. If the element of volume is in static equilibrium (the acceleration of the element of volume is zero), then the following equation applies:

$$dxdy(P + \frac{dP}{dh} dh) - Pdxdy + W = 0$$
.

The weight of the volume of air is given by:

 $W = \rho g dx dy dh$,

Therefore,

and the second of the second

 $dP = -\rho g dh \qquad (3-1)$

- 51 -

THE JOHNS HOPKINS UNIVERSITY APPLIED PHYSICS LABORATORY SILVER SPRING MARYLAND



This equation is called the hydrostatic equilibrium equation, and all three standard atmospheres discussed in this section obey this equation. A second equation that is valid in the homosphere is the perfect gas law which states:

$$\mathbf{P} = \mathbf{\rho} \mathbf{R} \mathbf{T} , \qquad (3-2)$$

Differentiating Eq. (3-2):

$$d\mathbf{P} = \mathbf{R}(\mathbf{T}d\boldsymbol{\rho} + \boldsymbol{\rho}d\mathbf{T}) \,. \tag{3-3}$$

Substituting Eq. (3-3) into Eq. (3-1) and expressing g in terms of g_0 :

$$\frac{d\rho}{\rho} + \frac{dT}{T} = -\frac{g_0}{RT} \frac{g}{g_0} dh . \qquad (3-4)$$

ł

all and the second second second second second second second second second second second second second second s

By definition:

$$dh_{G} = \frac{g}{g_{O}} dh$$
 (3-5)

- 52 -

THE JOHNS HOPKINS UNIVERSITY APPLIED PHYSICS LABORATORY BILVER SPRING MARYLAND

where h_G is called the geopotential altitude and h is the geometric altitude. If g is assumed to vary inversely with the square of the distance from the center of the earth:

$$\frac{g}{g_{o}} = \left(\frac{R}{R_{o} + h}\right)^{2}, \qquad (3-6)$$

Substituting Eq. (3-6) into Eq. (3-5) and integrating:

$$h_{G} = \frac{h}{1 + \frac{h}{R_{o}}}$$
(3-7)

 \mathbf{or}

ŀ

$$h = \frac{h_G}{1 - \frac{h_G}{R_o}}$$
 (3-8)

In the remainder of this section, the solution to Eq. (3-4) will be given in terms of the geopotential altitude. However, the results may be obtained in terms of h by using Eq. (3-8). It should be noted that, since $R_0 \approx 2.09 \times 10^7$ feet, $h \approx h_G$ for low altitudes. Even at h = 100 000 feet, h_G and h differ by only 477 feet.

In terms of h_{G} , Eq. (3-4) becomes:

$$\frac{d\rho}{\rho} + \frac{dT}{T} = -\frac{g_0}{RT} dh_G . \qquad (3-9)$$

The closed form solution to Eq. (3-9) depends upon the nature of the variation of T with h_G . For practical applications it is necessary to consider only two cases, a constant temperature (isothermal) relation and a linear relation.

- 53 -

THE JOHNS HOPKING UNIVERSITY APPLIED PHYSICS LABORATORY BILVER SPRING MARYLAND

Consider an altitude interval from h_{G_n} to h_G . Integrating over this altitude band, Eq. (3-9) becomes:

$$ln \frac{\rho}{\rho_n} + ln \frac{T}{T_n} = -\frac{g_o}{R} \int_{h_G}^{h_G} \frac{1}{T} dh_G$$
 (3-10)

If T is constant, Eq. (3-10) may be integrated to give:

$$\boldsymbol{\rho} = \boldsymbol{\rho}_{n} \exp\left(\frac{{}^{h}\mathbf{G}_{n} - {}^{h}\mathbf{G}}{H}\right)$$
(3-11)

where

$$H = \frac{RT}{g_{0}} \quad . \tag{3-12}$$

If T varies linearly over the altitude band, then:

$$T = T_n + \frac{dT}{dh_G} (h_G - h_{G_n})$$
 (3-13)

In this case, Eq. (3-10) may be integrated to give:

$$\boldsymbol{\rho} = \boldsymbol{\rho}_{n} \left(\frac{T}{T_{n}} \right) \qquad (3-14)$$

For calculations, it is frequently desirable to define the density and temperature variations with h_G by an equation rather than to use tables. If the atmosphere obeys the hydrostatic equation, it may be divided into altitude bands for which the temperature is accurately defined by linear variations of T with h_G . Then the density and temperature within a given band are defined using Eqs. (3-11), (3-13), and (3-14) and the conditions at either the lower or upper boundary of the band.

- 54 -

The characteristics of the 1962 Standard Atmosphere may be defined to within 0,5% accuracy using Eqs. (3-11), (3-13), and (3-14) and the data in Table 3-2.

For simple calculations, it is frequently assumed that the entire atmosphere is at a constant temperature. Such an atmosphere is called an isothermal or exponential atmosphere. In this case, Eq. (3-11) becomes

$$\boldsymbol{\rho} = \boldsymbol{\rho}_{0} \exp\left(-\frac{h_{G}}{H}\right). \qquad (3-15)$$

Values of ρ_0 and H are selected to provide the best accuracy for ρ in the altitude region of most interest for the particular study. The quantity H is called the scale height. From Eq. (3-11), the scale height is the change in altitude corresponding to a change in density (or pressure, since T is constant) by a factor of 1/e.

3.3 Nonstandard Atmosphere

Frequently it is necessary to consider atmospheres other than the Standard Atmospheres. The quantities of primary interest for studies of reentry body motion (neglecting such specialized studies as the effects of wind shear and wind gust) are density, temperature, horizontal wind direction, and horizontal wind speed. These data are available from observations, forecasts, or climatological characteristics. The latter data are characteristics based on statistical analyses of many observations.

Observations are made in two altitude regions. Balloons are used to obtain temperature, pressure, dew point, and wind data at altitudes from the surface up to the highest altitude possible. Frequently, the upper limit in altitude is about 30 000 feet. Sometimes the data extend to observations at about 100 000 feet. These observations are called RAOB's. In the northern hemisphere, RAOB's are obtained daily at 00Z and 12Z hours at many observation points. Rockets are used to make observations of

|----| ----

γ

- 55 -

temperature, density (sometimes), and winds as a function of geometric altitude for altitudes from about 60 000 to 200 000 feet. Data from these soundings are called ROCOB's and are made only at a very few locations, usually near 12Z hours Mondays, Wednesdays, and Fridays.

3. 3. 1 RAOB Data - Dry Air

The RAOB pressure and temperature data are converted to density data by Eq. (3-2). The geopotential altitude corresponding to the observed P and T are calculated using the hydrostatic equation. Since data are obtained at a large number of altitudes, the temperatures from one measurement to the next may be averaged and used as a constant so that Eq. (3-11) applies and may be written in the form:

$$h_{G_{n+1}} = h_{G_n} + \frac{R(T_n + T_{n+1})}{2 g_0} \ln \frac{\rho_n}{\rho_{n+1}}$$
(3-16)

where n corresponds to the measurement at one altitude and n+1 corresponds to the measurement at the succeeding altitude.

Equation (3-16) is evaluated in stepwise fashion starting at the surface conditions where h_{G_n} is the surface elevation of the observation location.

3. 3. 2 RAOB Data - Moist Air

The values of density computed using Eq. (3-2) are valid provided the effect of water vapor may be neglected. The effect of water vapor on density may be evaluated by writing the equation for the density of the air-water mixture as the sum of the partial densities:

$$\rho' = \frac{P_A}{RT} + \frac{F_w}{R_w T}$$
(3-17)

- 56 -

THE JOHNS HOPKING UNIVERSITY APPLIED PHYSICS LABORATORY BILVER SPRING MARYLAND

where

$$\mathbf{P} = \mathbf{P}_{\mathbf{A}} + \mathbf{P}_{\mathbf{W}} \quad . \tag{3-18}$$

In terms of ρ and using $R/R_w = 0,622$:

$$p' = p\left(1 - 0.378 \frac{P_w}{P}\right)$$
 (3-19)

If the moisture content is given in terms of relative humidity, $\phi = P_w/P_g$, rather than P_w , then:

$$\rho' = \rho \left(1 - 0.378 \ \phi \frac{P_{B}}{P} \right) , \qquad (3-20)$$

The water vapor pressure, P_w , is the saturation pressure at T_D , and P_s is the saturation pressure at T. Empirical equations for the saturation vapor pressures are:

$$P_{w} = \exp\left(22.65 - \frac{9900}{T_{D}}\right) \cdot$$

$$P_{g} = \exp\left(22.65 - \frac{9900}{T}\right) \cdot$$
(3-21)

The effect of water vapor is to decrease the density compared to a computed density which ignores the effect of moisture. The percent difference in density is given by:

$$\epsilon = 100 \left(\frac{\rho - \rho'}{\rho}\right) = \frac{37.8}{P} \exp\left(22.65 - \frac{9900}{T_D}\right)$$

$$= \frac{37.8}{P} \phi \exp\left(22.65 - \frac{9900}{T}\right).$$
(3-22)

A plot of the error as a function of altitude and dew point is given in Fig. 3-3. Also shown are the standard

- 57 -

temperatures for the 1962 Standard Day and the Standard Tropical Day. For these standard days, the effect of 100% relative humidity ($T_D = T$) on percent change in density is maximum at sea level and is about 0.7% for the 1962 Standard Day and about 1.8% for the Standard Tropical Day. Only for very hot humid days is water vapor likely to have any significant effect on density, and even this effect is significant only for very low altitudes.

3.3.3 ROCOB Data

The effect of moisture is negligible at altitudes where ROCOB data are obtained; therefore Eq. (3-2) is valid. When density and temperature both are available from an observation, the only data reduction required is the conversion from geometric altitude to geopotential altitude by Eq. (3-7). However the density data are usually not measured directly. In this case, Eq. (3-16) may be used in the following form:

$$\rho_{n+1} = \rho_n \exp\left[\frac{2 g_0}{R(T_n + T_{n+1})} \left(h_{G_n} - h_{G_{n+1}}\right)\right] (3-23)$$

where ρ_{n+1} is obtained in stepwise fashion using a known initial value of density (from RAOB data, for example), measured T, and h_G computed from Eq. (3-7) using measured h.

3.3.4 Winds

A wind is defined in magnitude (speed) and direction from which it is blowing, measured clockwise from the north. In some current forecast procedures, the predicted winds are not based on measured winds but upon measured pressures and temperature, using the geostrophic wind equations.

The geostrophic wind is obtained by balancing the horizontal pressure force acting on an elementary mass

- 58 -

"manler d'a

THE JOHNS HOPKINS UNIVERSITY APPLIED PHYSICS LABORATORY BILVER SPRING, MARYLAND

> of air by the Coriolis force. The Coriolis force is perpendicular to the horizontal velocity of the elementary mass and, in the Northern Hemisphere, is directed to the right (to the left in the Southern Hemisphere). The magnitude of the force acting on an elementary mass of air, ρ dxdydh, is given by the equation:

> > $F_c = 2V_w \rho \omega \sin \phi \, dx \, dy \, dh$. (3-24)

A horizontal pressure force acts on the elementary mass whenever a horizontal pressure gradient exists. Consider the following horizontal high pressure system. Let x be the horizontal distance perpendicular to the isobars, y the horizontal distance parallel to the isobars, and h the vertical dimension.



The pressure on the two surfaces of area dxdh are equal, so that the net pressure force acting on the mass in a direction parallel to the isobar is zero. The pressure on the two dxdy surfaces are unequal, but the weight of air is just balanced by the pressure force (hydrostatic

- 59 -

equilibrium is assumed), so that no net vertical force exists. Let the pressure acting on the dydh surface that is farthest from the high pressure center be P. Then the pressure on the dydh surface nearest the center is

 $F + \frac{\partial P}{\partial x} dx$. The corresponding forces are Pdydh and $(P + \frac{\partial P}{\partial x} dx)dydh$. Therefore, the net force is:

$$F_{\mathbf{P}} = \frac{\partial \mathbf{P}}{\partial \mathbf{x}} d\mathbf{x} d\mathbf{y} d\mathbf{h}$$
 (3-25)

The total horizontal pressure force is perpendicular to the isobar and directed away from the high pressure. Since the geostrophic wind is obtained by a balance of the pressure and Coriolis forces, the Coriolis force is perpendicular to the isobar and acts toward the high pressure center, as shown in the sketch. For a Coriolis force in this direction, the wind must be directed along the isobar with high pressure to the right (for the Northerm Hemisphere). Since $F_{c} = F_{D}$, from Eqs. (3-24) and (3-25):

$$V_{\rm w} = \frac{\partial P/\partial x}{2\rho \omega \sin \theta} \qquad (3-26)$$

Equation (3-26), the geostrophic wind equation, is accurate except for the first few thousand feet above the surface where friction effects cause the wind to be inclined away from the isobar toward low pressure and the magnitude is lower than that given by Eq. (3-26). Typically, the maximum deviation in direction is 25°, and the maximum deviation in speed is a factor of one-third to one-half the geostrophic wind, depending upon surface roughness. In addition, the equation is not valid at latitudes near the equator. The equation is valid for the Southern Hemisphere, but the direction of the velocity along the isobar is opposite to that discussed for the Northern Hemisphere.

Equation (3-26) is applied at various altitude levels to obtain the variation of wind speed with altitude. The

1 Same there

THE JOHNS HOPKING UNIVERSITY APPLIED PHYSICS LABORATORY SILVER SPRING MARYLAND

١

ł

ł

1

wind direction is obtained from the slope of the isobar at the location of interest. Therefore, accuracy in wind characteristics requires accurate estimates of density, horizontal pressure gradient, and shape of the isobars.

ł

10101-010-1111-0

and all the there to be the

THE JOHN'S HOPKING UNITERSITY APPLIED PHYSICS LABORATORY Silver Spring Maryland

REFERENCES

- 3-1 U.S. Standard Atmosphere, 1962, prepared under sponsorship of National Aeronautics and Space Administration, United States Air Force, and United States Weather Bureau, Washington, D. C., December 1962.
- 3-2 Climatic Extremes for Military Equipment, Mil. Std. 210A, 2 August 1967.
- 3-3 L. F. Fehlner and E. V. Nice, <u>Tabulation of</u> <u>Standard Atmospheres at 100-Foot Intervals of</u> <u>Altitude</u>, <u>APL/JHU TG 313-1</u>, <u>August 1958</u>.

Preceding page blank

- 63 -

We and the many one



......



- 14 A

1.00

The Later State of the

No. Marsh . To she we

THE JOHNS HOPKING UNIVERSITY APPLIED PHYSICS LABORATORY Silver Spring Maryland

N WIA - SEVE DOLLAR BUT TO A LAST CONTRACT

THE JOHNS HOPKING UNIVERSITY APPLIED PHYSICS LABORATORY Silver Spring Magyland





A Carlos Beer Car

ja,

- 66 -

THE JOHNS HOPKING UNIVERSITY APPLIED PHYSICS LABORATORY SILVER SPRING MARYLAND

ţ

ł





- 67 -

蟫.
က်
đ٥
7
*#
Ë.
-

Atmosphere
Standard
1962

4			н	Н	m	50	7	X
(thousands of feet)	ر <i>م</i> /م	P/P	(.R)	(feet)	(tt/sec)	(ft/sec)	(ît ² /sec)	(feet)
S. L	1.00	1.00	519	27. TE+3	9111	32.17	1.57E-4	2.18E-7
10	0, 739	0.638	483	25.8E+3	1077	32.14	2.01E-4	2.65E-7
20	0.533	0.460	447	23.9E+5	1037	32.11	2.52E-4	4.03E-7
30	0.375	0.298	412	22. 0E+3	669	32.08	3. 49E-4	5. 31 E-7
40	0.247	0.186	390	20.9E+3	968	32.05	5. 06E-4	8.81E-7
90	0.153	0.115	390	20.9E+3	968	32.02	3.16E-1	I.42E-6
60	0.949E-1	0.714E-1	390	20.9E+3	968	31.99	1.32E-3	2.29E-6
70	0.586E-1	0.443E-1	392	21.1E+3	116	31.96	2.14E-3	3.72E-6
80	0.361E-1	0.276E-1	398	21.4E+3	978	31.93	3.52E-3	6. 03E-6
06	0.224E-1	0.174E-1	403	21. 7E+3	964	31.90	5.74E-3	9. 73E-6
100	0.140E-1	0.110E-1	409	22. 0E+3	166	31.87	9.30E-3	1. 56E-5
150	0.145E-2	0.134E-2	479	25.9E+3	1073	31.72	1.02E-1	1.50E-4
200	0. 222E-3	0.195E-3	457	24.95+3	1048	31.57	6.42E-1	9. 81 E-4
250	0. 296E-4	0.201E-4	362	19. 2E+3	920	31.42	3.87E 0	7.35E-3
300	0.195E-5	0.125E-5	333	18. 3E+3	894	31.27	5. 61 E+1	1.12E-1
350	0.132E-6	0.112E-6	435	24. 2E+3	1021	31.12	1.03E+3	1.63E 0
400	0.152E-7	6.211E-7	694	39.8E+3	1235	30.97	1.23E+4	1,38E+1
- 0 0000 U	3							

- 68 -

.

ρ₀ = 0.002377 slug/ft³ 2

-

 $P_0 = 2116 \text{ Ib/}\text{ft}^2$

I.

ì

s
THE JOHNS HOPKING UNIVERSITY APPLIED PHYSICS LABORATORY SILVER SPRING MARYLAND

Table 3-2

Constants for the 1962 Standard Atmosphere

h _G	T	$\frac{dT}{dh}_{G}^{*}$ (°R/ft)	ρ
(feet)	(°R)		(slug/ft ³)
0.000	(-K) 518.670 389.970 389.970 411.570 487.170 487.170 487.170 325.170 325.170 379.170 469.170 1729.170 1999.170 2179.170 2431.170 2791.170	-3.56619991E-03	2, 37689982E-03
36089.238		0.00000000E+00	7, 06119929E-04
65616.750		5.48639800E-04	1, 70820000E-04
104985.875		1.53619982E-03	2, 56609928E-05
154199.375		0.00000000E+00	2, 76979972E-06
170603.625		-1.09729986E-03	1, 47349965E-06
200131.188		-2.19459995E-03	4, 87189993E-07
259186.313		0.0000000E+00	3, 88259984E-08
291153.188		1.69529999E-03	6, 15079898E-09
323002.625		2.83429981E-03	9, 65109992E-10
354753.250		5.68669662E-03	1, 90709989E-10
386406.125		1.14439987E-02	4, 72659967E-11
480780.813		8.63499939E-03	3, 56239951E-12
512045.938		5.77439740E-03	2, 24879924E-12
543215.188		4.06099856E-03	1, 55919982E-12
605268.375		2.92699994E-03	8, 43449984E-13
728243.750		2.38109985E-03	3, 03459948E-13
939894,688	3295.170	2.01519998E-03	6.95599719E-14
1234645,000	3889.170	1.63550000E-03	1.26079973E-14
1520799,000	4357.168	1.10109989E-03	3.05989996E-15
1798726,000	4663.168	7.32979970E-04	9.00309837E-16
2068776,000	4861.168	0.00 000000E+00	3.04629894E-16

*The temperature gradient is applicable to the altitude band beginning at the altitude for which the gradient is listed and ending at the next highest altitude.

- 69 -

THE JOHNE HOPKINE UNIVERSITY APPLIED PHYSICS LABORATORY SILVER SPRING. MARYLAND

おきるというないであるというないであるというないというというというというというという

ľ

I

I

Ĩ,

I

Ţ

I

Ĩ

I

I

¶*' | ||

I

1

į,

4. EQUATIONS OF MOTION

11

÷Й

4 EQUATION

in the state of the second sec

...

A CONTRACT CONTRACTOR

Preceding page blank

State and the second second

The motion of a reentry body during ballistic flight is affected primarily by five factors: の日本は国住住は「日本日本」のである。日本

A THE REPORT OF THE PARTY OF TH

- 1. Gravity,
- 2. Earth rotation,
- 3. Atmosphere,
- 4. Body dynamics,
- 5. Asymmetries.

Each factor is considered in simplified form to show the general effects on the trajectory.

- 73 -

THE JOHNS MOPKING UNIVERSITY APPLIED PHYSICS LABORATORY BILVER SPRING, MARYLANS

;

•

Ĩ

I

I

I

제위 // 해가:

: 1

I

I

11-1

10 4

2

1

I

Ĩ

Ĩ

Ĩ

، ۲۰ سانانین 11.14

1. U 1. L 1.

4.1 EFFECTS OF GRAVITY

Preceding page blank

- 75 -

N

ч

SYMBOLS FOR SECTION 4.1

Symbol	Definition	Typical Units
g	acceleration of gravity	ft/sec^2
I	constant defined by Eq. (4,1-15)	-
к	constant defined by Eq. (4.1-5)	-
R	range	feet or nmi
r.	distance from the center of the earth to a point on the trajectory	feet
r	r/r _o	-
t	time	seconds
v	velocity	ft/sec
$\overline{\mathbf{v}}$	V/Vgoro	-
γ	flight path angle	degrees
θ	central angle, measured from apogee	radians or degrees

- 77 -

Subscripts

and the second se

and a street

•

a	refers to conditions at apogee
e	refers to conditions on the surface of the earth
0	refers to initial conditions

Preceding page blank

Hear Popular Sugar Street

· :

THE JOHNS HOPKINS UNIVERSITY APPLIED PHYSICS LABORATORY SILVER SPRING MARYLAND

SUMMARY AND CONCLUSIONS

The effect of gravity only on the trajectory of a ballistic missile is considered. Equations that define the trajectory (velocity, time, flight path angle, altitude, and range) in terms of initial conditions are presented.

Plots are given which show the relationship of range and flight time as a function of initial velocity and flight path angle for minimum and nonminimum energy trajectories. Also shown are plots of range sensitivity to initial velocity and flight path angles. An example of the uses of the equations is given in Example 1, Section 4.6.

The major conclusions are:

1. A ballistic missile follows an elliptical flight path. For a given initial velocity, there is an initial flight path angle that results in maximum range, and the trajectory is called a minimum energy trajectory. For minimum energy trajectories, the flight path angle decreases from 45° for very short range to approximately 32° for a range of 3000 nmi; the velocity increases from 0 ft/sec at zero range to approximately 20 000 ft/sec at 3000-nmi range; the total flight time increases from 0 seconds at zero range to about 1300 seconds for 3000-nmi range; the trajectory apogec increases from 0 nmi at 0 nmi range to about 550 nmi at 3000-nmi range (a maximum apogee of about 700 nmi occurs for a range of 5500 nmi and a flight path angle of 22.5°).

2. Trajectories for which the initial flight path angles are greater than those for the minimum energy trajectory are called lofted trajectories; trajectories for which the initial flight path angles are less than those for the minimum energy trajectory are called delofted, or depressed, trajectories. For a given range, the total flight time increases with increasing flight path angle.

Preceding page blank - 79 -

THE JOHNS HOPKINS UNIVERSITY APPLIED PHYSICS LABORATORY BILVER SPRING MARYLANS

and a second state and a low the second second second second second second second second second second second s

3. The range for a minimum energy trajectory is extremely sensitive to initial velocity (about 2800 ft/ft/sec at 3000-nmi range) but is very insensitive to initial flight path angle. With increasing loft at a range of 3000 nmi, the sensitivity to initial velocity decreases but the sensitivity to initial flight path angle increases. For delofted trajectories at 3000-nmi range, the sensitivity to both initial velocity and initial flight path angle is greater than that for a minimum energy trajectory.

i 1

and the strate states and

THE JOHNE HOPKINE UNIVERSITY APPLIED PHYSICS LABORATORY BILVER SPRINE MARYLAND

Consider a nonrotating, spherical earth without an atmosphere. The geometry of the trajectory is shown in the sketch.



The trajectory of a body with initial velocity V_0 at r_0 and fired with a flight path γ_0 may be obtained by applying the laws of gravity, conservation of energy, and conservation of angular momentum as defined by Eqs. (4.1-1), (4.1-2), and (4.1-3), respectively.

$$g_0 r_0^2 = gr^2$$
 (4.1-1)

$$\frac{V_0^2}{2} - \frac{V^2}{2} = g_0 r_0 - gr \qquad (4.1-2)$$

$$r_{0}V_{0}\cos\gamma_{0} = rV\cos\gamma \qquad (4.1-3)$$

It may be shown that the trajectory is elliptical for $\frac{v_0^2}{2g_0r_0} < 1$, parabolic for $\frac{v_0^2}{2g_0r_0} = 1$, and hyperbolic for

- 81 -

THE JOHNS HOPKING UNIVERSITY APPLIED PHYSICS LABORATORY SILVER SPRING MARYLAND

> $v_0^2 > 1$. For ballistic vehicles, $\frac{v_0^2}{2g_0 r_0}$ is always less than 1 so that the trajectory is elliptical. Using Eqs. (4.1-1) through (4.1-3) and the properties of an ellipse, it may be shown that the trajectory is defined by the equation:

$$\overline{r} = \frac{\overline{V}_{o}^{2} \cos^{2} \gamma_{o}}{1 - K \cos \theta}$$
(4.1-4)

where

$$\overline{r} = \frac{r}{r_{o}}$$

$$\overline{V}_{o}^{2} = \frac{V_{o}^{2}}{g_{o}^{r_{o}}}$$

$$K = \left[1 - \overline{V}_{o}^{2} (2 - \overline{V}_{o}^{2}) \cos^{2} \gamma_{o}\right]^{1/2}, \quad (4.1-5)$$

The range is the arc length OB and is given by the equation:

$$\mathbf{R} = 2\mathbf{r}_{\mathbf{\theta}} \mathbf{\theta} \tag{4.1-6}$$

where θ_0 can be obtained from Eq. (4.1-4) by setting r = 1 and solving for $\cos \theta_0$. Thus:

$$\cos \theta_{0} = \frac{1 - \overline{V}_{0}^{2} \cos^{2} \gamma_{0}}{K}$$

$$\tan \theta_{0} = \frac{\overline{V}_{0}^{2} \cos \gamma_{0} \sin \gamma_{0}}{1 - \overline{V}_{0}^{2} \cos^{2} \gamma_{0}} \qquad (4, 1-7)$$

or

- 82 -

A strong of the provides surgers.

or

It should be noted that the equations derived above are general insofar as the reference altitude, r_0 , is concerned. That is, r_0 is not necessarily a surface condition. However, if r_0 is not a surface condition, care must be exercised in evaluating the reference velocity, $\sqrt{r_0g_0}$, and R_0 . In terms of surface conditions:

$$\sqrt{r_o g_o} = \sqrt{(r_e g_e) \frac{r_e}{r_o}} = 26\ 000 \sqrt{\frac{r_e}{r_o}}$$
 (4.1-8)

$$R_e = 2r_e \theta_o \,. \tag{4.1-9}$$

From Eqs. (4, 1-6) and (4, 1-7), it is observed that a given range may be obtained for various combinations of V_0 and γ_0 . However, there is one combination of V_0 and γ_0 that results in minimum V_0 . This trajectory, called the minimum energy trajectory (MET), represents the minimum booster impulse for a given range or, alternatively, represents the maximum range that may be obtained from a given booster. The minimum energy trajectory is obtained by operating on Eq. (4, 1-7) using standard procedures for obtaining maximum or minimum values. The results are the following:

$$\gamma_{0} = \frac{57.3}{4} \left(\pi - \frac{R}{r_{0}}\right)$$

$$\gamma_{0} = 45 - \frac{\theta_{0}}{2}$$

$$(4.1-10)$$

$$\overline{\nabla}_{o}^{2} = 1 - \tan^{2} \gamma_{o} \qquad (4.1-11)$$

It is noted that γ_0 decreases from 45° at very short range to 0° at maximum range ($R/r_0 = \pi$). The velocity increases from 0 at zero range to $V_0 = \sqrt{g_0 r_0}$ at maximum range. The latter velocity is the velocity for a circular

- 83 -

Al Martin Constants

orbit. For ranges that will be considered in this report (R = 1000 to 3000 nmi) the values of γ_0 and V_0 computed from Eqs. (4.1-6) and (4.1-7) are shown in Figs. 4-1 and 4-2. For all examples given in this section, $r_0 = r_e = 2.09 \times 10^7$ feet.

The time of flight may be derived in the following manner. The rate of change in θ is given by the following expression:

$$\dot{\theta} = -\frac{V\cos\gamma}{r} \quad . \tag{4.1-12}$$

The negative sign is required since, for convenience in deriving the equation of the ellipse, θ is measured from apogee. Using Eqs. (4.1-3) and (4.1-12) the equation for time becomes:

$$t = \frac{r_0}{V_0 \cos \gamma_0} \int_{\theta}^{\theta_0} \overline{r}^2 d\theta . \qquad (4, 1-13)$$

Substituting for \overline{r} from Eq. (4.1-4) and integrating gives the result:

$$t = \sqrt{\frac{r_o}{g_o}} \left(\overline{V}_o \cos \gamma_o\right)^3 I \qquad (4.1-14)$$

where I is given by the equation:

$$1 = \frac{1}{\overline{\nabla_{0}^{2}(2 - \overline{\nabla_{0}^{2}})\cos^{2}\gamma_{0}}} \left[\frac{K\sin\theta}{1 - K\cos\theta} + \frac{2}{\sqrt{1 - K^{2}}}\tan^{-1}\left(\frac{\sqrt{1 - K^{2}}\tan(\theta/2)}{1 - K}\right) \right] \begin{vmatrix} \theta_{0} \\ \theta_{0} \end{vmatrix}, (4.1-15) \\ \theta_{0} \end{vmatrix}$$

and is evaluated between the limits θ and θ_0 .

If one is interested in just the total trajectory time, Eqs. (4.1-14) and (4.1-15) may be simplified by noting

- 84 -

THE JOHNS HOPKINS UNIVERSITY APPLIED PHYSICS LABORATORY SILVER SPRING MARYLAND

that the trajectory is symmetrical about $\theta = 0$, and the equation for the time becomes:

$$t = \frac{2V_o}{g_o} \frac{\cos \gamma_o}{\left(2 - \overline{V_o}^2\right)} \left[\frac{K \sin \theta_o}{1 - K \cos \theta_o} + \frac{2}{\sqrt{1 - K^2}} \tan^{-1} \left(\frac{\sqrt{1 - K^2} \tan \left(\theta_o/2\right)}{1 - K} \right) \right]. \quad (4.1-16)$$

energy trajectory, K = $\tan \gamma_0$ and Eq. (4.1-16) may be written as:

$$= 2 \sqrt{\frac{r_o}{g_o}} \cos^3 \gamma_o \left[\tan \gamma_o \sqrt{1 - \tan^2 \gamma_o} + 2 \tan^{-1} \left(\frac{\sqrt{1 - \tan^2 \gamma_o} \tan \frac{R}{4r_o}}{1 - \tan \gamma_o} \right) \right]. \quad (4.1-17)$$

The time for a minimum energy trajectory varies from 0 seconds at zero range to $\pi \sqrt{r_0/g_0} = 2520$ seconds for maximum range (R = πr_0). For ranges of 1000 to 3000 nmi, the flight time is shown in Fig. 4-4.

The local flight path angle may be expressed in the form:

$$\tan \gamma = -\frac{\mathrm{d}\mathbf{r}}{\mathrm{r}\mathrm{d}\theta} \quad . \tag{4.1-18}$$

- 85 -

Solving for $dr/d\theta$ from Eq. (4.1-4) and substituting into Eq. (4.1-18):

$$\tan \gamma = \frac{K \sin \theta}{1 - K \cos \theta} \qquad (4.1-19)$$

Differentiating Eq. (4.1-19) with respect to t and using Eq. (4.1-12), it may be shown that \dot{y} is given by the following equation:

$$\dot{\gamma} = \frac{K(K - \cos \theta)}{(1 - K \cos \theta)^2} \frac{V \cos^3 \gamma}{r} , \qquad (4.1-20)$$

Evaluating Eq. (4, 1-20) for conditions near impact, it may be shown that $\dot{\gamma}$ is very small for 1000 to 3000 nmi trajectories. Therefore, in the absence of an atmosphere, the flight path during reentry is nearly a straight line. This fact will be useful later when we consider the effects of the atmosphere.

The velocity at any point along the trajectory may be obtained by solving Eqs. (4.1-1), (4.1-2), and (4.1-4) to obtain the following equation:

$$\overline{V} = \frac{\sqrt{K^2 - 2K \cos \theta + 1}}{\overline{V}_0 \cos \gamma_0} \qquad (4.1-21)$$

At apogee $\theta = \gamma = 0$, and from Eqs. (4.1-4) and (4.1-21):

$$\overline{r}_{a} = \frac{\overline{\nabla}_{o}^{2} \cos^{2} \gamma_{o}}{1 - K}$$
$$\overline{\nabla}_{a} = \frac{1 - K}{\overline{\nabla}_{o} \cos \gamma_{o}}$$

-t-

5.

j)

しんとある時間

「「「」の「一路」

- 86 -

orbit. For a minimum energy trajectory, K = $\tan \gamma_0$ and $\overline{V_0} = 1 - \tan^2 \gamma_0$, so that the altitude (r - r₀) and velocity at apogee are given by the equations:

$$h_{a} = r_{0} \left[(\sin \gamma_{0} + \cos \gamma_{0}) \cos \gamma_{0} - 1 \right] \qquad (4.1-22)$$

$$V_{a} = \frac{\sqrt{g_{o}r_{o}}}{\cos\gamma_{o}} \sqrt{\frac{1 - \tan\gamma_{o}}{1 + \tan\gamma_{o}}}$$
(4.1-23)

or

$$V_{a}/V_{o} = \frac{1}{\cos \gamma_{o} + \sin \gamma_{o}}$$
 (4.1-24)

For a minimum energy trajectory, the altitude at apogee is maximum ($h_a \approx 700$ nmi for $r_o = r_e$) for $\gamma_o =$ 22.5° ($R \approx 5500$ nmi) and varies from about 250 nmi for R = 1000 nmi to about 550 nmi for R = 3000 nmi. Since the sensible atmosphere extends to about 50 nmi, most of the trajectory occurs in the exoatmosphere. The ratio of velocity at apogee to initial velocity decreases from 1 at zero range to $1/\sqrt{2}$ for maximum range. For ranges of 1000 to 3000 nmi, V_a/V_o varies from 0.71 to 0.72.

Another consideration of interest is the sensitivity of range to the initial values of V_0 and γ_0 . These sensitivity values may be obtained in the following manner:

$$\frac{\partial \mathbf{R}}{\partial \mathbf{V}_{0}} = \frac{\partial \mathbf{R}}{\partial \theta_{0}} \frac{\partial \theta_{0}}{\partial \mathbf{V}_{0}} \quad . \tag{4.1-25}$$

 $\partial R/\partial \theta_0$ and $\partial \theta_0/\partial V_0$ may be obtained from Eqs. (4.1-6) and (4.1-7) so that:

$$\frac{\partial R}{\partial V_o} = 2 \sqrt{\frac{r_o}{g_o}} \frac{\overline{V_o} \sin 2\gamma_o}{K^2} . \qquad (4.1-26)$$

- 87 -

Similarly, $\partial R/\partial \gamma_0 \mod y$ be obtained from Eq. (4.1-7):

$$\frac{\partial R}{\partial \gamma_0} = \frac{2r_0}{57.3} \frac{\overline{V_0}^2 (\cos 2\gamma_0 - \overline{V_0}^2 \cos^2 \gamma_0)}{K^2} \quad (4.1-27)$$

Plots of Eqs. (4.1-26) and (4.1-27) are shown in Figs. 4-5 and 4-6, respectively. Note that the range of a minimum energy trajectory is sensitive to initial speed but not to initial flight path angle. With increasing loft angle, the range sensitivity to V_0 decreases but range sensitivity to γ_0 increases.

THE JOHNS HOPKING UNIVERSITY APPLIED PHYSICS LABORATORY SILVER BRING MARYLAND





- 89 -

「「ないない」というという





1

State & Sugar Sec.

THE JOHNE HOPKING UNIVERSITY APPLIED PHYSICS LABORATORY BILVER BPRING, MARYLAND

の話がないたというできます。

. . . .



Fig. 4-3 FLIGHT TIME VERSUS γ_{o} , R = 3000 NMI



THE JOHNS HOPKING UNIVERSITY APPLIED PHYSICS LABORATORY SILVER SPRING MARYLAND





1

の原料調査の法律には用いていていたいで、

and the second second second second second second second second second second second second second second second

· · ·

4

· · · · ·

A set of the set of

-

言明者も言語して

ļ

and the second second

1

!

- 92 -

THE JOHNS HOPKING UNIVERSITY APPLIED PHYSICS LABORATORY Silver Spring, Maryland



Fig. 4-5 RANGE SENSITIVITY TO V_a

and the second

- 93 -

THE JOHNS HOPKING UNIVERSITY APPLIED PHYSICS LABORATORY Silver Spring, Maryland

A THE PARTY AND A PARTY OF A PART



- 94 -

and the State of States and States

A D. Martin Barbara and A.

1.0

Charles and a state of the second

ALC: NO.

in the second states

1

٠.

THE JOHNE HOPKING UNIVERSITY APPLIED PHYSICS LABORATORY GILVER SPRING, MARYLAND

I

7 **W**-#

***** * ۰.

1 1 1

÷.

1

I

n' N

4.00

4.2 EFFECTS OF EARTH ROTATION

يا السبية ال

「「「「「」」」 4.2 EFFECTS OF EARTH ROTATION ٠ŧ

4

۰.

THE JOHNS HOPKING UNIVERSITY APPLIED PHYSICS LABORATORY BILVER SPRING MARYLAND

·

2

SYMBOLS FOR SECTION 4, 2

....

Symbol	Definition	Typical Units
CA	axial force coefficient	-
h	altitude	feet or nmi
I	refers to the impact location	-
L	refers to the launch location	-
R	range from L to I	nmi
r	$r_e + h$	feet
r e	earth radius = 2.09×10^7 = 3441	feet nmi
S	reference area	${\rm ft}^2$
t	total flight time	seconds
Δt	increment in reentry time	seconds
v	velocity in inertial coordinates	ft/sec
W	vehicle weight	pounds
β	vehicle ballistic coefficient, $\frac{W}{C_A S}$	1b/ft ²
γ	flight path angle in inertial coordinates (positive for launch angles)	degrees
ø	latitude, positive for N. latitude	degrees
Λ	longitude, positive for E. longitude	degrees
$\Delta \Lambda$	earth rotation angle during time t	degrees
ψ	azimuth angle, positive direction is clockwise from north	degrees
ψ_{LI}	azimuth angle from L to I (Eq. (4.2-19))	degrees
θο	half the range angle for nonrotating earth (Eq. (4.1-7))	radian

Preceding page blank

- 97 -

Sec. Sec. Burnetters

THE JOHNE HOPKING UNIVERSITY APPLIED PHYSICS LABORATORY BILVER SPRING MARYLAND

Symbol	Definition	Typical Units
θ _e	half the range angle for rotating earth (Eq. (4. 2-17))	radians
ω _e	earth rotation rate = 0, 729 $\times 10^{-4}$	rad/sec
Charle or marker		

181

1

111

.....

The second

4

.

.

* *

٠

÷-----

T

Sec. Warth and an

. . .

Subscripts

ł

e	refers to quantities measured relative to rotating earth
I	refers to impact condition
L	refers to launch condition

THE JOHNS HOPKINS UNIVERSITY APPLIED PHYSICS LABORATORY SILVER SPRING MARYLAND

SUMMARY AND CONCLUSIONS

The equations of motion given in the preceding section are applicable to the rotating earth case provided the trajectory is defined in the inertial axis system. Therefore, initial conditions must be computed taking into consideration the velocity of the rotating earth. In addition, the amount of earth rotation during the time interval from launch to impact must be accounted for. In this section equations are given for evaluating the magnitude of these effects and an example is given in Section 4.6, Example 2.

The major conclusions are:

1. Compared to a nonrotating earth calculation, the effect of a rotating earth on a missile fired eastward is to increase the initial inertial velocity and to decrease the initial flight path angle. Westward firings result in a decreased velocity and an increased flight path angle. The largest change in velocity (about 1500 ft/sec) occurs for launches at the equator, for eastward or westward firings, and for very low flight path angles.

2. Compared to a nonrotating earth calculation, the effect of a rotating earth on a missile fired eastward is to increase the range and flight time. The impact latitude is the same for both calculations. For a westward firing the range and flight time are decreased. The three basic laws used in Section 4.1 to define the equations of motion are valid for a coordinate system fixed in space (nonrotating earth). Two major effects of the earth's rotation on the flight of ballistic vehicles will be considered.

1. The launch site has a velocity caused by the earth's rotation so that launch velocity and launch direction are modified compared to the nonrotating earth solution.

2. The earth rotates while the missile is in flight, thereby affecting the location of the impact point.

Assume that a booster imparts to a vehicle a velocity V_e at an azimuth angle ψ_e and a flight path angle γ_e , where all three quantities are measured relative to a rotating earth. The earth geometry required for this study is shown in the sketch, page 102.

The vehicle will travel in an orbital (or trajectory) plane that contains the velocity vector in inertial space, V, and the center of the earth. The characteristics of the trajectory are defined by the equations derived in Section 4.1, provided inertial quantities are used. The inertial quantities may be obtained from earth reference quantities as shown below. The velocity vector is divided into three components:

Radial component = $V \sin \gamma$ (4.2.1)

North component = $V \cos \gamma \cos \psi$ (4.2-2)

East component = $V \cos \gamma \sin \psi$ (4.2-3)

Preceding page blank

THE JOHNS HORKING UNIVERSITY APPLIED PHYSICS LABORATORY BILVER BRING MARYLAND



In inertial coordinates the radial component and north component of velocity are the same as those measured in the earth fixed axes, but the east component is increased by ω_e r cos \emptyset . Therefore, the components of the inertial velocity, V, are:

Radial component =
$$V_{\mu} \sin \gamma_{\mu}$$
 (4, 2-4)

North component = $V_e \cos \gamma_e \cos \psi_e$ (4.2-5)

East component =
$$V_e \cos \gamma_e \sin \psi_e + \omega_e r \cos \emptyset$$
 (4.2-6)

- 102 -

1. I.

THE JOHNS HOPKING UNIVERSITY APPLIED PHYSICS LABORATORY SILVER SPRING MARYLAND

The azimuth angle, flight path angle, and velocity in inertial coordinates may be obtained from the following equations:

 $\tan \psi = \frac{\text{eastward component of velocity}}{\text{northward component of velocity}}$

$$\frac{\tan\psi}{\tan\psi_e} = 1 + \frac{\omega_e r \cos\theta}{V_e \cos\gamma_e \sin\psi_e}$$
(4.2-7)

$$\sin \gamma = \frac{\text{radial component of velocity}}{\text{total velocity}}$$

$$\frac{\sin\gamma}{\sin\gamma_{\rm e}} = \frac{\rm V_{\rm e}}{\rm V} \tag{4.2-8}$$

$$\frac{\mathbf{V}}{\mathbf{V}_{e}} = \left[1 + 2\cos\gamma_{e}\sin\psi_{e}\frac{\omega_{e}\mathbf{r}\cos\phi}{\mathbf{V}_{e}} + \left(\frac{\omega_{e}\mathbf{r}\cos\phi}{\mathbf{V}_{e}}\right)^{2}\right]^{1/2} (4.2-9)$$

For launch velocities of interest, the velocity ratio is given by the approximation:

$$\frac{V}{V_e} = 1 + \cos \gamma_e \sin \psi_e \frac{\omega_e^r \cos \varphi}{V_e}$$
(4.2-10)

$$\omega_e = 0.729 \times 10^{-4} \text{ rad/sec}$$
 (4.2-11)

$$r \approx r_e = 2.09 \times 10^7$$
 feet (4.2-12)

$$\omega_{\rm e} r_{\rm e} = 1520 \, {\rm ft/sec} \, . \qquad (4.2-13)$$

From Eqs. (4.2-8) and (4.2-10), it is observed that for eastward launches ($\psi_e = 0^\circ$ to 180°, $V/V_e > 1$ and tends to be maximum for launches from the equator ($\emptyset = 0$) for

- 103 -

.14, 1.

minimum energy trajectories (minimum V_e), for direct east firings ($\psi_e = 90$), and for low flight path angles ($\gamma_e \rightarrow 0$). For eastward trajectories the inertial velocity is increased and the flight path angle is decreased (compared to the values for a nonrotating earth); for westward firings, V is decreased and γ increased (compared to the values for a nonrotating earth). The maximum variation of V/V_e from unity is $\pm \omega_e r/V_e$. For $V_e \ge 10\,000$ ft/sec the maximum variation in V/V_e is about 15%.

Given values of V_e and γ_e , the inertial values of V and γ may be obtained from Eqs. (4.2-8) and (4.2-10). These values of V and γ may be used in equations given in Section 4.1. However, the range computed by Eq. (4.1-9) corresponds to the range measured along a sphere fixed in space. The range measured on the rotating earth (in the absence of an atmosphere) may be computed as shown below.

The orbital plane defined by V and the center of the earth remains fixed in space. During the flight time the earth rotates through an angle:

$$\Delta \Lambda = 57.3 \omega_{\rm c} t$$
, (4.2-14)

For given inertial launch conditions, the impact latitude is unaffected by ω_e and, from spherical trigonometry,

 $\sin \phi_{I} = [\sin 2\theta_{0} \cos \phi_{L} \cos \psi_{L} + \cos 2\theta_{0} \sin \phi_{L}] \quad (4.2-15)$

where θ_0 is the value computed from Eq. (4.1-7) for $\gamma_0 = \gamma$ and $V_0 = V$. The impact longitude is given by:

$$\Lambda_{I} = \Lambda_{L} - 57.3\omega_{e}t + \sin^{-1}\left(\frac{\sin\psi_{L}\sin2\theta_{o}}{\cos\theta_{I}}\right) \qquad (4.2-16)$$

where t is the flight time computed from Eqs. (4.1-7) and (4.1-16) for $V_0 = V$ and $\gamma_0 = \gamma$.

- 104 -

The of the shirt of

THE JOHNE HOPKING UNIVERSITY APPLIED PHYSICS LABORATORY SILVER SPRING MARYLAND

Given \emptyset_L , Λ_L , \emptyset_I , and Λ_I the range may be computed from the equation:

$$\mathbf{R} = 2\mathbf{r}_{e}\boldsymbol{\theta}_{e} \quad (4, 2-17)$$

From the spherical geometry shown in the sketch, the range angle $2\theta_e$ may be obtained from;



 $\cos 2\theta_{e} = \cos \theta_{I} \cos \theta_{L} \cos \Delta A + \sin \theta_{I} \sin \theta_{L} . \quad (4.2-18)$

The bearing of the impact from the launch site may be obtained from

$$\cos \psi_{\text{LI}} = \frac{\cos 2\theta_{\text{e}} \cos \phi_{\text{L}} - \cos \phi_{\text{I}} \cos \Delta \Lambda}{\sin 2\theta_{\text{e}} \sin \phi_{\text{L}}}, \qquad (4.2-19)$$

- 105 -

For $\emptyset_L = 0$, Eq. (4, 2-19) becomes:

$$\cos \psi_{\text{LI}} = \frac{\sin \phi_{\text{I}}}{\sqrt{1 - \cos^2 \phi_{\text{I}} \cos^2 \Delta \Lambda}} \qquad (4.2-20)$$

Sometimes V, γ , and ψ are given and it is desirable to obtain the corresponding quantities relative to a rotating earth. The required equations are:

$$\frac{\mathbf{V}_{\mathbf{e}}}{\mathbf{V}} = \left[1 - \frac{2\omega_{\mathbf{e}}\mathbf{r}\cos\gamma\sin\psi\cos\phi}{\mathbf{V}} + \left(\frac{\omega_{\mathbf{e}}\mathbf{r}}{\mathbf{V}}\right)^{2}\cos^{2}\phi\right]^{1/2} (4.2-21)$$

$$\frac{\sin \gamma_{\rm e}}{\sin \gamma} = \frac{\rm V}{\rm V_{\rm e}} \tag{4.2-22}$$

$$\frac{\tan\psi_e}{\tan\psi} = 1 - \frac{\omega_e r \cos \emptyset}{V \cos \gamma \sin \psi} . \qquad (4.2-23)$$

. :

1

all of the most of the transfer of

THE JOHNE HOPKINE UNIVERSITY APPLIED PHYSICS LABORATORY SILVER SPRING. MARYLAND

I

I

I

I

I

I

I

]

وللمرد اللط

ويعتبرون فكرده

É.

1991年1991年19月1日日本部署各部合理理论工作的研究性的研究性的研究性的性质。19月1日中国的基础的基础和基础和基础和基础和基础的研究和基础和基础和基础和基础和基础和基础和基础和基础和基础和基础和

4.3 EFFECTS OF ATMOSPHERE

4

Ĵ

HAP STOLEN I TERSTON STERE

4

· · · · · ·

ć,

ż

- 107 -

سهدهم الحال والمطاورة المطاور الإمارية ومراجع منادو والمتحدث والمعطوة محمولها والأطرافة معاهمه الكافية

-

ı.

÷

ł

÷

i

SYMBOLS FOR SECTION 4.3

Symbol	Definition	Typical Units
A	axial force	pounds
a	speed of sound	ft/sec
C,	axial force coefficient	-
C _m	pitch damping coefficient	-
ц С _м	normal force coefficient	-
	dC _N /da	-
D	damping factor (Eq. (4. 5-20))	-
d	reference length	feet
g	acceleration of gravity	ft/sec ²
- Bw	longitudinal acceleration	g
-x H	scale height = RT/g	feet
h	altitude above sea level	feet
h _{max}	altitude at V max	feet
I ₁ (K)	$\sum_{n=1}^{\infty} \frac{(K/2)^n}{n \cdot n!}$	-
1 ₂ (k)	$e^{-K}\sum_{n=0}^{\infty}\frac{(2K)^n}{1\cdot 3\cdot 5\cdot \cdot \cdot (2n+1)}$	
Ī	pitch (or yaw) moment of inertia	slug-ft ²
-Y K	parameter defined by Eq. (4.3-6)	-
k.	constant defined by Eq. (4.3-26)	-
1 L	reference length	feet
M	Mach number = V/a	-
m	vehicle mass	slugs
P	atmospheric pressure	lb/ft ²

Mar Carling

and a subtraction of

and the state of the

Precoding page blank

- 109 -

. . .

۰

وتقويلهم والمتكافة فتقارضهم والمراجع والمراجع والمراجع والمراجع والمراجع والمراجع والمراجع والمراجع والمراجع

<u>Symbol</u>	Definition	Typical Units
q	dynamic pressure = $\frac{\gamma}{2}$ PM ²	$1b/ft^2$
q	stagnation heating rate per unit area	Btu/ft ² sec
ຊື	total heat per unit area absorbed at the stagnation point	Btu/ft^2
R	gas constant = 1716	ft ² /sec ² - "R
Re	Reynolds number, VL/ν	-
R _N	vehicle nose radius	feet
S	reference area	ft^2
Т	atmospheric temperature	۰R
t	time	seconds
$\Delta t_1, \Delta t_2$	intervals of time defined by Eqs. (4.3-20) and (4.3-21)	seconds
v	vehicle velocity	ft/sec
W	vehicle weight	pounds
Х _{с. р.}	distance from vehicle nose to the c, p.	feet
x _{c.g.}	distance from vehicle nose to the c, g.	feet
ΔX	X _{C. D.} - X _{C. g.}	feet
α	vehicle angle of attack	radians
β	ballistic coefficient = $\frac{W}{C_A S}$	lb/ft^2
Ŷ	reentry flight path angle (always negative)	degrees
ν	kinematic viscosity	ft ² /sec
ρ	atmospheric density	slug/ft ³
ω _A	aerodynamic pitch frequency	rad/sec

- 110 -

, P

-

والمتعار والمتعار والمتعار فالمتعار والمتعار

÷.

THE JOHNS HOPKINS UNIVERSITY APPLIED PHYSICS LABORATORY Bilver Spring Maryland

Subscripts

÷

1 7

١,

ave	refers to average over the time interval
E	refers to reentry conditions (K=0)
i	refers to initial conditions
0	refers to sea level conditions
max	refers to maximum value except as noted
A dot over a	symbol means the first derivative with respect to time.

•• ••• ••

THE JOHNS HOPKING UNIVERSITY APPLIED PHYSICS LABORATORY SILVER SPRING MARYLAND

SUMMARY

When a vehicle moves through an atmosphere, aerodynamic forces are generated that affect vehicle motion. The magnitude of these forces is influenced strongly by the missile velocity and atmospheric density. In this section approximate equations are derived for missile velocity, ambient density (or pressure and temperature), and related parameters.

Assuming that the axial force is the only force acting on the missile, and making other appropriate simplifications, equations are derived for velocity, time, dynamic pressure, longitudinal deceleration, Reynolds number, stagnation point heating rate, and the total heat absorbed at the stagnation point as functions of a parameter (K) which, for a given missile, atmosphere, and trajectory, is defined by altitude only. Most of these quantities increase from zero at reentry, reach a maximum value, and then decrease prior to impact. Equations are given for the maxima and the value of K at which they occur.

An example illustrating the use of these equations is given in Section 4.6, Example 4.

- 113 -

Preceding page blank
THE JOHNS HOPKINS UNIVERSITY APPLIED PHYSICS LABORATORY SILVER SPRING MARYLAND

F

The boost phase of the trajectory is guided to such a high altitude that atmospheric effects during the exit phase are usually negligible. Therefore, this discussion is limited to the reentry phase.

Customarily, reentry into the sensible atmosphere is assumed to begin at an altitude of 400 000 feet. The velocity increases slightly under the effect of gravity, reaches a maximum value, and then decreases under the influence of aerodynamic drag.

The maximum velocity occurs when the axial force is just equal to the component of gravity along the flight path, or:

$$C_A \overline{q} S = -W \sin \gamma$$

Using the definitions of \overline{q} and β , this equation becomes:

$$\rho = -\frac{2\beta \sin \gamma}{V_{\rm max}^2}$$

If the reentry velocity, V_E , is given for some altitude above that corresponding to V_{max} , the maximum velocity may be estimated using the equation:

$$V_{max} = V_E + kg \frac{(h_E - h_{max})}{V_E}$$

where k is a constant having a magnitude between 0 (for $h_E \approx h_{max}$) and 1 (for $h_E >> h_{max}$). A precise value of k is not important since the difference between V_E and V_{max} is small and, in fact, is usually neglected. However, if a little more accuracy is desired, the equations for ρ and V_{max} may be solved by successive approximation.

The remainder of this section considers the motion of the vehicle during the deceleration phase. During this

Preceding page blank - 115 -

period the general behavior of the body may be obtained by simplified equations of motion based on the following assumptions:

- 1. The earth is flat and nonrotating.
- 2. The only force acting on the body is the axial force, and the axial force coefficient is constant.
- 3. The reentry flight path angle is constant.

The reentry phase is of relatively short duration and short range so that the assumption of a flat nonrotating earth is justified for this study. The assumption that the axial force is the only force acting on the body implies that the angle of attack is always zero and that the effect of gravity is negligible compared to the effect of axial force. More precise methods of computing trajectories may be used to show that the effect of α is usually small. Except for the very-high-altitude and, sometimes, the very-low-altitude portions of the reentry trajectory, the assumption that $\frac{W \sin \gamma}{\Delta}$ is small is justified for approximate solutions to the equations of motion. Precise methods of computing trajectories may be used to show that assumption 3 is valid for a large portion of most trajectories and that the validity tends to be improved for high weight and low axial force bodies and for steep reentry angle trajectories. For many reentry bodies and trajectories in current use, the assumption of constant γ is valid all the way to impact.

The geometry of the trajectory in which A is the only force acting is shown in the sketch.

THE JOHNS HOPKINS UNIVERSITY APPLIED PHYSICS LABORATORY Silver Bring Maryland



The equation of motion along the flight path is:

- ·

$$-A = \frac{W}{g} \frac{dV}{dt}$$
 (4.3-1)

Expressing the axial force in terms of coefficients as discussed in Section 2:

$$-C_{A} (1/2) \rho V^{2} S = \frac{W}{g} \frac{dV}{dt}$$
 (4.3-2)

194

Since the flight path is a straight line, $dh/dt = V \sin \gamma$ and Eq. (4.3-2) may be written in the form:

$$\frac{dV}{V} = -\frac{C_A g\rho S}{2W} \frac{dh}{\sin \gamma} . \qquad (4.3-3)$$

The term W/C_AS occurs frequently in reentry equations and is called the ballistic coefficient, β . For an atmosphere in static equilibrium (see Eq. (3-1)), dP = - ρ gdh. When these substitutions are made, Eq. (4.3-3) becomes:

- 117 -

THE JOHNE HORNING UNIVERSITY APPLIED PHYSICS LABORATORY BILVER SPRING MARYLAND

$$\frac{\mathrm{d}V}{\mathrm{V}} = \frac{\mathrm{d}P}{2\beta\,\sin\gamma} \quad (4.3-4)$$

1 1 1 1 1 1 1

Using initial conditions $V = V_i$ at $P = P_i$, Eq. (4.3-4) is readily integrated to give:

$$V = V_{1}e^{\frac{K_{1} - K}{2}}$$
 (4.3-5)

where K is defined by the equation:

$$K = \frac{-P}{\beta \sin \gamma} \quad (4, 3-6)$$

Equation (4.3-5) is valid provided $\frac{e^{K}}{K} \ll \frac{V_{E}^{2}}{2gH}$. This con-

dition is a restatement of assumption 2 listed above. Therefore the equation tends to become invalid at low reentry velocities and at very low and very high values of K. The limitation at very low values of K is not serious and was discussed at the beginning of this section. On the other hand, the limitation at high values of K may have significant consequences (see Section 5.1.2 for example). When the initial altitude is high, $K_i \approx 0$ and the equations may be simplified. For some applications of these equations the initial condition is not a high altitude condition. Therefore for completeness the K_i term is retained in Eq. (4, 3-5) and subsequent equations.

Other trajectory characteristics of interest may be derived (Ref. 4-1) and the results are given in Eqs. (4.3-7)through (4.3-29). The time from altitude h_i (defined by P_i) to any other altitude (defined by P) is given by:

$$t - t_{i} = \frac{-H_{ave}}{V_{E} \sin \gamma} \left[\ln \frac{P}{P_{i}} + I_{1}(K) - I_{1}(K_{i}) \right]$$
(4.3-7)

- 118 -

THE JOHNS HOPKINS UNIVERSITY APPLIED PHYSICS LABORATORY SILVER SPRING MARTLAND

where the I_1 function is plotted in Figs. 4-7 and 4-8. The dynamic pressure is given by the equation:

$$\overline{q} = \frac{-V_i^2 m \sin \gamma}{2 C_A SH} \left[\frac{K}{e^{K - K_i}} \right], \qquad (4.3-8)$$

The longitudinal acceleration is given by:

$$g_{X} = \frac{KV_{i}^{2} \sin \gamma}{2gHc} K K_{i}$$
 (4.3-9)

The Reynolds number based on body length is given by:

$$R_{e} = \frac{V_{i}L}{\nu e^{(K/2-K_{i}/2)}} \qquad (4.3-10)$$

The heating rate at the body stagnation point is indicative of the severity of the heating environment. This rate is given by:

$$q_{s} = \frac{412}{\sqrt{R_{N}\rho_{o}}} \left(\frac{V_{i}}{26\ 000}\right)^{3} \sqrt{\frac{P}{T}e^{3(K_{i}-K)}}.$$
 (4.3-11)

The total heat absorbed at the stagnation point from h_i to h is:

$$Q_{s} - Q_{s_{i}} = \frac{-1.69}{\sqrt{R_{N}\rho_{o}}} \left(\frac{V_{E}}{26\,000}\right)^{2} \frac{\sqrt{T_{ave}}}{s \ln \gamma} \left[\sqrt{F} I_{2}(K) - \sqrt{P_{i}} I_{2}(K_{i})\right] (4.3-12)$$

where $I_{0}(K)$ is plotted in Fig. 4-9.

ŝ

The dynamic pressure, longitudinal deceleration, Reynolds number, and stagnation heating rate all have a

- 119 -

THE JOHNS HOPKING UNIVERSITY APPLIED PHYSICS LABORATORY SILVER SPRING MARYLAND

> trend which increases from zero at reentry to a maximum value and then (depending upon the body and trajectory characteristics) decreases.

> The approximate pressure at which the maximum values of these functions are reached may be obtained by assuming T is a constant, $K_i = 0$, setting the derivative of the function with respect to K to zero, and solving for the maximum. The error introduced by assuming T is a constant is small. At $K_i = 0$, $V_i = V_E$ and the equations for the maxima and their \dot{K} , P, and $V/\dot{V}_{\rm E}$ values are shown below.

Equation	<u>_K</u>	P	$\frac{v/v_E}{E}$	
$\overline{q} = \frac{-\beta V_E^2 \sin \gamma}{0330T}$	1	-¢sinγ	0.608	(4,3-13)
$g_{X} = \frac{V_{E}^{2} \sin \gamma}{9330T}$	1	-gsin Y	0,606	(4,3-14)
$R_{e}/L = 0.368 \frac{V_{E}}{v}$	2	-2β sin γ	0, 388	(4,3-15)

 $q_{g} = \frac{144.3}{\sqrt{P_{0}}} \left(\frac{v_{e}}{26\,000} \right) \sqrt{\frac{-\beta \sin \gamma}{R_{N}T}}$ (4,3-16) $-(\beta/3) \sin \gamma$ 0.846 1/3

The longitudinal acceleration is a characteristic of considerable importance since it is a quantity that can readily be measured on a flight test vehicle. It is interesting to note that for an isothermal atmosphere $g_{X_{max}}$

is highly dependent upon $V_{\mathbf{E}}$, moderately dependent upon γ and T, and independent of the vehicle weight and axial force (β). However the altitude at which $g_{X_{max}}$ occurs is dependent upon β . The greater the β or the steeper the reentry angle, the lower the altitude for gXmax'

- 120 -

THE JOHNS HOPKINS UNIVERSITY APPLIED PHYSICS LABORATORY SILVER SPRING MARTLAND

The g_X versus time curve has two inflection points, which may be determined in the following manner:

$$\dot{\mathbf{g}}_{\mathbf{X}} = \frac{\mathrm{d}\mathbf{g}_{\mathbf{X}}}{\mathrm{d}\mathbf{K}} \frac{\mathrm{d}\mathbf{K}}{\mathrm{d}\mathbf{t}} \quad (4, 3-17)$$

It may be shown that:

$$\frac{\mathrm{d}K}{\mathrm{d}t} = -\frac{\mathrm{V}_{\mathrm{E}} \sin \gamma}{\mathrm{H}} \mathrm{Ke}^{-\mathrm{K}/2} \qquad (4.3-18)$$

and

$$\frac{V_E^3 \sin^2 \gamma}{2g(H)^2} K(1 - K)e^{-3K/2}$$
, (4.3-19)

The inflection point occurs where Eq. (4, 3-19) is maximum or minimum. It may be shown that this occurs at K = 1/3

and K = 2, which correspond to $\frac{g_X}{g_{X_{max}}} = 0.650$ and 0.736, respectively, or P = $-(1/3)\beta \sin\gamma$ and $-2\beta \sin\gamma$, respec-

tively.

Using Eq. (4.3-7), the time between the high altitude inflection point and $g_{X_{\text{max}}}$ is given by:

$$\Delta t_1 = -\frac{80T_{ave}}{V_E \sin \gamma}$$
(4.3-20)

and the time from $g_{X_{\max}}$ to the low altitude inflection point is given by:

$$\Delta t_2 = -\frac{77T_{ave}}{V_E \sin \gamma} \quad . \tag{4.3-21}$$

- 121 -

From Eqs. (4, 3-9) and (4, 3-14);

$$\frac{g_X}{g_{Max}} = 2.72 \frac{K}{e^K}$$
, (4.3-22)

Therefore, using just the telemetry g_X trace, the ratio of $g_X/g_{X_{max}}$ is a good indication of the K history. The K history can be converted to air pressure history from known β and γ (Eq. (4.3-6)), and the pressure history can be converted to altitude history from known atmospheric characteristics. A plot of $g_X/g_{X_{max}}$ versus K is given in Fig. 4-10.

Other quantities required for succeeding sections are listed below for convenient reference.

$$\omega_{A}^{2} = \frac{V_{E}^{2}C_{N} \Delta X m \sin (-\gamma)}{\frac{\alpha}{2C_{A}^{1}Y^{H}}} \frac{K}{e^{K}} \qquad (4.3-23)$$

$$\dot{\omega}_{A} = V_{E}^{2} \left[\frac{C_{N_{\alpha}} \Delta Xm \sin^{3}(-\gamma)}{8 C_{A} I_{Y} H^{3}} \right]^{1/2} \frac{(1 - K) \sqrt{K}}{e^{K}} \quad (4.3-24)$$

$$D = \left[k_1 - C_{m_q} \right] \left[\frac{\sin \left(-\gamma\right)}{2C_A \frac{\Delta X}{d} k_1 \frac{H}{d}} \right]^{1/2} \sqrt{K} \qquad (4, 3-25)$$

$$k_1 = C_{N_{\alpha}} \left(1 - \frac{I_X}{I_Y}\right) \frac{I_Y}{md^2}$$
(4.3-26)

- 122 -

THE JOHNS HOPKINS UNIVERSITY APPLIED PHYSICS LABORATORY BILVER SPRING MARYLAND

$$\frac{1}{q} = \frac{V_{\rm E}^{3} m \sin^{2} \gamma}{2C_{\rm A}^{3} {\rm SH}^{2}} \frac{(1-K) K}{e^{3K/2}}$$
(4.3-27)

$$\frac{d\overline{q}}{dK} = \frac{V_E^2 \min(-\gamma)}{2C_A SH} \frac{(1 - K)}{e^K}$$
(4.3-28)

$$\dot{P} = \frac{V_E W \sin^2 \gamma}{C_A SH} \frac{K}{e^{K/2}}$$
 (4.3-29)

The 1962 Standard Atmosphere characteristics required for evaluating these equations are listed in Table 3-1.

.....

L. C. C. Phys. Rev.

THE JOHNS HOPKINS UNIVERSITY APPLIED PHYSICS LABORATORY SILVER SPRING MARYLAND

С. С



Fig. 4-7 VARIATION OF (1 (K) WITH K

Preceding page blank

- 125 -

PHECEDING PAGE BLANK -NOT FILMED

,÷

1. Walter and the

ä

- 1



Fig. 4-8 VARIATION OF I1 (K) WITH K

į.

the properties of the the p

- 126 -

THE JOHNS HOPKING UNIVERSITY APPLIED PHYSICS LABORATORY SILVER SPRING MARYLAND

1.

14.14.14.1



٠

and and the second second second second second second second second second second second second second second s





.

. . ,

THE JOHNE HOPKINE UNIVERSITY APPLIED PHYSICS LABORATORY SILVER SPRIME MARYLAND

į

计无可能 的复数加速度 化乙基乙基乙基

:



- 128 -

!

٠

ł

i.

j i

) · :

1

;

1 L

Ĩ

Ì

1

t

THE JOHNE HOPKING UNIVERSITY APPLIED PHYSICS LABORATORY Silver Spring. Maryland

4141

Ι

1

I

1 З,

4.

-

¥ ...

u -

••• w w **

-

Î

ľ

2

4.4 EFFECTS OF BODY DYNAMICS

ميم ,

and a set the course

conduct the second states at

.

3

. سار

State and State of State

÷

- 129 -

112000022

......

. .

CHC:

The second second second

THE JOHNE HOPKINE UNIVERSITY APPLIED PHYSICS LABORATORY SILVER SPHINE MARVLAND

SYMBOLS FOR SECTION 4.4

Symbol	Definition	Typical Units
a	constant defined by Eq. $(4, 4-52)$	-
C ₁ , C ₂	constants in Eq. (4,4-33)	-
C ₃ , C ₄ , C ₅	constants defined by Eqs. (4.4-45), (4.4-58), and (4.4-61)	-
c _A	axial force coefficient	-
c _N	normal force coefficient	-
C _N	$d\mathbf{C}_{\mathbf{N}}/d\overline{\mathbf{\alpha}}$	-
C,	roll damping derivative	-
c _m	pitch damping derivative	-
d d	reference length	feet
^F x' ^F y' ^F z	aerodynami'c forces along body axes X, Y, Z	pounds
Fα	C _{Nα} qs	lb/rad
g	acceleration of gravity	ft/sec^2
gL	load factor = N/W	-
H	scale height = RT/g or	feet
	angular momentum	slug-ft ² /sec
^I x, ^I y, ^I z	moments of inertia about principal axes X, Y, Z	slug-ft ²
Jo	zero order Bessel function	-
к	$-\mathbf{P}/(\beta \sin \gamma)$	-
^k 1	constant defined by Eq. (4.4-30)	rad/sec
k ₂	constant defined by Eq. (4.4-31)	rad/sec^2

Preceding page blank

1

- 131 -

THE JOHNS HOPKING LINIVERSITY APPLIED PHYSICS LABORATORY BILVER BPRING MARTLAND

Symbol	Definition	Typical Units
l, m, n	aerodynamic rolling moment, pitching moment, and yawing moment, respec- tively	lb-ft
122	mass = W/π	slugs
111	шшаа ~ м/g	DIKED
Mp	roll damping factor, $C_{\ell_p} \frac{\overline{qSd^2}}{V}$	lb-ft-sec
M q	pitch damping factor, $C_{m_q} \frac{\overline{qSd}^2}{V}$	lb-ft-sec
Mr	yaw damping factor, $C_{n_r} \frac{\overline{q_{Sd}}^2}{V}$	lb-ft-sec
N	total lateral force, $\sqrt{F_Y^2 + F_Z^2}$	pounds
Р	ambient air pressure	$1b/ft^2$
p, q, r	roll rate, pitch rate, yaw rate, respectively	rad/sec
q	dynamic pressure	$1b/ft^2$
R	gas constant = 1716	ft ² /sec ² - °R
S	reference area	ft^2
Т	temperature	°R
t	time	seconds
u, v, w	missile velocity along X, Y, Z axes, respectively	ft/sec
v	missile velocity	ft/sec
W	missile weight	pound s
x	constant defined in Eq. (4.4-40)	-
ΔX	X - X	feet
x _{c.g.}	distance from missile nose to the c.g.	feet

1111

51.2

「夏山子」及出すべ

ł

1

CIZE AND A CRAME

- 132 -

THE JOHNS HOPKING UNIVERSITY APPLIED PHYSICS LABORATORY BILVER BRING MARYLAND

Symbol	Definition	Typical Units
X C. D.	distance from missile nose to the c, p,	feet
α	angle of attack in the XZ plane	radians
ā	vector angle of attack $ \vec{\alpha} = \sqrt{\alpha^2 + \beta^2}$	radians
ß	angle of attack in the XY plane	radians
	ballistic coefficient, W/C _A S	$1b/ft^2$
γ	flight path angle, negative for re- entry trajectories	degrees
¢	angle defined by Eq. $(4, 4-22)$	degrees
Ø	angle defined by Eq. $(4, 4-23)$	degrees
θ	angle between the missile X-axis and the local horizontal	degrees
ω _g	frequency defined by Eq. (4.4-21)	rad/sec
ω _T	transverse rate = $\sqrt{q^2 + r^2}$	rad/sec
ω	total spin vector = $\sqrt{p^2 + q^2 + r^2}$	rad/sec
ω _A	aerodynamic frequency defined by Eq. (4.4-36)	rad/sec
ω	frequency defined by Eq. (4.4-49)	rad/sec
τ _B	period defined by Eq. $(4, 4-24)$	seconds
$\tau_{\rm S}$	period defined by Eq. (4.4-25)	seconds
ψ	phase angle	degrees

. 2

4

ŝ,

N,

Subscripts

E	refers to	conditions at entry	
0	refers to	conditions at $t = 0$	

A dot over a symbol means the first derivative with respect to time. Two dots over a symbol mean the second derivative with respect to time.

- 133 -

THE JOHNS HOPKING UNIVERSITY APPLIED PHYSICS LABORATORY BILVER SPRING MARYLAND

SUMMARY AND CONCLUSIONS

In the preceding section, the effects of the atmosphere on vehicle velocity and related factors were considered. In the present section, attention is given to the effects of the atmosphere on the dynamic behavior of the vehicle. We are concerned here with the rotational motion of the vehicle about its center of gravity over relatively short periods of time. Six-degree-of-freedom equations of motion are presented and interpreted in terms of body motion characteristics in the absence of aerodynamic forces (for example, motion in the exoatmosphere, where gyroscopic motion is important) at low altitudes where aerodynamic forces dictate the motion and at high altitudes where aerodynamic and gyroscopic forces are both important. Dynamic motion is important in the analysis of body heating and lateral loads. Equations are given for body rotational frequencies that are observed on telemetry traces from on-board equipment or from tracking radar signal strength traces. Numerical illustrations are given in Examples 3 and 5, Section 4, 6,

The major conclusions are:

1. In the absence of aerodynamic forces, a reentry-type body that is spun about its longitudinal axis will follow a motion pattern determined by conditions at the initiation of the spin. If spin is applied so that neither pitch nor yaw rate is introduced, the vehicle will maintain a fixed attitude in space. If a pitch and/or a yaw rate (sometimes the vector sum is called the transverse rate) is experienced when spin is introduced, the missile's longitudinal axis traces a cone whose apex is at the missile center of gravity. The magnitude of the cone angle and the period of motion are functions of the transverse rate, missile roll rate, and moments of inertia about the longitudinal and transverse axes.

1

Preceding page blank

- 135 -

THE JOHNS HOPKINS UNIVERSITY APPLIED PHYSICS LABORATORY SILVER SPRING MARYLAND

> 2. If the body contains rate gyros, the body rate observed on the telemetry trace during the exoatmospheric portion of flight will show a sinusoidal trace. The half-amplitude is the transverse rate existing at the time of spin-up, and the period is a function of the roll rate and the moments of inertia about the longitudinal and transverse axes.

3. During the low altitude portion of the reentry phase, aerodynamic forces dictate the motion of the vehicle. Generally, a missile reenters the atmosphere with some initial angle of attack. To prevent excessive lateral loads and heating problems the missile is designed so that the aerodynamic forces cause the angle of attack to approach zero rapidly. It is shown that for a symmetrical missile the angle of attack follows a damped oscillatory motion provided certain static and dynamic stability criteria are met. The frequency of the oscillation is called the aerodynamic frequency. This frequency increases from zero at reentry, reaches a maximum, and then (depending upon the trajectory and reentry conditions) decreases until impact.

4. At high altitudes, both aerodynamic and gyroscopic moments are important. 'The gyroscopic moments tend to maintain the body at a fixed altitude; the aerodynamic moments tend to rotate the body so that zero angle of attack is maintained. Typically, gyroscopic motion predominates at altitudes above about 300 000 feet; aerodynamic moments predominate at altitudes below about 100 000 feet; both effects are important at the intermediate altitudes.

- 136 -

In Section 4.3 the body was constrained to motion along a straight-line flight path. Therefore the equations of motion were reduced to a one-degree-of-freedom problem. In this section, we consider the six-degree-of-freedom problem. However, the equations are simplified by making the following assumptions:

- 1. The effect of gravity will be neglected.
- 2. The body is assumed to have symmetry about the longitudinal axis.

The primary interest in this section is to consider how the body behaves over a relatively short time span. For this purpose the effect of gravity may be shown to be very small. By assuming the body to be symmetric about the longitudinal axis, the principal axes coincide with the body axes, and the equations of motion are simplified since the product-of-inertia terms are zero.

The equations of motion are presented in a body axis system (Fig. 4-11) which remains fixed with respect to the body. The body rate terms in the equations of motion are the rates sensed by rate gyros mounted in a vehicle; the body translational acceleration terms are those that would be measured by accelerometers mounted at the vehicle center of gravity. The six equations defining the six degrees of freedom are:

$$\mathfrak{P}_{\mathbf{X}} = \ell - qr(\mathbf{I}_{\mathbf{Z}} - \mathbf{I}_{\mathbf{Y}})$$
 (4.4-1)

$$q_{I_{Y}} = m - pr(I_{X} - I_{Z})$$
 (4.4-2)

$${}^{h}I_{Z} = n - pq(I_{Y} - I_{X})$$
 (4.4-3)

$$\dot{u} = \frac{F_X g}{W} + vr - wq \qquad (4.4-4)$$

- 137 -

THE JOHNS HOFKINS UNIVERSITY APPLIED PHYSICS LABORATORY SILVER SPRING MARYLAND

$$\dot{v} = \frac{\mathbf{F}_{\mathbf{Y}}\mathbf{g}}{\mathbf{W}} + \mathbf{w}\mathbf{p} - \mathbf{u}\mathbf{r}$$
(4.4-5)

$$\dot{w} = \frac{F_Z g}{W} + uq - vp$$
, (4.4-6)

The auxiliary equations required are:

$$\boldsymbol{\ell} = M_{p} p \qquad (4, 4-7)$$

$$m = F_Z \Delta X + M_q q \qquad (4, 4-8)$$

$$\mathbf{n} = -\mathbf{F}_{\mathbf{Y}} \Delta \mathbf{X} + \mathbf{M}_{\mathbf{r}} \mathbf{r}$$
 (4.4-9)

$$V = \sqrt{u^2 + v^2 + w^2}$$
 (4, 4-10)

$$\tan \overline{\alpha} = \frac{\sqrt{v^2 + w^2}}{u}$$
 (4.4-11)

$$\mathbf{F}_{\mathbf{X}} = -\mathbf{C}_{\mathbf{A}} \overline{\mathbf{q}} \mathbf{S} \tag{4.4-12}$$

$$\mathbf{F}_{\mathbf{Y}} = -\mathbf{F}_{\boldsymbol{\alpha}} \overline{\boldsymbol{\alpha}} \quad \frac{\mathbf{v}}{\sqrt{\mathbf{v}^2 + \mathbf{w}^2}} = -\mathbf{F}_{\boldsymbol{\alpha}} \boldsymbol{\beta}$$
(4.4-13)

$$F_{Z} = -F_{\alpha}\overline{\alpha} \quad \frac{w}{\sqrt{v^{2} + w^{2}}} = -F_{\alpha}\alpha \qquad (4.4-14)$$

$$\mathbf{F}_{\alpha} = \left(\frac{\mathbf{C}_{N} \overline{\mathbf{q}} \mathbf{S}}{\overline{\alpha}}\right) = \mathbf{C}_{N} \overline{\mathbf{q}} \mathbf{S} \quad . \tag{4.4-15}$$

- Lest

- 138 -

i

,1

ì

ł

No.

4, 4, 1 Zero Aerodynamic Forces

Consider first the case of flight in the exontmosphere where all aerodynamic terms are zero. In this region only gyroscopic effects exist.

Since the body is assumed to be symmetrical, $I_Z = I_Y$ and Eq. (4.4-1) reduces to the condition where p equals a constant. Differentiation of Eq. (4.4-2) gives:

$$a + pt \frac{(I_X - I_Z)}{I_Y} = 0$$
 (4.4-16)

Substituting for f from Eq. (4.4-3) and noting that $I_Z = I_V$:

$$a + p^2 \left(\frac{I_Y - I_X}{I_Y}\right)^2 q = 0$$
 (4.4-17)

The solution to this differential equation is:

$$q = \omega_T \cos \left[p(1 - \frac{I_X}{I_Y}) t + \psi \right]$$

If time is measured from the time when $q = q_{max}$, then $\psi = 0$ and

 $q = \omega_T \cos p(1 - \frac{I_X}{I_Y}) t$ (4.4-18)

$$\dot{q} = -\omega_T p(1 - \frac{I_X}{I_Y}) \sin p(1 - \frac{I_X}{I_Y}) t$$

- 139 -

THE JOHNS HOPKINS UNIVERSITY APPLIED PHYSICS LABORATORY BILVER SPRING MARYLAND

Substituting into Eq. (4, 4-2):

$$r = -\omega_T \sin p(1 - \frac{I_X}{I_Y}) t$$
 (4.4-19)

$$(4.4-20)$$

Therefore, the pitch or yaw rate measured by a rate gyro mounted on a vehicle is a sinusoidal oscillation of halfamplitude $\omega_{\rm T}$ having a frequency:

$$\omega_{g} = p(1 - \frac{I_{X}}{I_{Y}})$$
 (4.4-21)

·

This equation is convenient for checking the calibration constant for the roll rate gyro in flight test work. The term $\omega_{\rm T}$ is sometimes called the body transverse rate.

The solution to the equations of motion in body axes provides information regarding the time history of quantities that may be sensed by "on-board" instrumentation, but no information is provided on the motion of the vehicle in space.

The motion in space is obtained by solving the equations of motion written in the inertial axis system. It is shown (Ref. 4-2) that, in the absence of aerodynamic forces, the motion of the body may be described physically as a "body cone" rolling along a "space cone" (or "cone of precession"), as shown in the sketch, page 141. THE JOHNE HOPKING UNIVERSITY APPLIED PHYSICS LABORATORY SILVER SPRING MARYLAND

1

Ň



THE PROPERTY OF

The centerline of the space cone is the angular momentum vector which remains fixed in space. The centerline of the body cone is the vehicle's longitudinal axis of symmetry (roll rate axis). The semicone angle of the body cone is defined by the equation:

$$\tan \epsilon = \frac{\omega_{\rm T}}{\rm p} \tag{4.4-22}$$

The angle between the body centerline and the centerline of the space cone is constant and is given by:

$$\tan \phi = \frac{\omega_{\rm T}}{p} \frac{I_{\rm Y}}{I_{\rm X}}$$
(4.4-23)

where $\omega_{\rm T}$ is defined by Eq. (4.4-20). It should be noted that the magnitude of the total spin vector, $\omega = \sqrt{p^2 + q^2 + r^2}$,

- 141 -

Sec. Sec. Mary

is constant and lies along the space cone surface. Assuming a positive roll rate and $I_X < I_Y$, the body centerline rotates about H in a clockwise direction (as shown above) when viewed by an observer looking toward +H. The transverse rate vector rotates about the body centerline so that the period is given by:

$$\tau_{\rm B} = \frac{2\pi}{(1 - \frac{I_{\rm X}}{I_{\rm Y}})p}$$
(4.4-24)

and corresponds to the frequency given previously in Eq. (4.4-21). The body centerline precesses about the space cone with a period:

$$\tau_{\rm g} = \frac{2\pi}{\frac{I_{\rm X}}{I_{\rm Y}} p \sqrt{1 + \tan^2 \phi}} \qquad (4.4-25)$$

For recentry bodies of current design, $\tau_{\rm S}$ is an order of magnitude greater than $\tau_{\rm B}$.

The utility of the physical picture described above may be illustrated by the following example. Suppose we wish to determine the coning motion for a specified set of conditions at reentry. In particular, select a condition at t = 0 where $\overline{\alpha} = \overline{\alpha_0}$ and lies in the body XZ plane, which is the plane of the trajectory. Consider first the case of $q_0 = 0$ and $r = r_0$, where r_0 is positive and may be represented by a vector along the +Z body axis. Since ω lies on the surface on the space cone and the body cone is external to the space cone, the angular momentum vector lies in the trajectory plane and $\overline{\alpha}$ varies from $\overline{\alpha_0}$ (a maximum value) to $\overline{\alpha_0} - 2\emptyset$ (a minimum value). Similarly, if r_0 is negative, $\overline{\alpha}$ varies from $\overline{\alpha_0}$ (a minimum value) to a maximum value of $\overline{\alpha_0} + 2\emptyset$. The body-fixed Y-axis at t = 0 is directed toward the right of the trajectory as viewed by an observer looking toward the impact point. Therefore for an initial condition of pitch rate only, if q_0 is positive.

- 142 -

THE JOHNS HOPKING UNIVERSITY APPLIED PHYSICS LABORATORY BILVER OPRING MARYLAND



the angular momentum vector is directed to the right of the trajectory so that the nose of the vehicle remains in or to the right of the trajectory plane. Similarly, if q_0 is negative, the vehicle nose is always in or to the left of the trajectory plane.

4.4.2 Zero Gyroscopic Forces

Consider next the case where the aerodynamic terms are much greater than the gyroscopic terms (body rate terms). In addition we will make the following assumptions:

- 1. The trajectory region of interest is over a very short time period so that the dynamic pressure is constant.
- 2. The roll rate is zero.
- 3. The pitching motion is planar (i.e., only pitch or only yaw motion).
- 4. $\overline{\alpha}$ is small so that the use of linearized derivatives is valid, and $\cos \overline{\alpha} \approx 1$; $\sin \overline{\alpha} \approx \overline{\alpha}$.

Suppose that at some point along the trajectory the vehicle is perturbed to some $\alpha = \alpha_0$. We wish to study the

- 143 -

THE JOHNE HOPKING UNIVERSITY APPLIED PHYSICS LABORATORY Silver Spring Maryland

17 1 16- 1

> resulting motion of the vehicle under the influence of just aerodynamic forces. The geometry of the trajectory is shown in the diagram.



From Eqs. (4.4-2) and (4.4-8):

 $qI_Y = F_Z \Delta X + M_q q$.

From the geometry of the trajectory:

$$\begin{array}{c} \theta = \gamma + \overline{\alpha} \\ \dot{\theta} = q = \dot{\gamma} + \dot{\overline{\alpha}} \\ \ddot{\theta} = \dot{q} = \dot{\gamma} + \ddot{\overline{\alpha}} \end{array}$$

$$(4.4-26)$$

From Eqs. (4.4-6) and (4.4-26), using a small α approximation, and noting that $u = V \cos \alpha$ and $w = V \sin \alpha$, it may be shown that:

$$\mathcal{P} = -\frac{F_Z g}{WV} \qquad (4, 4-27)$$

- 144 -

Sec. Barren Same

THE JOHNS HOPKINS UNIVERSITY APPLIED PHYSICS LABORATORY SILVER SPRING MARYLAND

Using the linear relationship between \mathbf{C}_N and $\overline{\boldsymbol{\alpha}};$

$$\begin{array}{c} \mathbf{F}_{Z} = -\mathbf{C}_{N} \overline{\alpha} \overline{\mathbf{q}} \mathbf{S} \\ \mathbf{F}_{Z} = -\mathbf{C}_{N} \overline{\alpha} \overline{\mathbf{q}} \mathbf{S} \\ \mathbf{F}_{Z} = -\mathbf{C}_{N} \overline{\alpha} \overline{\mathbf{q}} \mathbf{S} \end{array} \right)$$

$$\begin{array}{c} (4.4-28) \\ \mathbf{F}_{Z} = -\mathbf{C}_{N} \overline{\alpha} \overline{\mathbf{q}} \mathbf{S} \end{array} \right)$$

Using Eqs. (4, 4-25) through (4, 4-28), it may be shown that the body motion is defined by the differential equation:

$$\frac{\dot{\alpha}}{\dot{\alpha}} + k_1 \frac{\dot{\alpha}}{\dot{\alpha}} + k_2 \overline{\alpha} = 0 \qquad (4.4-29)$$

where

$$\kappa_{1} = \frac{C_{N_{q}} \overline{q}S}{\frac{W}{g} V} - \frac{C_{m_{q}} \overline{q}Sd^{2}}{I_{Y}V}$$
(4, 4-30)

$$k_{2} = \frac{C_{N} \Delta X \overline{q} S}{I_{Y}} - \frac{C_{m} C_{N} (\overline{q} S d)^{2}}{\frac{W}{g} V^{2} I_{Y}} . \qquad (4.4-31)$$

For bodies of interest, the C_{mq} term of Eq. (4, 4-31) is very small compared to the $C_{N\alpha}$ term, and the equation for k_2 becomes:

$$\kappa_2 = \frac{C_N \Delta X \overline{q} S}{I_Y} , \qquad (4, 4-32)$$

- 145 -

11

For statically stable vehicles, k_2 is positive, and the solution to Eq. (4.4-29) is:

$$\overline{\alpha} = e^{-\frac{k_1}{2}t} \left[C_1 \sin t \sqrt{k_2 - \frac{k_1^2}{4}} + C_2 \cos t \sqrt{k_2 - \frac{k_1^2}{4}} \right] \quad (4.4-33)$$

where C_1 and C_2 are arbitrary constants defined by the initial conditions. Let $\overline{\alpha} = \overline{\alpha}_0$ and $\overline{\alpha} = 0$ at t = 0 so that $C_2 = \overline{\alpha}_0$ and $C_1 = 0$ and Eq. (4, 4-33) becomes:

$$\overline{\alpha} = \overline{\alpha}_0 e^{-\frac{k_1}{2}} \cos t \sqrt{k_2 - \frac{k_1^2}{4}} \qquad (4.4-34)$$

Therefore, for a statically stable vehicle, the $\overline{\alpha}$ history is oscillatory and is damped (converges to zero) provided $k_1 > 0$. If $k_1 < 0$, $\overline{\alpha}$ diverges even though the vehicle is statically stable. Some flight tests of reentry vehicles have been observed to exhibit this type of $\overline{\alpha}$ divergence. To assure $\overline{\alpha}$ convergence, then:

The frequency of the oscillation is given by the equation:

$$\omega_{A} = \sqrt{k_2 - \frac{k_1^2}{4}}$$

- 146 -

THE JOHNS HOPKINS UNIVERSITY APPLIED PHYSICS LABORATORY BUYER SPRING MARLAND

For most bodies $k_2 \gg \frac{k_1^2}{4}$ and the frequency becomes:

$$\omega_{A} = \sqrt{\frac{C_{N} \Delta X \overline{q} S}{I_{Y}}} \quad . \tag{4.4-36}$$

This frequency, called the aerodynamic pitch frequency, is of great importance in the motion of reentry bodies as will be shown later. It should be noted that ω_A is zero at reentry ($\overline{c_1} = 0$) and, for constant C_{N_Q} and

 ΔX , ω_A is proportional to $\sqrt{\overline{q}}$.

A solution to the equations of motion (Ref. 4-3), in which the dynamic pressure was allowed to vary, shows only small differences from the results given above except for very blunt bodies and nearly zero values of ΔX .

The aerodynamic frequency may be evaluated as a function of altitude (T, K) using equations derived in Section 4.3. From Eqs. (4.3.8) and (4.4.36):

$$\omega_{A}^{2} = \frac{V_{E}^{2}C_{N}\Delta Xm \sin(-\gamma)}{\frac{\alpha}{2C_{A}I_{Y}H}} \frac{K}{e^{K}}$$
(4.4-37)

4.4.3 Nonzero Gyroscopic and Aerodynamic Forces

At very high altitudes the dynamic motion of the body is governed by the gyroscopic terms in the equations of motion; at low altitudes where the dynamic pressure is high, the aerodynamic forces predominate. We look next at the transition from one region to the other.

Using the equations of Ref. 4-3 and the reentry characteristics given in Section 4.3, it may be shown that for p = q = r = 0 at reentry the α history is defined by the equation:

- 147 -

THE JOHNS HOPKINS UNIVERSITE APPLIED PHYSICS LABORATORY BILVER SPRING MARYLAND

$$\frac{\overline{\alpha}}{\overline{\alpha}_{\rm E}} = \frac{J_{\rm o} \left[-\frac{2 \omega_{\rm A} H}{V_{\rm E} \sin \gamma} e^{\frac{K}{2}} \right]}{\exp \left[\frac{K}{4C_{\rm A}} (C_{\rm N} - C_{\rm m_q} \frac{{\rm md}^2}{{\rm I}_{\rm Y}} - 2C_{\rm A}) \right]}$$
(4.4-38)

where $J_0(X)$ is the zero order Bessel function of the first kind plotted in Fig. 4-12. Writing ω_A in terms of K using Eq. (4, 4-37):

$$\frac{\overline{\alpha}}{\overline{\alpha}_{\rm E}} = \frac{J_{\rm c} \left(\sqrt{-\frac{2 \rm Hm}{\sin \gamma}} \frac{C_{\rm N} \Delta X}{C_{\rm A} I_{\rm Y}} K \right)}{\exp \left[\frac{K}{4C_{\rm A}} \left(C_{\rm N} - C_{\rm m} \frac{md^2}{I_{\rm Y}} - 2C_{\rm A} \right) \right]} \cdot (4, 4-39)$$

Therefore, for a given vehicle and trajectory, the $\overline{\alpha}$ history is a function of P and T only and P is the predominant factor. Even without the exponential damping function in the denominator of Eq. (4, 4-39), the $\overline{\alpha}$ history is a damped oscillation about $\overline{\alpha} = 0$, and the frequency is the aerodynamic frequency, ω_A . For conditions near entry ($\overline{\alpha}_E \approx 1$), $K \approx 0$ so that the exponential term in Eq.

 $(4, 4-39) \approx 1$, and writing K in terms of P:

$$\frac{\overline{\alpha}}{\overline{\alpha}_{\rm E}} \approx J_{\rm o} \left(\sqrt{\frac{2 {\rm SH}}{g \sin^2 \gamma}} \frac{C_{\rm N} \Delta X}{\alpha}_{\rm Y} P \right) = J_{\rm o}(X) \quad (4.4-40)$$

Equation (4.4-40) may be used to determine the altitude (P) at which the atmosphere begins to affect the body motion. Arbitrarily selecting $\alpha/\alpha_E = 0.99$ as the condition when the atmosphere begins to affect the body motion, from the Bessel function tables, the argument, X, corresponding to $\overline{\alpha}/\overline{\alpha}_E = 0.99$ is 0.2.

- 148 -

THE JOHN'S HOPKING UNIVERSITY APPLIED PHYSICS LABORATORY SILVER BERING MARYLAND

$$P = \frac{0.02 I_{y}g \sin^2 \gamma}{SHC_{N_{\alpha}} \Delta X} \qquad (4.4-41)$$

The altitude for the first peak in $\overline{\alpha}$ corresponding to X = 3.83 shown in Fig. 4-12 may be obtained from:

$$P = \frac{7.34 I_Y g \sin^2 \gamma}{SHC_N \Delta X} \qquad (4.4-42)$$

The $\overline{\alpha}$ envelope could be obtained directly from $J_0(X)$ tables; however, for $\overline{\alpha}/\overline{\alpha}_E$ less than about 0.8, $J_0(X)$ is approximated quite closely by the equation:

$$J_{O}(X) = \frac{\cos{(\frac{\pi}{4} - X)}}{\sqrt{\frac{\pi}{2} X}} \qquad (4.4-43)$$

Therefore, the envelope corresponding to Eq. (4.4-38) is:

$$\frac{1}{\alpha_{\rm E}} = \frac{\sqrt{\frac{-V_{\rm E} \sin \gamma}{\pi \omega_{\rm A} H}}}{\frac{C_{\rm 3} K}{e}}$$
(4.4-44)

where

$$C_3 = \frac{1}{4} \left(\frac{C_N}{C_A} - \frac{C_m}{C_A} \frac{md^2}{I_Y} - 1 \right)$$
 (4.4-45)

and is primarily the aerodynamic damping term. At high altitudes where e^{K} and $e^{-K} \approx 1$:

- 149 -

THE JOHNS HOPKING UNIVERSITY APPLIED PHYSICS LABORATORY SILVER SPRING MARYLAND

j J

$$\frac{\overline{\alpha}}{\overline{\alpha}_{\rm E}} = \left[\frac{2gI_{\rm Y}\sin^2\gamma}{\pi^2 {\rm SHC}_{\rm N}\Delta {\rm XP}}\right]^{1/4} . \qquad (4.4-46)$$

The α convergence at high altitudes is a function of body mass characteristics (I_Y, S, X_{c,g}), aerodynamic characteristics (C_{N $\alpha'}, X_{c,p}), type of trajectory (<math>\gamma$), and atmospheric properties (T, P). For a given trajectory, mass, and aerodynamic characteristics, α is inversely proportional to $\sqrt[4]{TP}$ or approximately to $\sqrt[4]{P}$. For a given mass, aerodynamic characteristics, and altitude, the α convergence is proportional to $\sqrt{\sin \gamma}$. That is, at a given altitude, the α for a shallow trajectory will have converged more than that for a steep trajectory. Increasing $C_{N_{\alpha'}}$ and ΔX tends to increase the convergence at a given altitude: increasing I_Y/S tends to decrease the convergence.</sub>

If $p \neq 0$ at reentry, the early α history is changed. The gyroscopic terms tend to maintain the body motion about the angular momentum vector, which is fixed in space and usually is not directed along the velocity vector. The aerodynamic terms tend to cause oscillation about the velocity vector. As a result, in the initial phase of reentry, α oscillates about an angle of attack which, itself, is gradually reduced as the aerodynamic forces increase. The equations defining α and β for the case of $\omega_{\rm T} = 0$ are (Ref. 4-4):

α	= A		14 4 45
₿		A sin wt	(4.4-4)

)

- 150 -

$$A = \frac{\overline{\alpha}_{E} \left[\sqrt{\omega_{A}^{2} + \left(\frac{1}{2} \frac{I_{X}}{I_{Y}} p\right)^{2} + \frac{1}{2} \frac{I_{X}}{I_{Y}} p} \right]}{2\sqrt{\omega_{A}^{2} + \left(\frac{1}{2} \frac{I_{X}}{I_{Y}} p\right)^{2}}}$$
(4.4-48)

$$\overline{\omega} = \sqrt{\omega_{\rm A}^{2} + \left(\frac{1}{2}\frac{I_{\rm X}}{I_{\rm Y}}p\right)^{2} - \frac{1}{2}\frac{I_{\rm X}}{I_{\rm Y}}p} \qquad (4, 4-49)$$

In the initial phase of reentry where $\omega_A \ll \frac{1}{2} \frac{I_X}{I_Y}$ p, the α envelope is given by:

$$\frac{\overline{\alpha}}{\overline{\alpha}_{\rm E}} \approx \left[1 - \left(\frac{\omega_{\rm A}}{\frac{I_{\rm X}}{I_{\rm Y}}} \right)^2 \right] , \qquad (4.4-50)$$

Where ω_A is not small compared to $\frac{1}{2} \frac{I_X}{I_Y} p$, it may be shown (Ref. 4-5) that the ratio $\overline{\alpha}$ with roll to $\overline{\alpha}$ without roll is given by the equation:

$$\frac{\alpha_{\rm p}}{\alpha_{\rm p=0}} = \sqrt{\frac{a}{\tanh a}}$$
(4.4-51)

$$a = \frac{-\pi}{2} \frac{HpI_X}{V_E I_Y \sin \gamma} \qquad (4.4-52)$$

For values of a greater than about 3, tanh a = 1, and the α history corresponding to Eq. (4.4-44) is:

- 151 -

 $\gamma := \gamma : f \to \gamma : f \to \gamma : f$

Ś.,

THE JOHNS HOPKINE UNIVERSITY APPLIED PHYSICS LABORATORY SILVER SPRING MARYLAND

$$\frac{\overline{\alpha}}{\overline{\alpha}_{\rm E}} = \frac{\sqrt{\frac{1}{2} \frac{\mathbf{I}_{\rm X}}{\mathbf{I}_{\rm Y}} \frac{\mathbf{p}}{\omega_{\rm A}}}{\frac{\mathbf{C}_{\rm 3} \mathbf{K}}{\mathbf{e}}} \cdot (4, 4-53)$$

Therefore, at high altitudes where $e \approx 1$, the $\overline{\alpha}$ envelope converges to $\overline{\alpha} = 0$ in proportion to $\sqrt{p/\omega_A}$.

For a nonrolling vehicle at reentry, $\overline{\alpha}$ begins to decrease when the aerodynamic moment is sufficiently high, and this altitude can be estimated using Eq. (4.4-41). For a rolling vehicle, $\overline{\alpha}$ begins to decrease when the aerodynamic moment is significant relative to the gyroscopic moment. This altitude may be estimated using Eq. (4.4-50)

and solving for ω_A at $\frac{\overline{\alpha}}{\overline{\alpha}_E} = 0.99$, so that:

$$\frac{\omega_{\rm A}}{\rm p} = 0.1 \frac{{\rm I}_{\rm X}}{{\rm I}_{\rm Y}} \quad . \tag{4.4-54}$$

Substituting for ω_A^2 using Eq. (4.4-37) and assuming $e^{K} = 1$;

$$\rho = \frac{P}{RT} = \frac{0.02I_X^2}{I_Y C_{N_A} \Delta X S V_E^2} p^2 . \qquad (4.4-55)$$

A rapid convergence in $\overline{\alpha}$ is desirable in order to prevent excessive lateral loads resulting from the $\overline{\alpha_E}$. The lateral load factor is given by:

$$\mathbf{g}_{\mathrm{L}} = \frac{\mathrm{N}}{\mathrm{W}} = C_{\mathrm{N}} \frac{\overline{\alpha}}{\alpha_{\mathrm{E}}} \frac{\mathrm{S}}{\mathrm{W}} \overline{\mathrm{q}} \frac{\overline{\alpha}}{\overline{\alpha_{\mathrm{E}}}} \qquad (4.4-56)$$

5.21%

- 152 -

THE JOHNS HOPKING UNIVERSITY APPLIED PHYSICS LABORATORY SILVER SPRING MARYLAND

> For a nonrolling vehicle, using $\alpha/\alpha_{\rm E}$ from Eq. (4.4-44) and \bar{q} from Eq. (4.3-8), it may be shown that the maximum envelope of $g_{\rm L}$ is given by:

$$\frac{g_{1}}{\overline{\alpha}_{E}} = -\frac{V_{E}^{2} \sin \gamma}{gH} \left[\frac{1}{\frac{1}{8\pi^{2}}} \left(\frac{C_{N}}{C_{A}} \right)^{3} \left(\frac{I_{Y}}{md^{2}} \right) \left(\frac{d^{2}}{\Delta X H} \right) \right]^{1/4} \frac{K^{3/4}}{C_{4}R} = (4, 4-57)$$

$$C_{4} = \frac{1}{2} \left[1 + \frac{1}{2} \left(\frac{C_{N_{\alpha}}}{C_{A}} - \frac{C_{m_{q}}}{C_{A}} - \frac{md^{2}}{T_{Y}} \right) \right] \cdot C_{3} + 0.75 , \quad (4.4-58)$$

Neglecting the small variation in H with K, the maximum occurs at:

$$K = \frac{3}{4} \frac{1}{C_4} \cdot (4.4-59)$$

Using Eq. (4.4-59) and substituting for the constants in Eq. (4.4-57), the magnitude of $(g_L/\overline{\alpha}_E)$ is given by:

$$\left(\frac{\mathbf{g}_{\mathrm{L}}}{\overline{\sigma}_{\mathrm{E}}}\right)_{\mathrm{max}} \approx 0.128 \frac{\mathrm{V}_{\mathrm{E}}^{2}}{\mathrm{g}\mathrm{H}} (-\mathrm{gln}\,\gamma)^{5/4} \left(\frac{\mathrm{C}_{\mathrm{N}_{\alpha}}}{\mathrm{C}_{\delta}}\right)^{3/4} \left(\frac{\mathrm{I}_{\mathrm{Y}}}{\mathrm{m}\,\Delta\mathrm{X}\mathrm{H}}\right) \qquad (4.4-60)$$

where

$$C_5 = \frac{1}{2}C_A + \frac{1}{4}C_{N_a} - \frac{1}{4}C_{m_q} \frac{md^2}{I_Y} = C_A C_4$$
 (4.4-61)

- 153 -
THE JOHNE HOPKING UNIVERSITY APPLIED PHYSICS LABORATORY BILVER SPRING MARYLAND

In evaluating Eq. (4.4-60), $\overline{\alpha}_{\rm E}$ must be in radians.

The corresponding equations for a rolling vehicle at reentry are:

-

$$\frac{\mathbf{g}_{\mathrm{L}}}{\overline{\mathbf{a}}_{\mathrm{E}}} = 0.42 \left[\left(\frac{\mathbf{C}_{\mathrm{N}_{\alpha}}}{\mathbf{C}_{\mathrm{A}}} \right)^{3} \left(\frac{1_{\mathrm{X}}}{\mathbf{I}_{\mathrm{Y}}} \right)^{2} \left(\frac{\mathrm{dp}^{2}}{\mathrm{g}} \right) \left(\frac{-\mathrm{V}_{\mathrm{E}}^{2} \sin \gamma}{\mathrm{gH}} \right)^{3} \left(\frac{\mathrm{I}_{\mathrm{Y}}}{\mathrm{md}^{2}} \right) \left(\frac{\mathrm{d}}{\mathrm{\Delta X}} \right) \right]^{1/4} \frac{\mathrm{K}^{3/4}}{\mathrm{C}_{4}\mathrm{R}}$$
(4.4-62)

and

$$\left(\frac{\mathbf{g}_{\mathrm{L}}}{\widetilde{\alpha}_{\mathrm{E}}}\right)_{\mathrm{max}} = 0.16 \left(\frac{\mathrm{V_{E}}^{2}}{\mathrm{gH}}\right)^{3/4} \left(-\sin\gamma\right)^{3/4} \left(\frac{\mathrm{C}_{\mathrm{N_{\alpha}}}}{\mathrm{C}_{\mathrm{\beta}}}\right)^{3/4} \left(\frac{\mathrm{I_{X}}}{\mathrm{I_{Y}}}\right)^{1/2} \left(\frac{\mathrm{I_{Y}}^{2}}{\mathrm{W}\Delta\mathrm{X}}\right)^{1/4}, \quad (4, 4-63)$$

The maximum value of g_L for a rolling vehicle also occurs at K = $\frac{3}{4} = \frac{1}{C_4}$,

The ratio of maximum g_L for a rolling vehicle to maximum g_L for a nonrolling vehicle, obtained from Eqs. (4.4-60) and (4.4-63), is:

$$\frac{g_{L_{p}}}{g_{L_{p=0}}} = 1.25 \left[\frac{-Hp}{V_{E} \sin \gamma} \frac{I_{X}}{I_{Y}} \right]^{1/2} . \qquad (4.4-64)$$

- 154 -

A LOT LE MARK MARK

.1...

THE JOHNE MOPKING UNIVERSITY APPLIED PHYSICS LABORATORY BILVER BRING MARVLAND

1



ŝ

en en en la sectador de como da **debar en e**scalada.

A Star Brown and American Section

•



- 155 -

THE JOHNS HOPKINS UNIVERSITY APPLIED PHYSICS LABORATORY SILVER SPRING MARYLAND



.....

- 156 -

...

.

, concernance of the second

and the second strange to practice

THE JOHNS HOPKING UNIVERSITY APPLIED PHYSICS LABORATORY BILVER BPRING, NARVLARS

1

م قول د. غ

I

inger Linger

T

W. m

T

à.,

I

ария (3-1) м

4 - 1

ŵ m

¶" #11

]

I

1

ş

ことのない しているのできるのできるのです しょうしん

A DEAL AND THE REPORT

a and a state of the

. .

4.5 EFFECTS OF ASYMMETRIES

4 4 4

9

5

1.

State of the second second

a subley a

in the second

٩đ

.

2 x - M

- 157 -

LAUTETER DE MERSENNER MENNEN MENNEN ARTER EN 1984 EN 1984 ANTER ANTER AND THE ANTER

.

SYMBOLS FOR SECTION 4.5

Symbol	Definition	Typical Units
А	amplification factor in angle of attack	-
^{a, a} 1 ^{, a} 2	constants defined by Eqs. (4,5-82), (4.5-87), and (4.5-88)	-
CA	axial force coefficient	-
C,	roll moment coefficient, $\ell/\overline{q}Sd$	-
C _l p	roll damping coefficient, $\partial C_{\ell} / \partial (pd/V)$	-
C _m	pitching moment coefficient, $m/\overline{q}Sd$	-
C _m	pitching moment coefficient at $\alpha = 0$	-
	$\sqrt{C_{m_0}^2 + C_{n_0}^2}$	
C _m q	pitch damping coefficient, $\partial C_m / \partial (qd/V)$	-
C _N	normal force coefficient	-
C _n	yawing moment coefficient at $\beta = 0$	-
с _N	$dC_N/d\alpha$	-
$\mathbf{C}_1, \mathbf{C}_2$	constants in Eq. (4.5-110)	rad/sec
c.g.	center of gravity	-
с,р.	center of pressure	-
D	damping function defined by Eq. (4.5-20)	-
d	reference diameter	feet
F	$\sqrt{F_Y^2 + F_Z^2}$	pounds
FA	asymmetry factor defined by Eq. (4.5–72)	-

Preceding page blank

- 159 -

-2

1.

THE JOHNE HOPKINE UNIVERSITY APPLIED PHYSICS LABORATORY BILVER SPRING MARYLAND

Symbol	Definition	Typical Units
F _x , F _y , F _z	forces along the X, Y, Z body axes	pounds
F a	$\mathbf{c}_{\mathbf{N}_{\alpha}} \mathbf{\overline{q}s}$	lb/rad
$F(\lambda, D)$	function defined by Eq. (4.5-47)	-
F ₁ , F ₂	factors defined by Eqs. (4.5-62) and (4.5-73)	-
$\mathbf{F}_3, \mathbf{F}_4$	factors defined by Eq. (4.5-125)	-
G	roll amplification factor defined by Eq. (4.5-37)	-
G _R	G required for roll "lock-in"	-
G'	roll amplification factor defined by Eq. (4.5-80)	-
g	acceleration of gravity	ft/sec 2
^g x	longitudinal acceleration	g
G ₁ , G ₂	roll amplification factors defined by Eqs. (4.5-81) and (4.5-84)	~
H	scale height, RT/g	feet
h	altitude	feet
^I x, ^I y, ^I Z	moments of inertia about X, Y, Z axes	slug-ft ²
¹ XY, ^I XZ, ^I YZ	z products of inertia about X, Y, Z axes (Eqs. (4.5-99) to (4.5-101))	slug-ft ²
к	$-P/(\beta \sin \gamma)$	-
k, k.	constants defined by Eqs. $(4, 5-59)$ and $(4, 5-60)$	-
l, m, n	aerodynamic moments about X, Y, Z	lb-ft
<i>l</i> o, mo, no	asymmetric moments about X, Y, Z	lb-ft
M	Mach number	-
м _р	roll damping moment/p	lb-ft-sec

وتعاويه والمناقب المراجع والمراجع والمراجع والمراجع والمناقل والمنافع والمراجع والم

ì

ì

٩,

Survey of the

ų

THE JOHNS HOPRING UNIVERSITY APPLIED PHYSICS LABORATORY BILVER SPRING MARYLAND

Symbol	Definition	Typical Units		
M	pitch damping moment/q	1b-ft-see		
та M	yaw damping moment/r	lb-ft-sec		
mr m	vehicle mass	slugs		
P	amhient air pressure	16/ft ²		
Ē	probability	-		
n a f	body rates about the X, Y, Z axes	rad/sec		
р. () - С	reduced roll rate = $p\sqrt{1 - I_X/I_Y}$	rad/sec		
p _o max	roll acceleration term defined by Eq. $(4, 5-38)$	rad/sec^2		
Δp	$p - p_E$ dynamic pressure ($\gamma / 2$)PM ²	rad/sec		
ч R	ras constant = 1716	ft ² /sec ² - °R		
r	radial c.g. offset, always positive	tect		
S	reference area	<u>s</u> t ²		
л Т	ambient air temperature	' R		
- t	time	reconds		
V	vehicle velocity	. c		
w	vehicle weight	pour ,վե		
x	distance from vehicle nose to the c.g.	feet		
Te, g, v	distance from vehicle nose to the c. p.	ſeet		
∩c.p. ∆X	$x_{c. p.} - X_{c. g.}$	feet		
$\left\{ \begin{array}{c} \mathbf{Y}_{c,g} \\ \mathbf{z} \end{array} \right\}$	center of gravity offset along Y and Z axis	feet		
² c.g.)	angle of attack in the XZ plane	radiuns		
α _δ	angle of attack resulting from 0 in the XZ plane at $\lambda \rightarrow \infty$	radians		

- 161 -

Section and

والواجان والمحسوف

19369 ta 11

46-341-44-03

البندائك

۰.

5

THUS PARTY IN COMPANY AND

A LANGE STATES

1

Symbol	Definition	Typical Units
$\overline{\alpha}$	total vector angle of attack = $\sqrt{\alpha^2 + d^2}$	radians
$\overline{\alpha}_{5}$	$\sqrt{\alpha_{\delta}^2 + \beta_{\delta}^2}$	radians
ل	angle of attack in the XY plane	radians
٣	or ballistic coefficient, W/C _A S	lb/ft^2
β _δ	angle of attack resulting from δ in the XY plane at λ → ∞	radians
δ _α	inclination of principal axis in the XZ plane	radians
^δ β	inclination of principal axis in the XY plane	radians
δ	$\sqrt{\delta_{\alpha}^{2} + \delta_{\beta}^{2}}$	radians
£	angle defined by Eq. $(4.5-83)$	degrees
γ	reentry flight path angle, always negative	degrees
	or ratio of specific heat, 1.4	-
Ø	space cone half-angle	degrees
	or angle defined by Eq. (4.5-31)	degrees
θ	$\sin^{-1} \frac{\overline{\alpha}_{o.g.}}{\overline{\alpha}_{o.g.}}$	degræs
λ	p_r/ω_A	-
٨	$\cos^{-1}\left[\frac{\left(\overline{\alpha_{o}} r/d\right)_{\min}}{\overline{\alpha_{o}} r/d}\right]$	degrees

and the second

ģ

تشيد منعه

1

0

. ..

ł,

-

the second second

- 162 -

THE JOHNS HOPKING UNIVERFITY APPLIED PHYSICS LABORATORY SILVER SPRING MARYLAND

100

!

1

Symbol	Definition	Typical Units
ψ	angle defined in Section 4.5.1	degrees
$\Delta \psi$	angle defined in Section 4, 5, 1	degrees
τ	period of oscillation	seconds
ω _A	aerodynamic pitch frequency	rad/sec
ω _g	$p(1 - I_X/I_Y)$	rad/sec
ū _τ	transverse rate vector, $\vec{q} + \vec{r}$	rad/sec
Subscripts		
c.g.	refers to c.g. or is caused by c.g. offset	
E	refers to reentry condition	
е	refers to equilibrium condition	
i	refers to conditions at t = 0 or to an elem	entary quantity
М	caused by asymmetric moments	
max	refers to maximum value	
min	refers to minimum value	

o refers to conditions at p = 0

R refers to resonance conditions

S.L. refers to sea level conditions

A dot over a symbol means the first derivative with respect to time. Two dots over a symbol means the second derivative with respect to time. An arrow over a symbol means a vector quantity.

موجد دادر وال

THE JOHNS HOPKINS UNIVERSITY APPLIED PHYSICS LABORATORY BILVER SPRING MARYLAND

SUMMARY AND CONCLUSIONS

A reentry body is normally spun about its longitudinal axis to provide stability during the exoatmospheric phase of flight and to reduce dispersion resulting from phenomena that occur during reentry. Ideally, the roll rate remains nearly constant during reentry. However, under the influence of mass and aerodynamic asymmetries, the roll rate may change radically. Under certain conditions the roll rate may change so that the roll rate and the aerodynamic pitch frequency are nearly the same for long intervals of time. When this occurs the roll rate is said to "lock-in" with the aerodynamic frequency. This situation is sometimes disastrous since the angles of attack and thus the lateral loads become very large. The loading situation may be aggravated further by aerothermoelastic interaction. On the other hand, the asymmetries may cause the magnitude of the roll rate to decrease and change sign. In this case the missile will be driven off course in proportion to and in the direction of the lateral loads existing at the time the roll rate is nearly zero. Even if the roll rate passes rapidly through zero roll rate, the impact dispersion may be excessive for high performance vehicles.

The six-degree-of-freedom equations are modified to include various types of asymmetry, and the conditions under which excessive loads and/or dispersion may exist are discussed.

The major conclusions are:

1. The effects on body motion of aerodynamic asymmetries (abnormal pitching, yawing, and rolling moments, and center of pressure offset from the longitudinal axis of the missile) and mass asymmetries (center of gravity offset from the longitudinal axis of the missile, products of inertia, and unequal pitch and yaw moments of inertia) are considered. All asymmetries considered

Preceding page blank _ 165 -

have some effect on body motion. However, for reasonable magnitudes of asymmetry, the effects of unequal pitch and yaw moments of inertia are negligible, the effects of any one of the other asymmetries are small, but the effects on vehicle performance of some combinations of the asymmetries may be catastrophic.

2. Unacceptable vehicle performance is likely if either resonance (resulting in large lateral loads) or zero roll rate (resulting in large impact dispersion) occurs during reentry. In either case the asymmetries must cause both a trim angle of attack and a roll torque. Therefore, a center of gravity offset combined with either an asymmetric pitching (or yawing) moment or with nonzero products of inertia or a pure rolling moment asymmetry combined with either of these asymmetries is apt to result in unacceptable performance.

3. The conditions under which adverse performance is likely to occur are dependent upon many variables so that each vehicle must be analyzed separately. Nevertheless, the following trends may be stated. The probability of encountering difficulties resulting from combined asymmetries is increased for missiles:

- o Having low stability margin
- o Having resonance at an altitude near maximum dynamic pressure
- o Having low normal force per degree angle of attack and/or a low pitch damping moment
- o Flying shallow trajectories
- o Having slender shapes
- o Having small dimensions.

4. Detailed equations are presented for evaluating the probability of encountering "lock-in" or spin through zero roll rate for a missile of any arbitrary design. Comments are made regarding the accuracy of these equations. Illustrations of the use of these equations are given in Examples 5, 6, and 7, Section 4.6. THE JOHNS HOPKINS UNIVERSITY APPLIED PHYSICS LABORATORY BILVER SPRING MARYLAND

> In the derivation of the equations given in Section 4.4, it was assumed that the body was symmetrical. We consider next the effects on vehicle motion of asymmetry in mass and external shape.

Asymmetries in mass and external shape result from design and manufacturing tolerances or from inflight phenomena and may cause one or more of the following anomalies:

Aerodynamics

- 1. A change in normal force characteristics
- 2. A change in pitching moment characteristics
- 3. A lateral shift in center of pressure
- 4. A pure roll torque.

Mass

- 5. A lateral shift in center of gravity
- 6. Nonzero products of inertia
- 7. Unequal pitch and yaw moments of inertia.

The effect of asymmetries on normal force is usually negligible but the effect on the pitching moment is very significant. A lateral shift in the center of presoure has the same effect as a lateral shift in the center of gravity, and these two effects will be combined in the following discussion. For convenience, the effect will be referred to as a center of gravity offset effect, but it is important to remember that the same effect results from a center of pressure offset. This section will be limited to the following asymmetries:

- 1. Asymmetric moment, mo and no
- 2. Center of gravity offset, Z_{c.g.} and Y_{c.g.}
- 3. Nonzero products of inertia, I_{XY} and I_{XZ}

- 167 -

THE JOHNS HOPKINS UNIVERSITY APPLIED PHYSICS LABORATORY BILVER SPRING MANYLAND

يعادمهم فأسترك والمستحد

4. Unequal pitch and yaw moments of inertia

1.11

5. Pure roll torque, l_0 .

The asymmetries have negligible effect on all portions of the trajectory except for the reentry phase.

To account for the asymmetries, the equations given in Section 4.4 must be modified to the following (see the sketch on page 170):

$$b I_{X} = l - qr(I_{Z} - I_{Y}) + I_{XY}(q - pr) + I_{XZ}(r + pq) + I_{YZ}(q^{2} - r^{2})$$
 (4.5-1)

$$d_{1_{Y}} = m - pr(I_{X} - I_{Z}) + I_{XY}(b + rq) + I_{XZ}(r^{2} - p^{2}) + I_{YZ}(r - pq)$$
 (4.5-2)

$$f_{Z} = n - pq(I_{Y} - I_{X}) + I_{XY}(p^{2} - q^{2}) + I_{XZ}(p - rq) + I_{YZ}(q + pr)$$
(4.5-3)

$$I = M_{p}p + F_{Y}Z_{c,g} = F_{Z}Y_{c,g} + I_{o}$$
 (4.5-4)

$$m = F_Z^{\Delta X} + M_q q - F_X^Z_{c, g,} + m_o$$
 (4.5-5)

$$n = -F_{Y} \Delta X + M_{r} r + F_{X} Y_{c,g} + n_{o}$$
(4.5-6)

4.5.1 Effect of Asymmetric Moment

Assuming that only an asymmetric moment (m_0 and n_0) exists, Eq. (4.5-1) reduces to:

$$pI_{X} = M_{p}p, \qquad (4.5-7)$$

Therefore the roll rate decreases slightly under the influence of viscous damping, M_p . The trim equations $(\dot{q} = \dot{r} = 0)$ become:

$$F_{Z} \Delta X + M_{q}q + m_{o} = pr(I_{X} - I_{Z})$$
 (4.5-8)

$$-F_{Y} \Delta X + M_{r} r + n_{o} = pq(I_{Y} - I_{X})$$
 (4.5-9)

e en en berenden ander

- 168 -

THE JOHNS HOPKING UNIVERSITY APPLIED PHYSICS LABORATORY SILVER SPRING MARYLAND

Consider first the trim conditions for p = 0. The pitch damping terms are negligible for this case, and the trim equations become:

$$\mathbf{F}_{Z_{O}} = -\frac{\mathbf{m}_{O}}{\Delta \mathbf{X}} \quad (4.5-10)$$

$$\mathbf{F}_{\mathbf{Y}_{O}} = \frac{\mathbf{n}_{O}}{\Delta \mathbf{X}} \qquad (4.5-11)$$

Writing F_{Y} and F_{Z} in terms of α and β (Eqs. (4, 4-13) to (4, 4-15)) and expressing m_0 and n_0 in coefficient form $(m_0 = C_{m_0} qSd and n_0 = C_{n_0} qSd)$;

$$\alpha_{0} * \frac{C_{m_{0}}}{C_{N_{0}} \frac{\Delta X}{d}} \cdot \qquad (4.5-12)$$

$$\beta_{0} = -\frac{C_{n_{0}}}{C_{N_{\alpha}}\frac{\Delta X}{d}} \qquad (4, 5-13)$$

The total angle of attack is:

$$\overline{\alpha}_{0} = \frac{\left[\left(C_{m_{0}}\right)^{2} + \left(C_{n_{0}}\right)^{2}\right]^{1/2}}{C_{N_{\alpha}}\frac{\Delta X}{d}} = \frac{\overline{C}_{m_{0}}}{C_{N_{\alpha}}\frac{\Delta X}{d}} \qquad (4.5-14)$$

and the plane of $\overline{\alpha_o}$ defined by the direction of F_o is given by the equation:

$$\cot \psi_{0} = \frac{F_{Y_{0}}}{F_{Z_{0}}} = \frac{C_{n_{0}}}{-C_{m_{0}}} = \frac{-\beta_{0}}{-\alpha_{0}}$$
 (4.5-15)

- 169 -

THE JOHNS HOPKING UNIVERSITY APPLIED PHYSICS LABORATORY SILVER SPRING MARVLAND



Note that for ψ_0	Ξ	0°	to	9 0 °	β	is	•	and α	is	-
	=	9 0 •	to	18 0°	β	is	+	and α	i s	-
	æ	180°	to	270°	β	នេ	+	and α	18	÷
	8	27 0 °	to	36 0°	β	is	-	and α	នេ	+

The angle ψ_0 also defines the plane of the asymmetry. That is, for zero roll rate the total angle of attack, α_0 , is in a plane containing the asymmetry. For example, if the asymmetry results from a flat spot on the body, the flat is located at a circumferential location $\psi = \psi_0$ or $\psi = \psi_0 + 180^\circ$ (depending upon the fore and aft location of the asymmetry).

The effect of roll rate is to modify the magnitude of α and to cause the total lateral force to be rotated out of the plane of the asymmetry. These effects are given by the following equations (Ref. 4-6) based on theory given in Refs. 4-7 and 4-8.

- 170 -

1

THE JOHNS HOPKINS UNIVERSITY APPLIED PHYSICS LABORATORY SILVER SPRING MARYLAND

$$A = \left|\frac{\overline{\alpha}}{\overline{\alpha}_{0}}\right| = \left[\left(1 - \lambda^{2}\right)^{2} + D^{2}\lambda^{2}\right]^{-1/2}$$
(4.5-16)

$$\tan \Delta \psi = \frac{-D\lambda}{1-\lambda^2}$$
(4.5-17)

A
$$\cos \Delta \psi = \frac{1 - \lambda^2}{(1 - \lambda^2)^2 + D^2 \lambda^2}$$
 (4.5-18)

$$A = \sin \Delta \psi = \frac{-D\lambda}{(1 - \lambda^2)^2 + D^2 \lambda^2}$$
(4.5-19)

$$D = \frac{\frac{F_{\alpha}}{mV} \left(1 - \frac{I_{x}}{I_{y}}\right) - \frac{M_{q}}{I_{y}}}{\omega_{A} \sqrt{1 - \frac{I_{x}}{I_{y}}}}$$
(4.5-20)

$$\lambda = \frac{p\sqrt{1 - \frac{I_X}{I_Y}}}{\omega_A} = \frac{p_r}{\omega_A}$$
(4.5-21)

From Eq. (4, 5-16) it is apparent that the amplification factor increases from 1 at $\lambda = 0$ to a maximum value $\left[D^2 (1 - \frac{D^2}{4}) \right]^{-1/2}$ at $\lambda = \sqrt{1 - \frac{D^2}{2}}$ and then decreases to 0 at $\lambda \rightarrow \infty$. Thus, at λ greater than about $\sqrt{2}$ (depending upon the value of D), the trim angle of attack is attenuated by the effects of roll rate. Typical values of D are of the order of 0 to 0.2 so that A_{\max} occurs near $\lambda = 1$ and $A_{\max} \approx 1/D$. The condition of $\lambda = 1$ is called the pitchyaw-roll, or simply the roll resonance condition. Since

- 171 -

Sends and a most

* 5 de

THE JOHNS HOPKING UNIVERSITY APPLIED PHYSICS LABORATORY SILVER SPRING MARYLAND

 $I_X/I_Y \approx 0.1$, the resonance condition is very close to the condition $p/\omega_A = 1$. Note that near resonance the pitch damping term, D, plays a very important roll in limiting α .

The value of D is always positive for conditions (static and dynamic stability) for which Eqs. (4.5-16) and (4.5-17) are valid. Therefore, from Eq. (4.5-17) for positive roll rates the plane of α for a rolling vehicle is rotated counterclockwise ($\Delta \psi$ is negative) from ψ_0 . At resonance, the plane is rotated 90° from ψ_0 ; at very high values of λ , the plane is rotated 180° from ψ_0 .



For example, assume that an angle of attack is produced by a flap at the aft end of the body as shown in the sketch on page 173.

alise hear ^atanada a satu

THE JOHNS HOPKINS UNIVERSITY APPLIED PHYSICS LABORATORY FILVER SPRING MARYLAND



At zero roll rate, this flap produces a + γ or a force F = F₀ along the -Z axis. As the roll rate increases in a positive direction, F rotates counterclockwise. At resonance F = F_R = F₀/D and is directed along the -Y axis. As the roll rate increases further, F continues to rotate counterclockwise and decreases in magnitude. In the limit (very high roll rates) F - 0 and is directed along the +Z axis.

To further illustrate the behavior of Eqs. (4.5-16) and (4.5-17), Fig. 4-13 shows plots of A and $\Delta \psi$ as functions of λ for D = 0 and 0.2. Note that neither A nor $\Delta \psi$ changes very rapidly for λ from 0 to about 0.5. Practically all the change in $\Delta \psi$ occurs from $\lambda = 0.8$ to 1.4. For λ greater than about 1.4, $\overline{\alpha}_0$ is attenuated.

The effect of damping is small except near $\lambda = 1$ where the effect on both A and $\Delta \psi$ is large. As the reentry body descends through the atmosphere, ω_A increases, and unless the body is given a very high spin rate the resonance condition will occur at least once during reentry. If $\overline{\alpha}_0$ were zero, the resonance condition

- 173 -

would have no effect on the trajectory. If the only asymmetry were the asymmetric moments $(n_0 \text{ and } m_0)$, the effect of resonance would normally be minor since p is essentially constant (changes by viscous damping only) and ω_A would pass through the resonance condition too rapidly to produce significant changes in the trajectory. The α would increase somewhat over the α_0 value, resulting in corresponding increases in body lift and drag, but the time duration would be short. Therefore, the effect of an asymmetric moment, by itself, has negligible effect on the trajectory for reasonable values of m_0 and n_0 .

4, 5, 2 Offset Center of Gravity

The effect on trim conditions of the center of gravity offset (again neglecting M_{c}) for p = 0 is given by:

$$\mathbf{F}_{\mathbf{Z}} = \mathbf{F}_{\mathbf{X}} \frac{\mathbf{Z}_{\mathbf{C}, \mathbf{g},}}{\Delta \mathbf{X}}$$
(4.5-22)

THE REAL PROPERTY.

「「「「」」

A state of the state of the second

$$F_{Y} = F_{X} - \frac{Y_{c,g,}}{\Delta X} \qquad (4.5-23)$$

Substituting for F_X , F_Y , and F_Z from Eqs. (4.4-12) through (4.4-15):

$$\alpha_{0} = \frac{C_{A}}{C_{N_{o}}} \frac{Z_{c.g.}}{\Delta X}$$
(4.5-24)

$$3_{0} = \frac{C_{A}}{C_{N_{\alpha}}} \frac{Y_{c.g.}}{\Delta X}$$
(4.5-25)

$$\overline{\alpha}_{0} = \frac{C_{A}}{C_{N_{\alpha}}} \left[\left(\frac{Z_{c.g.}}{\Delta X} \right)^{2} + \left(\frac{Y_{c.g.}}{\Delta X} \right)^{2} \right]^{1/2} = \frac{r}{\Delta X} \frac{C_{A}}{C_{N_{\alpha}}} (4.5-26)$$

- 174 -

The effect of roll rate is to modify the magnitude of α_0 and to change the plane in which it acts in accordance with Eqs. (4,5-16) and (4,5-17). However, unlike the aerodynamic asymmetry, the center of gravity offset affects the roll rate equation (see Eq. (4,5-4)). The effect on roll rate is illustrated by the sketch.



For zero roll rate, $\overline{\alpha_0}$ is in a plane containing the center of gravity, and F_0 acts through the center of pressure directed away from the center of gravity as shown by ψ_0 . However, for a positive roll rate the force vector F is rotated counterclockwise from F_0 as explained previously. Therefore, F results in a negative roll torque that tends to decrease the roll rate. A similar argument for a negative roll rate shows that the center of gravity offset results in a positive roll torque, again tending to drive the roll rate toward zero. Therefore, the effect of the center of gravity offset alone behaves as a roll damping effect and, in fact, must be accounted for when attempting to extract C_{ℓ} from experimental roll rate decay data. For

- 175 -

THE JOHNS HOPKINS UNIVERSITY APPLIED PHYSICS LABORATORY SILVER SPRINT MARYLAND

> reasonable values of the center of gravity offset, the effect on roll rate is small except near resonance where the roll moment arm is maximum (see sketch above).

4.5.3 Combined Asymmetry (\overline{C}_{m_0} and r)

Both the center of gravity offset and asymmetric moment result in minor effects on the trajectory when considered separately. However, this is not the case when the two types of asymmetry are combined. Using the described procedures, the trim angles of attack at zero roll rate for this case are given by the equations:

$$\alpha_{o} = \frac{C_{m_{o}} + C_{A} \frac{Z_{c.g.}}{d}}{C_{N_{\alpha}} \frac{\Delta X}{d}}$$
(4.5-27)

$$B_{o} = \frac{C_{A} - \frac{Y_{c.g.}}{d} - C_{n_{o}}}{C_{N_{\alpha}} - \frac{\Delta X}{d}}$$
(4.5-28)

$$\overline{\alpha}_{0} = \left[\alpha_{0}^{2} + \beta_{0}^{2}\right]^{1/2}$$
(4.5-29)

The direction of F_0 is given by the equation:

$$\cot \psi_{0} = \frac{-\beta_{0}}{-\alpha_{0}} = \frac{C_{n} - C_{A}}{-C_{m} - C_{A}} = \frac{\frac{Y_{c.g.}}{d}}{-C_{m} - C_{A}} = \frac{Z_{c.g.}}{d}$$
 (4.5-30)

The proper quadrant for ψ_0 is again determined as indicated in Section 4.5.1. An important variable for this combined asymmetry is the angle, \emptyset , between the plane of the center of gravity and the plane of $\overline{\alpha}$.

- 176 -

See Orie

THE JOHNS HOPKINS UNIVERSITY APPLIED PHYSICS LABORATORY BILVER SPRING VARYLAND



$$\phi = \psi - \psi_{c.g.}$$
 (4.5-31)

where

 $\begin{array}{l} \cot \psi_{c,g,} = \frac{Y_{c,g,}}{Z_{c,g,}} \\ \psi_{c,g,} = 0^{\circ} \text{ to } 90^{\circ} & \text{ if } Y_{c,g,} \text{ is + and } Z_{c,g,} \text{ is +} \\ = 90^{\circ} \text{ to } 180^{\circ} & \text{ if } Y_{c,g,} \text{ is - and } Z_{c,g,} \text{ is +} \\ = 180^{\circ} \text{ to } 270^{\circ} & \text{ if } Y_{c,g,} \text{ is - and } Z_{c,g,} \text{ is -} \\ = 270^{\circ} \text{ to } 360^{\circ} & \text{ if } Y_{c,g,} \text{ is + and } Z_{c,g,} \text{ is -} \end{array}$

It must be assumed that the center of gravity offset and moment asymmetries are completely independent variables so that ϕ_0 nay have any value. For purposes of illustrating the significance of ϕ_0 it is convenient to consider "in plane" asymmetries ($\phi_0 = 0^\circ \text{ or } 180^\circ$) and "out of plane" asymmetries ($\phi_0 = 90^\circ \text{ or } 270^\circ$). For the in plane asymmetry, if $\phi_0 = 180^\circ$, the effect is the same as that

- 177 -

THE JOHNS HOPKING UNIVERSITY APPLIED PHYSICS LABORATORY BILVER BERING MARYLAND

> described previously for the center of gravity offset. If $\phi_0 = 0^\circ$, the magnitude of the roll rate will increase indefinitely (\dot{p} is always positive) if the initial roll rate is positive and will always increase negatively if the initial roll rate is negative. This case is shown in the sketch,



diaments in a subscription

一一日本日、二日本日本日、日本日本日

حيدهم يعنصه

At p = 0, F_0 passes through the center of gravity. If the roll rate is initially positive F must be counterclockwise from F_0 as shown, so the a positive roll torque develops driving p to a larger positive value. Since Fmust lie between ψ_0 and $\psi_0 = 180^\circ$, the roll torque is always positive. Similarly, if the initial roll rate is negative, the roll torque produced by the asymmetries is always negative.

For out of plane asymmetries, the case of $\phi_0 = 90^\circ$ is sketched for an initial positive roll rate. Let the asymmetric force be located initially at F (counterclockwise from F₀ since p is positive). The roll torque is negative and p decreases. For decreasing p, λ decreases (assuming ω_A is constant) and F rotates clockwise. The roll rate becomes zero when F = F₀ and then the vehicle begins to spin up in the negative direction. This process continues

- 178 -

THE JOHNE HOPKINE UNIVERSITY APPLIED PHYSICS LABORATORY BILVER SPRING MARVLAND



until F rotates 90° from F_0 . This point is an equilibrium roll rate position, since at higher negative roll rates the roll torque becomes positive, driving F to a less negative roll rate. This equilibrium roll position also corresponds to the resonance condition. For $\phi_0 = 270^\circ$ the same phenomenon occurs except that the roll torque causes spin-up to resonance at positive roll rates and therefore does not spin through zero roll rate.

The potential problems resulting from these combined asymmetries are;

- 1. Excessive spin rates resulting in structural or payload failures
- 2. Prolonged periods of roll resonance resulting in high lateral loads and high drag (structural failure and excessive impact dispersion)
- 3. Spin through zero roll rate resulting in large impact errors (see Section 5. 4. 2).

. mu

88,238, 119, 119, 3

- 179 -

THE JOHNS HOPKING UNIVERSITY APPLIED PHYSICS LABORATORY Silver Spring Maryland

> Both in plane and out of plane components of the asymmetry can have disastrous results onvehicle performance. However, the in plane asymmetries are less likely to cause difficulty because significant roll torques are developed only at conditions near resonance. Also, this type of asymmetry cannot produce the problem of spin through zero.

4.5.4 Effect of Combined Asymmetries on Roll Rate

All the potential problems resulting from a combination of asymmetric moment and center of gravity offset listed in Section 4.5.3 exist because of changes in roll rate during reentry. In this section we will consider the roll behavior in more detail.

If $I_Y = I_Z$ and if the products of inertia are zero, the roll acceleration (Eq. (4, 5, 1)) becomes:

$$p = \frac{M_p p + F_Y Z_{c.g.} - F_Z Y_{c.g.} + \ell_o}{I_X} . \qquad (4.5-33)$$

相當書

Ł

Neglecting the roll damping and pure roll torque terms and expressing the numerator in terms of F, r, and \emptyset :

$$\dot{p} = \frac{-Fr \sin \emptyset}{I_X} \qquad (4.5-34)$$

Since F = $C_{N_{\alpha}} \overline{q} S \overline{\alpha}$, Eq. (4.5-34) may be written:

$$\dot{p} = \frac{-C_N q S \alpha r \sin \theta}{I_X} \qquad (4.5-35)$$

The roll acceleration may be written as a product of two terms:

$$\dot{p} = G\dot{p}_{0}$$
 (4.5-36)

- 180 -

THE JOHNS HOPKINS UNIVERSITY APPLIED PHYSICS LABORATORY GLUER SPRING MARYLAND

where

$$G = -\frac{\overline{\alpha}}{\overline{\alpha}_{o}}\sin\phi = -A\sin\phi \qquad (4.5-37)$$

$$\dot{P}_{o_{\max}} = \frac{C_{N} \overline{q} S \overline{\alpha}_{O} r}{I_{X}} \qquad (4.5-38)$$

The G term contains the effects of roll rate and ϕ_0 ; the $\dot{p}_{0\max}$ term contains the body geometry, environmental, and basic asymmetry terms. Note that $\dot{p}_{0\max}$ is the <u>positive</u> roll acceleration that would exist for zero roll rate if the plane of α_0 were normal to the center of gravity plane (that is, $\phi_0 = 270^\circ$). Since $\phi = \phi_0 + \Delta \psi$, Eq. (4.5-37) may be written:

$$G = -A \cos \Delta \psi (\sin \phi + \cos \phi \tan \Delta \psi), \quad (4.5-39)$$

Substituting for tan $\Delta \psi$ and A cos $\Delta \psi$ from Eqs. (4.5-17) and (4.5-18), Eq. (4.5-39) becomes:

$$G = \frac{D\lambda \cos \phi_{0}^{2} - (1 - \lambda^{2}) \sin \phi_{0}^{2}}{(1 - \lambda^{2})^{2} + D^{2} \lambda^{2}} , \qquad (4.5-40)$$

- Second Strate

an the states of some

We will define $\dot{p} = 0$ as the roll equilibrium condition. For this condition either \dot{p}_{omax} or G must be zero. For a reentry body in the sensible atmosphere ($\bar{q} \neq 0$), \dot{p}_{omax} is zero only if $\bar{\alpha}_0$ or r is zero. Generally, neither $\bar{\alpha}_0$ nor r is zero and the roll rate of a reentry body having combined asymmetries will vary until G = 0. Defining λ_e as the value of λ at G = 0, from Eq. (4.5-40), $\lambda_e = \pm \infty$, or λ_e must satisfy the equation:

- 181 -

THE JOHNS HOPKING UNIVERSITY APPLIED PHYSICS LABORATORY BILVER SPRING MARYLAND

$$\tan \phi_0 = \frac{D\lambda_e}{1 - \lambda_e^2}$$
 (4.5-41)

Equation (4.5-41) may be solved for λ_{o} :

$$\lambda_{e} = \frac{-D \cot \phi_{o} \pm \sqrt{(D \cot \phi_{o})^{2} + 4}}{2} , \qquad (4.5-42)$$

For very large values of D cot ϕ_0 ($\phi_0 \approx 0^\circ$ or 180°);

$$\lambda_e = 0$$
 . (4.5-43)

For very small values of D cot ϕ_0 ($\phi_0 \approx 90^\circ$ or 270°);

$$\lambda_{e} = 1 - \frac{D}{2} \cot \phi_{o}$$
 or $-(1 + \frac{D}{2} \cot \phi_{o})$, (4.5-44)

Equation (4.5-42) is plotted as a function of D cot ϕ_0 in Fig. 4-14.

The roll equilibrium condition, λ_e , may be stable or unstable; that is, if the roll rate is changed slightly, λ changes slightly from λ_e . If the roll torque developed at λ tends to drive λ back to λ_e , the equilibrium condition is stable; if the roll torque developed tends to drive λ away from λ_e , the condition is unstable. The λ_e condition is stable if $(dG/d\lambda)_{\lambda_e}$ is negative and unstable if $(dG/d\lambda)_{\lambda_e}$ is positive. It may be shown that $dG/d\lambda$ is given by:

$$\frac{dG}{d\lambda} = \frac{D \cos \phi_0 \left[1 + (2 - D^2)\lambda^2 - 3\lambda^4\right] + 2\lambda \sin \phi_0 \left[(D^2 - 1) + 2\lambda^2 - \lambda^4\right]}{\left[(1 - \lambda^2)^2 + D^2\lambda^2\right]^2} \quad (4.5-4b)$$

- 182 -

28 Sections

THE JOHNS HOPKINS UNIVERSITY APPLIED PHYSICS LABORATORY BILVER SPRING MARYLAND

Therefore, the condition for λ_e stability is that:

$$F(\lambda_{\rho}, D) > \tan \phi_{\rho}$$
 (4.5-46)

where

F (
$$\lambda$$
, D) = $\frac{D \left[3\lambda^4 - (2 - D^2) \lambda^2 - 1 \right]}{2\lambda \left[(D^2 - 1) + 2\lambda^2 - \lambda^4 \right]}$. (4.5-47)

For in plane asymmetries the condition for stability reduces to:

$$\cos \phi_0 \left[1 + (2 - D^2) \lambda^2 - 3\lambda^4 \right] < 0$$
. (4.5-48)

Since $\lambda_e = 0$, the condition is stable for $\phi_0 = 180^\circ$ and unstable for $\phi_0 = 0^\circ$.

For out of plane asymmetries, the condition for stability reduces to:

$$\sin \phi_{0} \lambda \left[(D^{2} - 1) + 2\lambda^{2} - \lambda^{4} \right] < 0 . \qquad (4.5-49)$$

Since $\lambda_e = \pm 1$, the condition is unstable for $\lambda_e = \pm 1$ and $\phi_o = 90^\circ$ or $\lambda_e = -1$ and $\phi_o = 270^\circ$; the condition is stable for $\lambda_e = \pm 1$ and $\phi_o = 270^\circ$ or $\lambda_e = -1$ and $\phi_o = 90^\circ$.

Another characteristic of interest is the maximum values of G. The term maximum is used to mean either a positive or negative maximum. Maximum values occur at values of λ obtained by setting the numerator of Eq. (4.5-45) to zero giving the expression:

$$\tan \phi = F(\lambda, D) \qquad (4.5-50)$$

المرجو وكعاد وأكا

For $\phi_0 = 0^\circ$ or 180°, neglecting terms smaller than D², Eq. (4.5-50) may be solved for values of λ and G_{max} giving the result:

- 183 -

THE JOHNS HOPKINS UNIVERSITY APPLIED PHYSICS LABORATORY SILVER SPRING MARYLAND

$$\lambda = \pm (1 - \frac{D^2}{8})$$
 (4.5-51)

From Eq. (4.5-40):

$$G_{\max} = \pm \frac{(1 + \frac{D^2}{16}) \cos \phi}{D}$$
, (4.5-52)

Therefore, at
$$\lambda \approx \pm 1$$
 $G_{\max} \approx \frac{1}{D} \text{ for } \phi_0 \equiv 0^\circ$
 $\lambda \approx -1$ $G_{\max} \approx -\frac{1}{D} \text{ for } \phi_0 \equiv 0^\circ$
 $\lambda \approx \pm 1$ $G_{\max} \approx -\frac{1}{D} \text{ for } \phi_0 \equiv 180^\circ$
 $\lambda \approx -1$ $G_{\max} \approx -\frac{1}{D} \text{ for } \phi_0 \equiv 180^\circ$

For $\phi_0 = 90^\circ$ or 270°, Eq. (4.5-50) may be solved for λ , giving the following equation:

$$\lambda = \pm \sqrt{1 \pm D} \quad . \tag{4.5-53}$$

From Eq. (4.5-40):

$$G_{max} = \pm \frac{\sin \phi_o}{D(2 \pm D)}$$
 (4,5-54)

The value of G is the same for $\pm \lambda$. The \pm sign in Eq. (4.5-54) corresponds to the $\pm D$ term in Eq. (4.5-53). Therefore for $\emptyset_0 = 90^\circ$, $G_{\max} \approx 1/2D$ at $\lambda = \pm \sqrt{1+D}$, and $G_{\max} \approx -1/2D$ at $\lambda = \pm \sqrt{1-D}$. For $\emptyset_0 = 270^\circ$, $G_{\max} \approx -1/2D$ at $\lambda = \pm \sqrt{1+D}$ and $G_{\max} \approx 1/2D$ at $\lambda = \pm \sqrt{1-D}$. Note that, for a given $\overline{\alpha}_0$ and r (i.e., $\frac{1}{2} - \frac{1}{2} - \frac{1$

- 184 -

n mit meter om som konstalen och s 1991 – Dri Maria Maria (Salahara For the general value of \emptyset_0 , the maximum values of G may be obtained from a plot of $F(\lambda, D)$. Since $F(-\lambda, D) = -F(\lambda, D)$, it is necessary to plot only the $+\lambda$ values of the function. The maximum values of G for $+\lambda$ are obtained at $F(\lambda, D) = \tan \emptyset_0$; the values of $-\lambda$ are obtained at $F(\lambda, D) = -\tan \emptyset_0$.

A summary of the characteristics of G is given in the table for in plane and out of plane asymmetries.

1	0*		90•		180	•	270*		
G.	0 _{max}	λ	G _{max}	λ	G _{niax}	λ	Gmax	λ	
Positive ^G max	1/0	$1 - \frac{D^2}{4}$	1/20	±√1+D	1/D	$-1+\frac{p^2}{4}$	1/20	± 1 - D	
Negative O _{max}	-17D	-1+ <mark>0</mark> 2	-1/20	±√1•D	-1/D	1 - <mark>D²</mark>	+1/2D	± √1+D	
λ.	0		±1		U		±1		
λ. Stability	Unstable		Unstable at +1 Stable at -1		Stab	10	Stable at +1 Unsiable at -1		

Typical plots of G versus λ for D = 0.2 are shown for $\emptyset_0 = 0^\circ$, 45°, and 90° in Fig. 4-15. The values of G at a given λ for $\emptyset_0 = 180^\circ$ and 270° are the negatives of the values for $\emptyset_0 = 0^\circ$ and 90°, respectively. Note that for in plane asymmetries the roll torque from combined asymmetries is small except for a narrow band of λ in the vicinity of ±1 (i. e., near resonance conditions). However, for out of plane asymmetries the roll torque may be appreciable for all values of λ between about -1.5 to 1.5 except at $\lambda = \pm 1$.

The function $F(\lambda, D)$ is shown for D = 0.2 in Fig. 4-16. Using Eq. (4.5-50) as the condition for G_{max} , note that, except for small values of ϕ_0 (less than $\phi_0 = 45^{\circ}$), all values of G_{max} tend to be near $\lambda = 0$ or $\lambda = \pm 1$. Furthermore the magnitude of G_{max} tends to be small when it occurs at values of λ other than ≈ 1 . For example, the maxima for $\phi_0 = 45^{\circ}$ or $F(\lambda, D) = 1$ are shown in Fig.

- 185 -

4-16 to occur at $\lambda = 0.12$, 0.74, and 1.06. From Fig. 4-15 the corresponding values of G_{max} are -0.7, -0.9, and 4.1, respectively.

4.5.5 Roll Resonance

One of the potential problems resulting from asymmetries is prolonged periods of roll resonance. During this period the angle of attack increases, resulting in large lateral acceleration and high drag. If the resonance occurs during the heating portion of the trajectory, the problem may be amplified by distortion resulting from asymmetric heating since at $\overline{\alpha}$ one ray along the body remains on the windward side of the body and is heated more than a corresponding ray on the leeward side.

In this section we will consider conditions under which prolonged periods of resonance may occur. The term "lock-in" will be used to describe a roll history in which p_r tends to follow ω_A so that $\lambda \approx$ constant. That is, $\dot{p}_r/\dot{\omega}_A = 1$. It should be noted that this does not necessarily imply that resonance ($\lambda = \pm 1$) exists. However, asymmetries are usually small enough that roll amplification is required to cause significant changes in roll rate and the necessary amplification can occur only for conditions at or near resonance.

Typical values of roll rate at reentry are 1 to 2 Hz, and the static margin typically is 3% to 4% of body length. For these conditions, the peak value of ω_A is much greater than p_E so that $\lambda = 1$ will occur at least once during reentry. If the range is short enough or the reentry angle shallow enough, the resonance condition may occur a second time, as shown in the sketch on page 187.

- 186 -



A vehicle having no roll torque would exhibit a reduced roll rate (p_r) trace shown by 1. The roll rate would pass through "first" and "second" resonance without incident. However, if asymmetries exist, the roll rate may lock-in on ω_A as shown by 2 through 5. Curve 2 is typical of the case of large asymmetries that exist early in the reentry phase. In this case lock-in may occur at "first" resonance. In the more usual case, asymmetries develop as a result of the heating environment so that "spin-up" (cases 3 and 4) or "spin-down" (case 5) occurs after "first" resonance. For cases 3 and 4 the angle ϕ_0 is between 180° and 360° so that "spin up" occurs; for case 5, ϕ_0 is between 0° and 180° so that "spin down" occurs. Cases 3 and 4 indicate that p may lock-in to ω_A (case 4) or may cut through the ω_A curve (case 3).

The roll lock-in conditions are the same whether the vehicle has positive or negative roll rate at reentry.

- 187 -

nen andere er de eren her andere er andere er andere er andere Angele an gestalt af New Society (Spin Antes) er er er societ To simplify the presentation, only positive roll rate will be considered in the remainder of this discussion.

If the reduced roll acceleration (\dot{p}_r) , produced by asymmetries, is equal to or greater than $\dot{\omega}_A$, lock-in may occur. In equation form, lock-in may occur when:



(4,5-55)

All the second second

Using Eqs. (4.4-37) and (4.3-18), it may be shown that $\dot{\omega}_A$ is given by the equation (neglecting variations in temperature with altitude):

$$\dot{\omega}_{A} = V_{E}^{2} \left[-\frac{C_{N} \Delta X m \sin^{3} \gamma}{8 C_{A} I_{Y} H^{3}} \right]^{1/2} \frac{(1-K)\sqrt{K}}{e^{K}} .$$
 (4.5-56)

Using Eqs. (4, 3-8), (4, 5-36), and (4, 5-38), p is given by the equation:

$$\phi = \frac{-G V_E^2 m \sin \gamma C_N}{2I_X H C_A} \overline{\alpha}_o r \frac{K}{e^K} . \qquad (4.5-57)$$

Therefore, lock-in may occur if the following inequality is satisfied:

$$\overline{\alpha}_{0} \frac{\mathbf{r}}{\mathrm{d}} \mathbf{G} \ge \left[\frac{-\mathbf{C}_{A} \Delta \mathbf{X}}{2 \mathbf{C}_{N} \mathbf{H}} \frac{\mathbf{I}_{\mathbf{X}}}{\mathrm{md}^{2}} \frac{\mathbf{I}_{\mathbf{X}} \sin \gamma}{\mathbf{I}_{\mathbf{Y}} \left(1 - \frac{\mathbf{I}_{\mathbf{X}}}{\mathbf{I}_{\mathbf{Y}}}\right)} \right]^{1/2} \frac{1 - \mathbf{K}_{\mathrm{R}}}{\sqrt{\mathbf{K}_{\mathrm{R}}}} \quad . \tag{4.5-58}$$

Equation (4.5-58) may be simplified further by noting that, for a given altitude (or given value of K), the minimum

- 188 -

asymmetry that may result in lock-in is given by Eq. (4.5-58) for $G = G_{max}$. If trim conditions are realized (see Section 4.5.10), then G_{max} is given approximately by (see Eqs. (4.5-52) and (4.5-54)):

$$G_{max} = k/D$$
 (4,5-59)

where k has the following values:

Using Eq. (4.5-20) and the definitions of $F_{\pmb{\alpha}}$ and M_q , an expression for D is:

$$D = \left[k_1 - C_{m_q}\right] \left[\frac{-\sin\gamma}{2C_A \frac{\Delta X}{d} k_1 \frac{H}{d}}\right]^{1/2} \sqrt{K} \qquad (4.5-60)$$

where

$$k_1 = C_{N_{\alpha}} (1 - \frac{1_X}{1_Y}) \frac{1_Y}{md^2}$$

Substituting Eqs. (4.5-59) and (4.5-60) into (4.5-58), and noting from previous discussions that G_{max} occurs near the resonance condition for in plane or out of plane asymmetries;

$$\overline{\alpha}_{0} \frac{r}{d} \ge F_{1} \frac{(1 - K_{R})}{k}$$
(4.5-61)

- 189 -

THE JOHNS HOPKINS UNIVERSITY APPLIED PHYSICS LABORATORY Silver Spring, Maryland

where

$$F_{1} = \frac{-(I_{X}/I_{Y}) \sin \gamma}{2C_{N_{\alpha}} \frac{H}{d} \left(1 - \frac{I_{X}}{I_{Y}}\right)} \left[k_{1} - C_{m_{q}}\right] \qquad (4.5-62)$$

and $K_{\mathbf{R}}$ is the value of K at the resonance condition.

The general behavior of Eq. (4.5-61) is shown in the sketch.



And the other of the second

Following reentry (K = 0), assume that resonance (G \approx G_{max}) occurs at an altitude where K_R = K₁. If $\overline{\sigma}_0$ r/d \geq

 $(\overline{\sigma}_{O} r/d)_{1}$, lock-in is likely to occur at first resonance provided k > 0 and lock-in will continue at least to K = 1 unless dynamic effects (roll rate and $\overline{\sigma}$ oscillations) or roll damping effects (see Section 4.5.6) cause "break out." Break out will occur at K_R = 1 unless ϕ_{O} is such that \dot{p} and therefore G change sign near λ = 1. Therefore, from Fig. 4-15, break out will occur at K_R = 1 for in plane asymmetry (ϕ_{O} = 0° or 180°) but may not occur for out of plane asymmetries (ϕ_{O} = 90° or 270°). Break out would occur, however, even for the out of plane asymmetry at K₃. To be assured of freedom from lock in at all conditions:

$$\left(\overline{\alpha}_{o}\frac{\mathbf{r}}{\mathbf{d}}\right) \leq \left(\overline{\alpha}_{o}\frac{\mathbf{r}}{\mathbf{d}}\right)_{1}$$
 or $\left(\overline{\alpha}_{o}\frac{\mathbf{r}}{\mathbf{d}}\right)_{2}$,

whichever is the more restrictive. The subscript 2 refers to the second resonance condition.

Neglecting a change in roll rate during reentry, it may be shown that $|1 - K_2| \ge |1 - K_1|$. Since the slope of the $\overline{\alpha_0}$ r/d curve for K = 0 to 1 is the negative of the slope for K > 1, the $(\overline{\alpha_0} r/d)_1$ limit is always the more restrictive condition. Therefore, from the resonance standpoint, it is desirable to select a low value of p_E so that first resonance occurs at as high an altitude as possible. The maximum asymmetry allowable is obtained by evaluating Eq. (4.5-61) at K = 0 so that to avoid lock in:

 $\overline{\alpha}_{0} \frac{r}{d} < \frac{F_{1}}{k} \qquad (4.5-63)$

In Eq. (4.5-61), the asymmetries are given in terms of $\overline{\alpha}$ and r. The asymmetric moment coefficient is a better parameter for some purposes since the moment coefficient is a direct function of the asymmetry whereas $\overline{\alpha}_0$ depends also upon $C_{N_{\alpha}}$ and vehicle stability (ΔX). The components of $\overline{\alpha}$ are shown in the sketch on page 192.

- 191 -
THE JOHNS HOPKINS UNIVERSITY APPLIED PHYSICS LABORATORY SILVER SPRING MARYLAND



The component of $\overline{\alpha_0}$ resulting from the center of gravity offset is always directed away from the center of gravity, whereas the component resulting from asymmetric moments may be in any plane. The components of $\overline{\alpha_0}$ are (see Eqs. (4.5-27), (4.5-28), and (4.5-29)):

$$\overline{\alpha}_{o_{c,g.}} = \frac{C_{A} \frac{1}{d}}{C_{N_{\alpha}} \frac{\Delta X}{d}}$$
(4.5-64)

$$\overline{\alpha}_{o_{M}} = \frac{C_{m_{o}}}{C_{N_{\alpha}} \frac{\Delta X}{d}}$$
(4.5-65)

where

(4,5-66)

- 192 -

 $\overline{C}_{m_o} = \sqrt{C_{m_o}^2 + C_{n_o}^2}$

THE JOHNS HOPKINS UNIVERSITY APPLIED PHYSICS LABORATORY SILVER SPRING MARYLAND

For $\phi_0 = 0^\circ$, $\overline{\alpha}_{0M}$ must be directed toward the center of gravity and $\overline{\alpha}_{0M} > \overline{\alpha}_{0_{C,g}}$. Then:

$$\overline{Z}_{0} = \frac{\overline{C}_{m_{0}} - C_{A} \frac{1}{d}}{C_{N_{\alpha}} \frac{\Delta X}{d}}$$
 (4.5-67)

For $\phi_0 = 180^{\circ}$:

$$\widetilde{\alpha}_{0} = \frac{\widetilde{C}_{m_{0}} + C_{A} \frac{r}{d}}{C_{N_{\alpha}} \frac{\Delta X}{d}} \qquad (4.5-68)$$

For $\phi_0 = 90^\circ$ or 270°:

$$\overline{\alpha}_{o} = \left[\left(\frac{\overline{C}_{m_{o}}}{C_{N_{\alpha}} \frac{\Delta \overline{X}}{d}} \right)^{2} - \left(\frac{C_{A} \frac{\overline{r}}{d}}{C_{N_{\alpha}} \frac{\Delta \overline{X}}{d}} \right)^{2} \right]^{1/2}.$$
 (4.5-69)

Usually, $\overline{\alpha}_{oc. g.} << \overline{\alpha}_{OM}$, so that Eq. (4.5-69) is given approximately by the expression:

$$\overline{\alpha}_{o} = \frac{\overline{C}_{m_{o}}}{C_{N_{\alpha}} \frac{\Delta X}{d}} \left[1 - \frac{1}{2} \left(\frac{C_{A}}{\overline{C}_{m_{o}}} \frac{r}{d} \right)^{2} \right]. \quad (4.5-70)$$

Substituting for $\overline{\alpha}_0$, Eq. (4.5-61) becomes:

$$F_{A} < F_{2} \frac{\Delta X}{d} \frac{(1 - K_{R})}{k}$$
 (4.5-71)

- 193 -

THE JOHNS HOPKING UNIVERSITY APPLIED PHYSICS LABORATORY BILVER SPRING, MARYLAND

where

$$F_{A} = \overline{C}_{m_{o}} \frac{r}{d} \left[1 - \frac{C_{A}}{\overline{C}_{m_{o}}} \frac{r}{d} \right] \quad \text{fcr } \phi_{o} = 0^{\circ}$$

$$= \overline{C}_{m_{o}} \frac{r}{d} \left[1 + \frac{C_{A}}{\overline{C}_{m_{o}}} \frac{r}{d} \right] \quad \text{for } \phi_{o} = 180^{\circ}$$

$$= \overline{C}_{m_{o}} \frac{r}{d} \left[1 - \frac{1}{2} \left(\frac{C_{A}}{\overline{C}_{m_{o}}} \frac{r}{d} \right)^{2} \right] \quad \text{for } \phi_{o} = 90^{\circ} \text{ or } 270^{\circ}.$$

$$F_{2} = C_{N_{o}} F_{1} \quad (4.5-73)$$

The limiting asymmetry values may be obtained for any in plane or out of plane asymmetry using Eq. (4.5-71) with the appropriate value of k given in the table in Section 4.5.5. Note however that $(1 - K_R)/k$ must be positive.

The possibilities for lock-in are shown using the roll amplification diagrams shown c. page 195.

At reentry, for a positive roll, $\lambda \to +\infty$ so that $G \to 0$ and the roll rate remains constant. As the altitude decreases, ω_A increases and therefore λ decreases. As $\lambda \to 1$, G is no longer negligible and, if the \dot{p}_p produced at points a ($\phi_0 = 0^\circ$, 90°, or 270°) is equal to or greater than $\dot{\omega}_A$, lock-in may oc-ur at some value of λ between a and b where G is just large enough to satisfy Eq. (4.5-55). As the maxir um dynamic pressure (K - 1) is approached, $\dot{\omega}_A$ decreases, becoming zero at K = 1, and then it becomes negative. As a result p must reverse sign if lock-in continues. Neither the $\phi_0 = 0^\circ$ nor $\phi_0 = 90^\circ$ asymmetry case can produce the \dot{p} characteristics required

- 194 -

THE JOHNS HOPKINS UNIVERSITY APPLIED PHYSICS LABORATORY SILVER SPRING MARYLAND



to satisfy the $\dot{\omega}_A$ variation, and lock-in cannot be maintained at K > 1. However, lock-in may continue to exist for $\phi_0 = 270^\circ$ until the negative prequired for lock-in is greater than that given by point c. For the $\phi_0 = 180^\circ$ case, lock-in cannot occur since k (or G_{max}) is < 0. The vehicle would pass through first resonance with a slight reduction in roll rate. Both p and $\dot{\omega}_A$ tend to cause a rapid transit (both effects tend to reduce λ) through the resonance condition. As ω_A increases, λ decreases to some low value and p remains nearly constant. However as ω_A decreases at K > 1, λ increases. Lock-in at this time (second resonance) requires a negative p and may occur between points b and c if p at c is large enough. For $\phi_0 = 90^\circ$, the shape of the G curve is such that lock-in

- 195 -

F

could occur at second resonance between b and c. However, if the torque at c is large enough to cause lock-in and if the asymmetry existed at reentry, then lock-in would have occurred between a and b at first resonance, and λ would never be less than 1 + D. Therefore lockin at second resonance would not occur. If, on the other hand, the asymmetry developed during reentry and after first resonance, then second resonance lock-in for $\emptyset_0 =$ 90° is possible.

Note that lock-in may occur only when dG/d1 is negative since, for positive slopes, λ is unstable for reasons discussed for λ_e stability in Section 4.5.4.

The most restrictive limitation occurs for first resonance lock-in with the in plane asymmetry with $\phi_0 = 0$. For this case, with first resonance near K = 0, if lock-in is to be avoided;

$$\mathbf{F}_{A} = \overline{\mathbf{C}}_{m_{o}} \frac{\mathbf{r}}{\mathbf{d}} \left[1 - \frac{\mathbf{C}_{A}}{\overline{\mathbf{C}}_{m_{o}}} \frac{\mathbf{r}}{\mathbf{d}} \right] < \mathbf{F}_{2} \frac{\Delta \mathbf{X}}{\mathbf{d}} \quad (4.5-74)$$

or
$$\frac{C_A}{\overline{C}_{m_o}} \frac{r}{d} \ll 1$$
;
 $\overline{C}_{m_o} r \ll dF_2 \frac{\Delta X}{d}$. (4.5-75)

From Eqs. (4.5-62), (4.5-73), and (4.5-75), note that the allowable asymmetry is governed by the following parameters:

- 196 -

, topping a strange rape.

THE JOHNE HOPKING UNIVERSITY APPLIED PHYSICS LABORATORY Bilver Spring Maryland

		tends to be low for:	
Parameter	Related Factor		
Vehicle stability margin	∆X/d		
Resonance altitude	1 - κ _R	κ _R -1	
Plane of asymmetries	k	0° or 180° (compared to 90° or 270°)	
Aerodynamic coefficients	$C_{N_{\alpha}}$ and $C_{m_{q}}$	Low C _N and C _{mq}	
Type of trajectory	sin y	Low γ (shallow reentry angles)	
Atmosphere	Н	High H	
Vehicle slenderness	${}^{I}x^{I}y^{and} {}^{C}m_{q}$	Low $\mathbf{I}_{\mathbf{X}}^{I} \mathbf{I}_{\mathbf{Y}}^{And} \mathbf{C}_{\mathbf{n}_{\mathbf{q}}}$	
Mass distribution factor	I _Y /md ²	Low Ly/md ²	
Vehicle size	d	Low d	

Since there are so many variables, it is not possible to present generalized curves of allowable asymmetry. However several observations are noteworthy. Consider a family of bodies differing only by a scale factor. Values of $C_{N_{\alpha}}$, $C_{m_{q}}$, I_X/I_Y , and I_Y/md^2 will be approximately the same for all bodies. If it is further specified that all bodies have the same stability margin (same $\Delta X/d$), then for a given atmosphere and trajectory, Eqs. (4.5-62), (4.5-73), and (4.5-75) show that the allowable C_{m_o} r is proportional to d^2 . Therefore, small bodies may be particularly sensitive to mass and aerodynamic asymmetries.

Note also that for a given allowable F_A the allowable \overline{C}_{m_0} (Eq. (4.5-72)) has a minimum value for $\phi_0 = 0^\circ$, 90°, or 270°. For $\phi_0 = 0^\circ$, the minimum occurs at:

$$\frac{r}{d} = \frac{1}{2} \frac{\overline{C}_{m}}{\overline{C}_{A}}$$

- 197 -

THE JOHNS HOPKINS UNIVERSITY APPLIED PHYSICS LABORATORY BILVER SPRING MARYLAND

and the minimum \overline{C}_{m_0} is given by:

$$\overline{C}_{m_0} < 2 \left[C_A \frac{\Delta X}{d} F_2 (1 - K_R) \right]^{1/2}$$
 (4.5-76)

For $\oint_{\Omega} = 90^{\circ}$ or 270°, the minimum occurs at:

$$\frac{r}{d} = \sqrt{2/3} \frac{\overline{C}_{m_0}}{C_A}$$
 (4.5-77)

and the minimum value is given by:

$$\overline{C}_{m_o} < 2.29 \left[C_A \frac{\Delta X}{d} F_2 (1 - K_R) \right]^{1/2}$$
. (4.5-78)

Therefore for a fixed value of F_A the minimum value for \overline{C}_{m_0} for an out of plane asymmetry is about 15% greater than that for an in plane asymmetry.

4.5.6 Effect on Lock-in of Roll Damping and Pure Roll Torque

The roll damping, M_p , and pure asymmetric roll torque, ℓ_o , terms in Eq. (4.5-33) are usually negligible. However, under some conditions these terms could be significant. For example, at usual roll rates M_p is small, but if lock-in occurs and drives the roll rate to high values, M_p may become a significant torque and may, in fact, break the lock-in condition.

Rewriting Eq. (4.5-36) to include the M_p and l_o terms:

p = G' p_omax

(4.5-79)

Section Section

- 198 -

THE JOHNE HOPKING UNIVERSITY APPLIED PHYSICS LABORATORY SILVER SPRING MARYLAND

τ,

where

$$G' = G + G_1 + G_2$$
 (4.5-80)

The amplification factor G is defined by Eq. (4.5-37). The factor G_1 is defined by the equation:

$$G_{1} = \frac{M_{p}p}{I_{X}p_{o}}$$
(4.5-81)
$$= \frac{C_{i}p}{C_{N_{\alpha}}\overline{\alpha}_{o}\frac{r}{d}} .$$

Using Eq. (4, 5-21) to eliminate p:



and using Eq. (4.3-23) for ω_A :

$$G_{1} = \left[\frac{-\Delta XK \sin \gamma}{2HC_{A}k_{1}}\right]^{1/2} \frac{C_{\mu}}{\overline{\alpha}_{o} \frac{r}{d}} \lambda = a\lambda . \qquad (4.5-82)$$

The value of a is always negative. Therefore if the G curve is drawn, the G' = G + G₁ (assuming G₂ = 0) values are the ordinates measured from a λ axis which is rotated counterclockwise through an angle, ϵ , where:

 $\tan \epsilon = a$, (4.5-83)

- 199 -

THE JOHNS MORKING UNIVERSITY APPLIED PHYSICS LABORATORY SILVER SPRING MARYLAND



Similarly:

$$G_2 = \frac{l_o}{I_x p_{o_{\text{max}}}} = \frac{C_{l_o}}{C_{N_{ij}} \bar{\alpha}_o \frac{r}{d}}$$
 (4.5-84)

1

Į

of many defension in the second

If $G_1 = 0$, the values of $G' = G + G_2$ are the ordinates of the G curve measured from a λ axis that is shifted by an amount $\frac{-C_{\ell_0}}{C_{N_{\alpha}} \overline{\alpha}_{0} \frac{r}{d}}$ as sketched (for a negative value of C_{ℓ_0}).



9,9

THE JOHNS HOPKINS UNIVERSITY APPLIED PHYSICS LABORATORY Silver Spring. Maryland



Suppose an in plane asymmetry exists and $\overline{\alpha}_0$ and r/d are such that an amplification factor, $G = G_R$ (see Eq. (4.5-58)), could result in a lock-in condition at first resonance if $C_{\ell p} = 0$. Lock-in may then occur at point A in the sketch.



- 201 -

a,

والمتحاجبة

- a coll - Chief and -

the manner and

However, suppose $C_{\ell p} \neq 0$ so that ϵ for resonance at the initial roll rate is given by ϵ_1 . The resonance condition then is obtained by drawing the line BC as shown, giving a lock-in condition at B. As resonance continues, K increases and therefore ϵ increases. If ϵ exceeds the value shown by ϵ_2 , the torque required to sustain resonance is unavailable and break out occurs.

Using Eqs. (4.5-58) and (4.5-60), and noting that $\tan \epsilon = \frac{G_{max} - G_R}{\lambda_G_{max}} \approx G_{max} - G_R$, it may be shown that roll

damping will cause break out when the following inequality is satisfied:

$$C_{\ell_{p}} \geq \frac{2(H/d)C_{A}k_{1}\overline{\alpha}_{o}\frac{r}{d}}{\left[k_{1}-C_{m_{q}}\right]\sin\left(-\gamma\right)}\frac{k}{K_{R}} - C_{A}\frac{I_{X}}{md^{2}}\left(\frac{1-K_{R}}{K_{R}}\right). (4.5-85)$$

Roll damping tends to decrease the possibility of lock-in at first resonance and, if lock-in occurs, tends to increase the possibility of break out. The opposite effects occur at second resonance.

When C_{ℓ_0} exists (but not C_{ℓ_p}), lock-in cannot exist unless

$$G_{\max} - G_R - \frac{C_{\ell_o}}{C_{N_o} \overline{a_o} \frac{r}{d}} > 0 .$$

Using Eqs. (4.5-58) and (4.5-60), lock-in cannot occur at first resonance if C_{I_0} has a sign opposite to that of the initial roll rate and a magnitude given by the following inequality:

- 202 -

THE JOHNS HOPKING UNIVERSITY APPLIED PHYSICS LABORATORY SILVER SPRING MARYLAND

$$C_{\ell_0} > \frac{a_1 \overline{\sigma}_0 \frac{r}{d}}{\sqrt{K_R}} - \frac{a_2 (1 - K_R)}{\sqrt{K_R}}$$
 (4.5-86)

where

$$1 \ge K_R > 0$$

$$a_{1} = \frac{C_{N_{\alpha}}^{k}}{k_{1} - C_{m_{q}}} \sqrt{\frac{2C_{A} \frac{\Delta X}{d} k_{1} \frac{H}{d}}{\sin(-\gamma)}}$$
(4.5-87)

$$a_{2} = \frac{I_{X}}{I_{Y}} \sqrt{\frac{C_{A} \frac{\Delta X}{d} k_{1} \sin(-\gamma)}{2\left(1 - \frac{I_{X}}{I_{Y}}\right)^{2} \frac{H}{d}}}.$$
 (4.5-88)

If the sign of C_{ℓ_0} is the same as that of the initial roll rate, the effect of C_{ℓ_0} is to increase the probability of lock-in; if C_{ℓ_0} and p_E have opposite signs, the effect is to decrease the probability of lock-in.

4, 5, 7 Spin Through Zero Roll Rate

In the previous sections, the problem of roll resonance was considered. A second problem resulting from variations of roll rate during reentry occurs when the roll rate becomes equal to or passes through zero. When p = 0 and an asymmetry, $\overline{\alpha}_0$, exists, the lateral acceleration resulting from $\overline{\alpha}_0$ is fixed in space and the missile moves off course in the direction of F_0 . The impact dispersion produced may be much larger than that required for satisfactory performance. In this section we will consider the conditions for which spin through zero may occur.

Referring to curve 5 in the first sketch in Section 4.5.5, spin down through zero roll rate may occur when

- 203 -

THE JOHNS HOPKINS UNIVERSITY APPLIED PHYSICS LABORATORY SILVER BPRING MARYLAND

> an out of plane component of asymmetry exists and is in a direction to reduce the magnitude of the roll rate. Therefore, one condition for p = 0 is that $180^\circ > \emptyset_0 > 0^\circ$. The second condition is that the magnitude of \mathring{p} is large enough to result in p = 0 prior to impact.

> For the asymmetries of interest in this problem, the vehicle usually passes through first resonance with little change in roll rate so that most of the change in roll rate occurs at λ near zero so that roll amplification is negligible. The roll acceleration, then, is given by Eq. (4.5-35):

$$\dot{p} = \frac{-C_N \sigma}{\gamma S \sigma_0 r \sin \phi_0} , \qquad (4.5-89)$$

For very small angles of attack $(\overline{\alpha}_{n})$:

$$g_{X} = \frac{1}{g} \frac{dV}{dt} = -\frac{C_{A} qS}{W} ,$$

Therefore:

$$\overline{qS} = -\frac{m}{C_A} \frac{dV}{dt} . \qquad (4.5-90)$$

Substituting Eq. (4.5-90) into (4.5-89);

$$\frac{dp}{dt} = \frac{C_N \overline{\alpha_0} r \sin \phi_0}{I_X} \frac{m}{C_A} \frac{dV}{dt} . \qquad (4.5-91)$$

If p and V are assumed to be the only dependent variables, Eq. (4.5-91) may be integrated to give:

$$p - p_E - \Delta p = \frac{C_N \alpha_J^{r} \sin \phi_0}{C_A I_X} m V_E (\frac{V}{V_E} - 1), \qquad (4.5-92)$$

- 204 -

THE JOHNS HOPKINE UNIVERSITY APPLIED PHYSICS LABORATIORY BILVER BPRING MARYLAND

Substituting for V/V_E using Eq. (4.3-5):

$$\Delta p = -\frac{C_N \frac{\bar{\alpha}_o}{\alpha} \frac{r}{d} \sin \phi_o}{C_A (I_X / m d^2)} \frac{V_E}{d} (1 - e^{-K/2}) . \qquad (4.5-93)$$

If $|\Delta p| > p_E$ at K = K_{SL}, spin through zero will occur prior to impact. The altitude (or K) at which zero roll rate occurs may be obtained from Eq. (4.5-93):

$$e^{-K/2} = 1 - \frac{p_E C_A \frac{I_X}{md^2}}{C_N \overline{\alpha}_o \frac{r}{d} \frac{V_E}{d} \sin \theta_o}$$
(4.5-94)

or

í

1

$$K = -2 \ln \left[1 - \frac{p_E C_A \frac{I_X}{md^2}}{C_N \alpha^0 \sigma^{\frac{r}{d}} \frac{V_E}{d} \sin \phi_0} \right] \cdot (4, 5-95)$$

The minimum asymmetries which may cause zero roll rate occur for $\phi_0 = 90^\circ$. For this case, Eq. (4.5-95) may be written in terms of the moment coefficient using Eq. (4.5-70):

$$K = -2 \ln \left\{ 1 - \frac{p_E C_A \frac{I_X}{md^2} \frac{\Delta X}{d}}{\overline{C}_{m_o} \frac{r}{d} \frac{V_E}{d} \left[1 - \frac{1}{2} \left(\frac{C_A}{\overline{C}_{m_o}} \frac{r}{d} \right)^2 \right]} \right\}$$
(4.5-96)

If spin through zero is to be avoided for the entire trajectory, Eqs. (4.5-95) and (4.5-96) indicate that the following inequalities must be satisfied:

- 205 -

44

14

THE JOHNS HUPKINS UNIVERSITY APPLIED PHYSICS LABORATORY SILVIA SPRING MARYLAND

$$\overline{\alpha}_{0} \frac{\mathbf{r}}{\mathrm{d}} \leq \frac{\frac{\mathbf{p}_{E}}{\mathbf{V}_{E}} \frac{\mathbf{I}_{X}}{\mathrm{md}^{2}} \frac{\mathbf{C}_{A}}{\mathbf{C}_{N}}}{1 - \exp\left(\frac{\mathbf{P}_{S, L}}{2\beta \sin \gamma'}\right)}$$
(4.5-97)

or in terms of \overline{C}_{m_0} :

$$\mathbf{F}_{A} = \overline{\mathbf{C}}_{\mathbf{m}_{O}} \frac{\mathbf{r}}{\mathbf{d}} \left[1 - \frac{1}{2} \left(\frac{\mathbf{C}_{A}}{\overline{\mathbf{C}}_{\mathbf{m}_{O}}} \frac{\mathbf{r}}{\mathbf{d}} \right)^{2} \right] \leq \frac{\frac{\mathbf{p}_{E}^{d}}{\mathbf{V}_{E}} \mathbf{C}_{A} \frac{\mathbf{I}_{X}}{\mathbf{m}d^{2}} \frac{\Delta X}{\mathbf{d}}}{1 - \exp\left(\frac{\mathbf{P}_{S}}{2\beta \sin \gamma}\right)} \quad (4.5-98)$$

Using the assumptions listed in Section 4.5.5, page 197, note that the allowable \overline{C}_{m_0} r is proportional to d². Therefore, as in the case of the roll resonance problem, small vehicles are particularly susceptible to the spin through

zero problem.

The probability of encountering the spin through zero problem on a given vehicle is highly dependent upon the missile-to-missile variation in ϕ_0 . If the asymmetries are such that ϕ_0 is always 90°, and if $\overline{\alpha}_0$ r/d always exceeds the value given by Eq. (4.5-97), then all the missiles will spin through zero. On the other hand, if ϕ_0 always lies between 180° and 360°, the probability of encountering spin through zero is zero regardless of the value of $\overline{\alpha}_0$ r/d. If the missile-to-missile variation in ϕ_0 is random, the probability of encountering spin through zero is given by:

$$\overline{\mathbf{P}} = \frac{\Lambda}{\pi}$$

- 206 -

Same Buck to the Spingle

THE JOHNS HOPKING UNIVERSITY APPLIED PHYSICS LABORATORY SILVER SPRING MARYLAND

「「「「「「「「「「」」」」」」

í

where Λ is an angle between 0 and $\pi/2$ radians defined by:

$$\Lambda = \cos^{-1} \left[\frac{\left(\overline{\alpha}_{o} r/d \right)_{min}}{\overline{\alpha}_{o} r/d} \right]$$

and $(\overline{\alpha}_{0} r/d)_{\min}$ is the value of $\overline{\alpha}_{0} r/d$ given by Eq. (4.5-97). For this assumption, the probability of encountering spin through zero varies from 0 at $\overline{\alpha}_{0} r/d = (\overline{\alpha}_{0} r/d)_{\min}$ to 50% at $\overline{\alpha}_{0} r/d \gg (\overline{\alpha}_{0} r/d)_{\min}$.

4.5.8 Effect on Products of Inertia

The products of inertia are defined by the following equations:

$$\mathbf{I}_{\mathbf{X}\mathbf{Y}} = \Sigma \mathbf{m}_{\mathbf{i}} \mathbf{X}_{\mathbf{i}} \mathbf{Y}_{\mathbf{i}} \tag{4.5-99}$$

$$I_{XZ} = \Sigma m_i X_i Z_i \qquad (4.5-100)$$

$$\mathbf{I}_{\mathbf{Y}\mathbf{Z}} = \boldsymbol{\Sigma}\mathbf{m}_{\mathbf{i}}\mathbf{Y}_{\mathbf{i}}\mathbf{Z}_{\mathbf{i}}$$
(4.5-101)

The I_{YZ} term is negligible for most reentry bodies and is omitted from this analysis. If I_{XY} and I_{XZ} are not zero, the principal axes pass through the center of gravity but are rotated through an angle δ measured clockwise from the body centerline. For small asymmetries, the rotation angles are (for $I_Y = I_Z$):

$$\delta_{\beta} = \frac{I_{XY}}{I_{Y} - I_{X}}$$
(4.5-102)

$$\delta_{\alpha} = \frac{-I_{XZ}}{I_{Y} - I_{X}}$$
 (4.5-103)

- 207 -

THE JOHNE HOPKINE UNIVERSITY APPLIED PHYSICS LABORATORY BILVER SPRING MARYLAND

$$\delta = \sqrt{\delta_{\beta}^2 + \delta_{\alpha}^2} \qquad (4.5-104)$$

where δ_{β} is a rotation about the Z axis and δ_{α} is a rotation about the Y axis,

Consider a mass asymmetry in which $I_{XZ} = 0$ but $I_{XY} \neq 0$, as illustrated by the sketch.



Assume the vehicle is symmetrical except for two small masses, m_1 , located at (X_1, Y_1) and $(-X_1, -Y_1)$ so that $I_{XY} = 2m_1X_1Y_1$. If the missile is now rotated about the X axis, the centrifugal force acting on each m_1 produces a yawing moment about the Z axis.

Returning to the case where $I_{XY} \neq 0$, $I_{XZ} \neq 0$, and the mass asymmetry forces are much greater than the aerodynamic forces ($\lambda \rightarrow \infty$), and retaining only the larger asymmetry terms (p is moderately large but I_{XY} , I_{XZ} , q, r, b, d, and rare assumed to be sufficiently small that the square or product of two small terms may be neglected) of Eqs. (4.5-1), (4.5-2), and (4.5-3), the equations of motion become:

- 208 -

THE JOHNS HOPKINS UNIVERSITY APPLIED PHYSICS LABORATORY SILVER SPRING MARYLAND

$$\dot{\mathbf{p}} = \mathbf{0}$$
 (4.5-105)

$$\beta = r\omega_g - p^2 \frac{I_{XZ}}{I_Y}$$
 (4.5-106)

$$t = -q\omega_g + p^2 \frac{I_{XY}}{I_Y}$$
 (4.5-107)

where $\omega_{g} = p\left(1 - \frac{I_{X}}{I_{Y}}\right)$.

Differentiating Eq. (4.5-106):

$$\ddot{q} = \omega_{g} t$$
. (4.5-108)

Using Eqs. (4.5-107) and (4.5-108):

$$\ddot{q} + \omega_g^2 q = \omega_g p^2 \frac{I_{XY}}{I_V} \cdot \qquad (4.5-109)$$

The solution to this differential equation is:

$$q = C_1 \sin \omega_g t + C_2 \cos \omega_g t + p \frac{I_{XY}}{I_Y - I_X}$$
 (4.5-110)

$$\dot{q} = C_1 \omega_g \cos \omega_g t - C_2 \omega_g \sin \omega_g t . \qquad (4.5-111)$$

From Eqs. (4.5-106) and (4.5-111):

r = C₁ cos
$$\omega_{g}^{t}$$
 - C₂ sin ω_{g}^{t} + p $\frac{I_{XZ}}{I_{Y} - I_{X}}$ (4.5-112)

- 209 -

ŝ

THE JOHNS HOPKINS UNIVERSITY APPLIED PHYSICS LABORATORY SILVER SPRING MARYLAND

From Eqs. (4.5-110) and (4.5-112):

$$\left(q - p \frac{I_{XY}}{I_Y - I_X}\right)^2 + \left(r - p \frac{I_{XZ}}{I_Y - I_X}\right)^2 = C_1^2 + C_2^2 .$$

Therefore, the q - r plot is a circle of radius $\sqrt{C_1^2 + C_2^2}$ with center at q = p $\frac{I_{XY}}{I_Y - I_X}$, r = p $\frac{I_{XZ}}{I_Y - I_X}$. In other words, the telemetry traces for a body having nonzero I_{XY} and I_{XZ} show a q which oscillates about the value q = p $\frac{I_{XY}}{I_Y - I_X}$ and an r which oscillates about r = p $\frac{I_{XZ}}{I_Y - I_X}$. The constants C_1 and C_2 may be evaluated from given conditions at t = 0.

$$C_1 = r_i - p \frac{I_{XZ}}{I_Y - I_X}$$
 (4.5-113)

$$C_2 = q_1 - p \frac{I_{XY}}{I_Y - I_X}$$
 (4.5-114)

Equations (4.5-113) and (4.5-114) may be used to evaluate I_{XY} and I_{XZ} from flight test data if p, q, and r are known accurately.

In space, where the aerodynamic forces are zero, the <u>principal X axis</u> moves along a cone. The cone halfangle is given by the equation (see Eq. 4.4-23):

$$\tan \phi = \frac{\sqrt{C_1^2 + C_2^2}}{p} \frac{I_Y}{I_X} . \qquad (4.5-115)$$

- 210 -

The period of the rotation about the space cone is given by Eq. (4.4-25). Since the body centerline rolls around the principal axis, the angle of at tack oscillates with a relatively long period (τ_s) and a peak-to-peak amplitude of $2\emptyset$ with a superimposed short period (1/p) oscillation with peak-to-peak amplitude of 2δ .

The constant shift in q and/or r resulting from nonzero products of inertia may be interpreted in terms of angle of attack using the following equations (see Ref. 4-9).

$$\beta_{g} = \frac{q}{p} = \frac{I_{XY}}{I_{Y} - I_{X}}$$
(4.5-116)

$$\alpha_{\delta} = \frac{r}{p} = \frac{I_{XZ}}{I_{Y} - I_{X}}$$
(4.5-117)

When the missile descends to altitudes where the aerodynamic forces are no longer negligible, assuming that α_0 and β_0 may exist as well as α_0 and β_0 , the total angle of attack is given by the equation (see Ref. 4-9):

$$|\overline{\alpha}| = A \left[(\beta_0 + \lambda^2 \beta_{\delta})^2 + (\alpha_0 + \lambda^2 \alpha_{\delta})^2 \right]^{1/2} \quad (4.5 - 118)$$

where A is defined by Eq. (4.5-16). The direction of the trim force is defined by the equation:

$$\psi = \psi_0 + \Delta \psi \qquad (4.5 - 119)$$

where

Nykana kalendari di kalendari kalendari kalendari kalendari di kalendari di kalendari di kalendari di kalendari

$$\tan\psi_{0} = \frac{-\alpha_{0}}{-\beta_{0}} \qquad (4.5-120)$$

if α_0 and $\beta_0 \neq 0$.

$$\tan \psi_{0} = \frac{I_{XZ}}{I_{XY}}$$
(4.5-121)

- 211 -

THE JOHNS HOPKING UNIVERSITY APPLIED PHYSICS LABORA'I'ORY BILVER SPRING MARYLAND

if
$$\boldsymbol{\alpha}_{0} = \boldsymbol{\beta}_{0} = 0$$
, and

$$\Delta \psi = \tan^{-1} \left(\frac{\alpha_0 + \lambda^2 \alpha_0}{\beta_0 + \lambda^2 \beta_0} \right) - \tan^{-1} \left(\frac{D\lambda}{1 - \lambda^2} \right) - \tan^{-1} \frac{\alpha_0}{\beta_0}$$
or
$$\Delta \psi = -\tan^{-1} \left(\frac{D\lambda}{1 - \lambda^2} \right)$$
(4.5-122)

if either α_0 and β_0 or α_0 and $\beta_0 = 0$.

The proper quadrant for ψ_0 may be obtained from the signs of the numerator and denominator in Eq. (4.5-120). For $\alpha_0 = \beta_0 = 0$, the trim angle of attack is just that resulting from products of inertia.

Let

$$\overline{\alpha}_{\delta} = \sqrt{\alpha_{\delta}^2 + \beta_{\delta}^2} \qquad (4.5-123)$$

Then, the amplification factor in terms of $\overline{\alpha}_{\delta}$ is given by:

Amplification factor =
$$\left| \frac{\overline{\alpha}}{\overline{\alpha}_{0}} \right| = \lambda^{2} A$$
 (4.5-124)

- 212 -

Beneficials and

The amplification factor is zero at $\lambda = 0$, has the same amplification as that for aerodynamic trim at $\lambda = 1$ (namely, A), and approaches unity as $\lambda \rightarrow \infty$. Therefore the products of inertia may be important at λ near or greater than unity and, in combination with the center of gravity offset, could result in resonance lock-in. The effect may either augment or diminish the effect of other asymmetries, depending upon the signs and magnitudes (see discussion below) of IXY and IXZ. However, the product of inertia asymmetry in combination with the center of gravity offset cannot result in spin through zero since the roll torque at p = 0 is zero.

Consider a vehicle having a positive roll rate at reentry and a combined in-plane asymmetry consisting of a center of gravity offset along the +Y axis and a positive product of inertia I_{XY} . At reentry ($\lambda \rightarrow \omega$) a positive I_{XY} results in a positive β (Eq. (4.5-116)) or a force F_{∞} , along the -Y axis. Since F rotates clockwise as λ decreases, F_Z is always negative and the roll acceleration resulting from F is always positive, and therefore lock-in at first resonance is a possibility.



- 213 -

THE JOHNE HOPKING UNIVERSITY APPLIED PHYSICS LABORATORY SILVER SPRING MANYLAND

Solving Eqs. (4, 5-1) and (4, 5-2) simultaneously and retaining just the \dot{p} contributions resulting from aerodynamic forces:

$$\dot{p} = \frac{\ell}{I_X} + \frac{I_XY}{I_X} \frac{m}{I_Y}$$

Substituting for l and m using Eqs. (4, 5-4) and (4, 5-5);

$$\dot{p} = -\frac{F_Z Y_{c.g.}}{I_X} + \frac{I_{XY}}{I_X} \frac{F_Z \Delta X}{I_Y}$$

Note that presulting from the second term on the right side of the equation is negative since F_Z is always negative. Similarly, if a negative I_{XY} had been selected F_Z would always be positive. Therefore this term results in a negative p regardless of the sign of I_{XY} and, for positive roll rates at reentry, tends to prohibit first resonance lock-in but tends to cause second resonance lock-in. In a manner similar to that used in deriving the criterion for lock-in for an aerodynamic asymmetry and a center of gravity offset (Eq. (4, 5-61)), the criterion for lock-in with a product of inertia and a center of gravity offset is:

$$\overline{\alpha}_{\delta} \frac{\mathbf{r}}{\mathbf{d}} - \overline{\alpha}_{\delta}^{2} \left(1 - \frac{\mathbf{I}_{\mathbf{X}}}{\mathbf{I}_{\mathbf{Y}}} \right) \frac{\Delta \mathbf{X}}{\mathbf{d}} - \mathbf{F}_{1} \left[1 - \mathbf{K}_{\mathbf{R}} \right] \ge 0$$

This inequality is identical to the inequality for an aerodynamic asymmetry and a center of gravity offset (Eq. (4.5-61)) only for the particular case of a neutrally stable vehicle ($\Delta X/d = 0$), which is not of practical interest. For any statically stable vehicle (positive $\Delta X/d$), $\overline{\alpha_0}$ r/d is always greater than $\overline{\alpha_0}$ r/d for a given F₁ and K_R. Solving for $\overline{\alpha_0}$ at the boundary (the above equation set to zero) using the binomial theorem:

- 214 -

THE JOHNS HOPKING UNIVERSITY APPLIED PINYSICS LABORATORY SILVER SPRING MARYLAND

$$\overline{\mathbf{x}}_{\delta} = \left(\mathbf{F}_{3}\frac{\mathbf{r}}{\mathbf{d}}\right) \left[1 \pm \sqrt{1 - \frac{\mathbf{F}_{4}\left|1 - \mathbf{K}_{R}\right|}{\left(\mathbf{F}_{3}\frac{\mathbf{r}}{\mathbf{d}}\right)^{2}}}\right]$$
(4.5-125)

where

$$\mathbf{F}_{3} \sim \frac{1}{2\left(1 - \frac{\mathbf{I}_{X}}{\mathbf{I}_{Y}}\right)\frac{\Delta \mathbf{X}}{\mathbf{d}}}$$
$$\mathbf{F}_{4} = 2\mathbf{F}_{1}\mathbf{F}_{3} \cdot \mathbf{I}_{3}$$

Note that lock-in cannot occur unless:

$$\mathbf{F}_{3}\frac{\mathbf{r}}{\mathbf{d}} \geq \sqrt{\mathbf{F}_{4} |1 - \mathbf{K}_{R}|} \quad .$$

For the minimum value of $F_3 r/d$,

$$\overline{\alpha}_{\delta} = F_3 \frac{r}{d} = \sqrt{F_4 |1 - K_R|}$$

Another lock-in requirement is that:

$$\overline{\alpha}_{\delta_{\min}} \leq \overline{\alpha}_{\delta} \leq \overline{\alpha}_{\delta_{\max}}$$

where \bar{a}_{0min} and \bar{a}_{0max} are given by Eq. (4, 5-125) using

the negative and positive signs of the radical term, respectively. The general requirements for lock-in for this type of combined asymmetry are as shown in the sketch for a particular value of $K_{\rm B}$.

THE JOHNS HOPKING UNIVERSITY APPLIED PHYSICS LABORATORY SILVER SPEING MARYLAND



Note that $(F_3 r/d)_{min}$ decreases from $\sqrt{F_4}$ at reentry $(K_R = 0)$ to 0 at maximum dynamic pressure $(K_R = 1)$ and then increases to $\sqrt{F_4} | 1 - K_{S, L_*} |$ at impact. Current reentry vehicles are designed such that first resonance occurs at $K_{R} \approx 0$, so that $| 1 - K_R | \approx 1$. Assuming p does not change during reentry, it may be shown that $|1 - K_R| > 1$ at second resonance. Therefore lock-in is not possible at first or second resonance for any value of $\overline{\alpha}_5$ if the vehicle can be constructed so that:

- 216 -

THE JOHNS HORKINS UNIVERSITY APPLIED PHYSICS LABORATORY BILVER BRING MARTLAND

$$\frac{r}{4} \leq \frac{\sqrt{F_4}}{F_3} = \sqrt{\frac{2F_1}{F_3}} - \epsilon$$

* Substituting for F_1 and F_3 , this inequality becomes:

$$\frac{\mathbf{r}}{\mathrm{d}} \leq \left\{ 2 \frac{\mathbf{I}_{\mathbf{X}}}{\mathbf{I}_{\mathbf{Y}}} \frac{\Delta \mathbf{X}}{\mathrm{H}} \left[\left(1 - \frac{\mathbf{I}_{\mathbf{X}}}{\mathbf{I}_{\mathbf{Y}}} \right) \frac{\mathbf{I}_{\mathbf{Y}}}{\mathrm{md}^{2}} - \frac{\mathbf{C}_{\mathrm{m}_{\mathbf{q}}}}{\mathbf{C}_{\mathrm{N}_{\boldsymbol{\alpha}}}} \right] \sin \left(-\gamma \right) \right\}^{1/2}$$

For a given shape, I_X/I_Y , I_Y/md^2 , and $C_{m_q}/C_{N_{c2}}$ tend to be independent of size. Then, for a given shape and $\Delta X/d$, the allowable center of gravity offset varies as $d^{3/2}$ and $[\sin(-\gamma)]^{1/2}$. Therefore, small bodies flying shallowangle trajectories are more susceptible to lock-in than are large bodies flying steep-angle trajectories.

4.5.9 Unequal Pitch and Yaw Moment of Inertia

For a body whose only asymmetry is unequal pitch and yaw moment of inertia, the equations of motion (neglecting small terms) become:

$$bI_{X} = (I_{Y} - I_{Z})qr$$
 (4.5-126)

$$dI_{v} = m - pr(I_{v} - I_{z})$$
 (4.5-127)

$$h_{Z} = n - pq(I_{Y} - I_{X})$$
 (4.5-128)

The significance of the $(I_Y - I_Z)$ qr term in Eq. (4.5-126) is shown (for $I_Z > I_Y$) in the sketch. The body has mass symmetry except for two small equal masses located on the Y axis equally distant (ΔY) from the center of gravity. At some instant, the body has a transverse rate of rotation shown by the vector ω_T . At this time the

- 217 -

THE JOHNS HOPKINS UNIVERSITY APPLIED PHYSICS LABORATORY SILVEE SPRING MARYLAND



body rotates about an axis coincident with $\vec{\omega}_{T}$, and the centrifugal force acting on m_1 produces a couple $-2m_1 \Delta Y^2 qr = (I_Y - I_Z)qr$ about the X axis. The roll torque is negative when $\vec{\omega}_T$ lies in quadrants I and III and positive when $\vec{\omega}_T$ lies in quadrants II and IV. At very high altitudes ($\lambda \rightarrow \infty$), the terms m and n in Eqs. (4.5-127) and (4.5-128) are negligible and the equations for q and r are those given by Eqs. (4.4-18) and (4.4-19):

 $q = \omega_T \cos \omega_g t \qquad (4.5-129)$

$$\mathbf{r} = -\omega_{\mathrm{T}} \sin \omega_{\mathrm{g}} \mathbf{t} \quad (4.5-130)$$

At t = 0, $\vec{\omega}_{T}$ lies along the positive Y axis and, for positive p, rotates counterclockwise with constant amplitude and frequency as t increases. Therefore, the asymmetric mass and the rotation of $\vec{\omega}_{T}$ result in a roll rate oscillation. Two cycles of oscillation in roll rate occur for each revolution of $\vec{\omega}_{T}$ about the Z axis. The equation for the roll rate history is obtained by substituting Eqs. (4.5-129) and (4.5-130) into (4.5-126) and integrating to give:

- 218 -

THE JOHNS HOPKINS UNIVERSITY APPLIED PHYSICS LABORATORY SILVER SPRING MARYLAND

$$\mathbf{p} - \mathbf{p}_{i} = \frac{\omega_{T}^{2}}{2\mathbf{p}_{i}} \left(\frac{\mathbf{I}_{Y} - \mathbf{I}_{Z}}{\mathbf{I}_{Y} - \mathbf{I}_{X}} \right) \frac{\mathbf{I}_{Y}}{\mathbf{I}_{X}} \sin^{2} \omega_{\mathbf{g}_{i}}^{t}$$
(4.5-131)

where

$$\omega_{\mathbf{g}_{i}} = \mathbf{p}_{i} \left(1 - \frac{\mathbf{I}_{\mathbf{X}}}{\mathbf{I}_{\mathbf{Y}}}\right) \quad .$$

Therefore, the roll rate trace in the exoatmosphere oscillates as a squared sine function with a peak-to-peak amplitude of:

$$\left|\frac{\omega_{\mathrm{T}}^{2}}{2\mathsf{p}_{\mathrm{i}}}\left(\frac{\mathsf{I}_{\mathrm{Y}}-\mathsf{I}_{\mathrm{Z}}}{\mathsf{I}_{\mathrm{Y}}-\mathsf{I}_{\mathrm{X}}}\right)\frac{\mathsf{I}_{\mathrm{Y}}}{\mathsf{I}_{\mathrm{X}}}\right|.$$

The frequency of the oscillation is $2\omega_{g_1}$.

When the vehicle descends to altitudes where aerodynamic forces are no longer negligible, the $\vec{\omega}_{\rm T}$ develops a pattern of motion that depends upon the initial conditions and the vehicle's aerodynamic characteristics.

Unless the rotation of $\bar{\omega}_{T}$ about the Z axis becomes very slow, a net change in the magnitude of p is unlikely and the effect on the reentry trajectory is negligible. Under some conditions it has been found (Ref. 4-10) that the direction of rotation $\bar{\omega}_{T}$ about the Z axis actually changed directions during reentry. During the period when the rotation changed sign, a significant reduction in roll rate occurred.

The possibility of encountering difficulty as a result of unequal I_Y and I_Z appears to be remote, and this type of asymmetry is not considered further in this report.

- 219 -

THE JOHNS HOPKINS UNIVERSITY APPLIED PHYSICS LABORATORY BILVER BRING MARYLAND

4, 5, 10 Applicability of Trim Characteristics at Resonance

In preceding subsections, it was assumed that trim conditions are attained at first resonance. However, the resonance condition is approached very rapidly, and large changes in trim occur. Unless the missile can respond very quickly to the changing trim conditions, the missile will pass through resonance without attaining large amplifications in the angle of attack. The weathercock frequency, ω_A , is an indication of the ability of the missile to respond to a change in trim. If ω_A is high the missile response time is short; if ω_A is low the missile response time is long. Since

$$\omega_{A}^{2} = \frac{C_{N} \Delta x \bar{q} s}{I_{Y}} ,$$

 ω_A is proportional to \sqrt{q} . Therefore, no matter how much static stability (ΔX) the missile has, the response time is always long at very high altitudes. If resonance occurs at a sufficiently high altitude, trim conditions are not attained and the loads and lock-in criteria based on trim equations are not valid.

The conditions for which trim theory is valid have not been examined in the literature. In order to provide some insight into this problem, a brief study was made of the lock-in criteria using a six-degree-of-freedom trajectory simulation.

Trajectories were computed for three bodies having the following mass and aerodynamic characteristics: THE JOHNS HOPKINS UNIVERSITY APPLIED PHYSICS LABORATORY SILVER SPRING MARYLAND

Characteristic	Body A	Body B	Body C
CA	0,104	0,104	0,104
C _N a	2	0,88	2
с _т	- 1	- 1	-2. 28
∆X /d	0.114	0.261	0,114
\mathbf{x}^{I}	30	3	3
m	31,1	31.1	31.1
d	3.5	3.5	3.5
^{I}Y	300	3 0	30

Body A is the body used for all examples in this report, and additional characteristics for this body are given in Table 2-1. Bodies B and C have the same value of F_1 (see Eq. (4.5-62)) as has A. Therefore, the lock-in criterion (based on trim conditions) for a combined asymmetry consisting of a center of gravity offset and an aerodynamic trim (see Eq. (4.5-61)) is the same for all three bodies. However, for a given altitude, ω_A is higher for B and C than that for A by a factor of about 3, and therefore, for resonance at high altitudes, B and C are more likely to attain trim conditions than A.

The trajectory initial conditions used in this study are $V_E = 20\ 000\ ft/sec$, $\gamma_E = -30^\circ$, and $h_E = 400\ 000\ feet$. An in plane combined asymmetry was used for the first set of calculations. The asymmetry consisted of a center of gravity offset (r/d = 0.001) along the +Y axis and a positive yaw moment coefficient (C_{n_0}). The altitude for first resonance was varied from about 40 000 to 140 000 feet by changing the initial roll rate. For each initial roll rate, C_{n_0} was increased until lock-in occurred. Out of plane asymmetries were considered using the same

- 221 -

center of gravity offset with positive and negative values of pitch moment coefficient (C_{m_0}). Values of $\overline{\alpha}_0$ are expectly related to C_{n_0} and C_{m_0} (see Eqs. (4, 5-67),

(4, 5-68), and (4, 5-69)).

Three aspects of trim theory are important in the establishment of lock-in criteria.

1. Maximum angle of attack occurs at $\lambda = 1$

$$\left(\text{ or } \omega_{A} = p_{E} \sqrt{1 - \frac{I_{X}}{I_{Y}}} \right).$$

- 2. At $\lambda = 1$ the amplification in angle of attack $(\overline{\alpha}/\overline{\alpha}_0 \text{ or } \overline{\alpha}/\overline{\alpha}_0)$ is 1/D.
- 3. At $\lambda = 1$ the plane of the angle of attack (defined by $\Delta \psi$) has rotated 90° from its value at reentry (K = 0).

The results from the six-degree-of-freedom trajectories for body A with $p_E = 1000^\circ$ /sec (corresponding to first resonance at 89 000 feet)show that none of the three assumptions was valid for this resonance altitude. The magnitude and plane of $\overline{\alpha}$ lagged the trim values and the peak amplification was much less than 1/D. The sketch shows a comparison of the $\overline{\alpha}$ and $\Delta \psi$ time histories as given by trim and simulation results.

The trim $\overline{\alpha}$ history reached a peak value at $\lambda = 1$ where lock-in was assumed to occur and then decreased with increasing time, since D increased with time. The trim value of $|\Delta \psi|$ decreased from 180° to 90° at $\lambda = 1$ and remained at 90° as time increased. The simulation results show that $\overline{\alpha}$ built up to a maximum value but lock-in did not occur. Following $\overline{\alpha}_{max}$, the missile began a coning motion with no large excursions in roll rate. The peak $\overline{\alpha}$ was about one-third of the trim value at the initial first resonance altitude; the peak $\overline{\alpha}$ occurred about 1.5 seconds

- 222 -

THE JOHNS HOPKING UNIVERSITY APPLIED PHYSICS LABORATORY SILVER SPRING MARYLAND



later, or about 15 000 feet lower in altitude than that corresponding to first resonance altitude; the plane of α rotated through an angle of 90° about 0.7 second later, or about 7000 feet lower in altitude than that corresponding to first resonance altitude. The angle of attack amplification effect results in larger values of the asymmetries required for lock-in than those given by trim theory. Under some conditions the lag effects would be similar to lowering the first resonance altitude and, thus, asymmetries required for lock-in based on trim theory would be too high. However, for altitudes where first resonance usually occurs, the amplification effect is predominant and the required asymmetries for lock-in based on trim theory are too low.

- 223 -

The condition required for lock-in is given by Eq. (4.5-58). If trim conditions are attained, then Eq. (4.5-58)may be simplified to Eq. (4, 5-61). Both equations are plotted in Fig. 4-17 for body A for an in plane (k = 1) and an out of plane (k = 1/2) asymmetry. Note that the assumption that trim conditions exist implies very large values of the roll amplification factor G at the higher altitudes. Also shown are the $\overline{\alpha}_{0}$ r/d values required for lock-in obtained from simulation results. At the lower altitudes trim theory provided the correct criterion for lock-in for in-plane asymmetry; at the higher altitudes the values of $\overline{\alpha}_{o}$ r/d required for lock-in obtained from simulation results were somewhat higher than those given by trim theory. In fact, the lock-in boundary is very nearly a constant G contour. The results also indicate that the $\overline{\alpha}_{O}$ r/d value required for lock-in for the out of plane asymmetry is about twice that required for the in plane asymmetry as predicted by trim theory.

Similar trajectories for bodies having in plane asymmetries were calculated for bodies B and C for an altitude of 136 000 feet only. These results (Figs. 4-18 and 4-19) show that bodies B and C achieve an effective roll amplification of about 18 compared to a value of about 12 for body A. Although B and C have the same response, C was much closer to a trim condition than was B. Note that for an altitude of 136 000 feet the trim condition for C corresponds to $G \approx 27$, whereas trim for B corresponds to $G \approx 60$.

These results indicate that ω_A is one factor which affects the attainment of trim but it is not the only factor, and much work is required in order to define conditions for which trim theory is applicable.

Trajectories were also computed for body A for a combined in-plane asymmetry consisting of a product of inertia and a center of gravity offset. In this case α_6 was held constant at 0.636° and r was varied until lock-in occurred. The altitude for first resonance was 89 000 feet. The results show that lock-in occurred at a value of

- 224 -

 $\overline{\alpha}_{0} r/d \approx 2.6 \times 10^{-5}$. If the asymmetry had been an aerodynamic asymmetry and a center of gravity offset, the simulation value of $\overline{\alpha}_{0} r/d$ would have been 1, 2 × 10⁻⁵ (see Fig. 4-17). This ratio of about 2 to 1 between the two types of combined asymmetries is nearly the ratio obtained using trim Eqs. (4, 5-61) and (4, 5-125). Therefore, although trim was not attained for either set of asymmetries, the relative behavior in terms of lock-in criteria was properly preferred using trim equations.

Until additional data become available regarding the applicability of the trim equations, the following comments appear to be valid:

1. Trim equations provide a valid criterion for the lock-in condition provided the roll amplification factor implied by trim is not too high.

2. At resonance altitudes where trim conditions imply very high values of G, trim is not attained, and trim theory predicts asymmetry boundaries that are too low.

3. Values of $\overline{\alpha}_0$ r/d required for lock-in at very high altitudes tend to follow a constant G contour rather than that defined by the trim equation.

4. The limits of G for which trim conditions are attained are probably a function of many variables. For the bodies, trajectory, and initial reentry conditions used in this study, values of G greater than 10 to 20 were not attained. Until more work is done on this problem, trim theory may be used to estimate the asymmetries required for lock-in. However, the results for conditions corresponding to values of G greater than about 20 should be considered as questionable.

5. For an out of plane asymmetry, the required $\overline{\alpha}_0$ r/d for lock-in is larger than that for the in plane asymmetry as predicted by trim theory. Also the relative magnitude of combined asymmetry, $\overline{\alpha}_0$ and r, and combined asymmetry, $\overline{\alpha}_0$ and r, are properly defined by trim equations.

- 225 -

THE JOHNS HOPKINE UNIVERSITY AF BILVER SPRING MARYLAND

n: ir

1

Â

ĭ

1

1

1.



VARIATION OF A AND Δ ψ WITH λ Fig. 4-13

:

のうちのないのでのできょう

- 227 -

THE JOHNS HOPKING UNIVERSITY APPLIED PHYSICS LABORATORY Silver Spring Martiand

. 8







ý

. .

Contraction of the

.

n or the second of the
THE JOHNS HOPKING UNIVERSITY APPLIED PHYSICS LABORATORY SILVER SPRING MARYLAND





1 1

1

li i 5

「「「「「「「」」」」「「「「」」」」」」」」」」」」」」」

1

4

1

- 229 -

THE JOHNE HOPKINE UNIVERSITY APPLIED PHYSICS LABORATORY BILVER BRING MARTLAND

e e



- 230 -

.

A STRATE STRATE

11 **i** i i i

··· .

THE JCHNE HOPKING UNIVERSITY APPLIED PHYSICS LABORATORY SILVER SPRING MARYLAND





- 231 -

Average and the second second

THE JOHNS HOPRING UNIVERSITY APPLIED PHYSICS LABORATORY BILVER BRING MARYLAND



1

A REAL PROPERTY.

.

.



- 232 -

THE JOHNS HOPKINS UNIVERSITY APPLIED PHYSICS LABORATORY Bilver Spring Maryland





- 233 -

Den ska ut. 10 5

THE JOHNS MORKINS UNIVERSITY APPLIED PHYSICS LABORATORY BILVER SPRINS, MARYLANS

1676

l

I

J

.

ាំស

1 10.

2.0

÷ la H

* **We** ...

16.22

j

I

•••

.

÷.

ころというと、このいいとの時間を見たいというかれるのである。 しのいたをまたのである したい

iy i N 10

. ÷., 4.6 EXAMPLES

の数になって

C EXAMPLES

I

i.

And and a second for the State of the State

Preceding page blank

.. . .

- 235 -

and the second second second second second second second second second second second second second second second

THE JOHNS HOPKING UNIVERSITY APPLIED PHYSICS LABORATORY BILVER SPRING MARYLAND

F .

SUMMARY

Examples illustrating the material presented in Section 4 are presented in this subsection. The following specific problems are considered:

Example	1	-	Minimum energy trajectory for non- rotating earth
Example	2	-	Ballistic trajectory for rotating earth
Example	3	-	Body motion (oscillation) character- istics in the excatmosphere
Example	4	-	Histories of reentry velocity and related factors
Example	5	-	Body loads as affected by a combined asymmetry (center of gravity offset and a distorted body)
Example	6	-	Roll lock-in as affected by the com- bined asymmetry of Example 5
Example	7	-	Roll lock-in as affected by a com- bined asymmetry - center of gravity

offset and nonzero product of inertia

يور الحالة والاليونو (California) - يوا ويكرف المراجع التي المالية

Preceding page blank

- 237 -

4.6.1 Example 1

Compute the characteristics of a minimum energy trajectory (MET) for a nonrotating earth for a launch from $\phi_{L} = 0^{\circ}$, $\Lambda_{L} = 0^{\circ}$ to a target at $\phi_{T} = 45^{\circ}$, $\Lambda_{L} = 45^{\circ}$.

The range is (Eqs. (4, 2-17) and (4, 2-18)) R = 3610 nmi,

The range angle is (Eq. (4.2-18)) $2\theta_0 = 2\theta_0 = 60^\circ$.

The azimuth angle at launch is (Eq. (4, 2-20)) $\psi_{LT} = 35, 4^{\circ}$,

For an MET, the initial flight path angle is (Eq. (4.1-10))

$$\gamma_{-} = 45^{\circ} - 15^{\circ} = 30^{\circ}$$
.

From Eq. (4.1-11) $\overline{V}_0 = 0.815$ and, assuming the initial altitude to be sea level, the initial velocity is

 $V_{2} = 0.815 (26 000) = 21 200 \text{ ft/sec}.$

The peak altitude of the trajectory is (Eq. (4.1-22)) h_a = 635 nmi.

The velocity at peak altitude is (Eq. (4.1-24)) V_a = 15 500 ft/sec, or about 73% of the initial velocity.

The time to impact is (Eq. (4.1-17)) t = 1480 seconds.

4.6.2 Example 2

For a rotating earth, determine the characteristics of a trajectory fired with initial conditions V_0 , γ_0 , and $\psi_{1,1}$ (relative to earth) computed in Example 1.

The inertial quantities V, γ , and ψ at launch are (Eqs. (4.2-7), (4.2-8), and (4.2-10)):

Preceding page blank

- 239 -

THE JOHNS HOPKINS UNIVERSITY APPLIED PHYSICS LABORATORY BILVER SPRING MARYLAND

$$\frac{\tan \psi_{\rm L}}{\tan \psi_{\rm e}} = 1 + \frac{1520(1)}{21\ 200\ \cos 30\ \sin 35.4} = 1.142$$

$$\tan \psi_{\rm L} = 1.142\ \tan 35.4$$

$$\psi_{\rm L} = 39.1^{\circ}$$

$$\frac{V_{\rm L}}{V_{\rm e}} = 1 + \cos 30\ \sin 35.4\left[\frac{1520(1)}{21\ 000}\right] = 1.036$$

$$V_{\rm L} = 1.036\ (21\ 200) = 22\ 000\ ft/sec$$

$$\overline{V}_{\rm L} = \frac{22\ 000}{26\ 000} = 0.845$$

$$\frac{\sin \gamma_{\rm L}}{\sin \gamma_{\rm e}} = \frac{21\ 200}{22\ 000} = 0.963$$

$$\sin \gamma_{\rm L} = 0.963\ (0.50) = 0.482$$

$$\gamma_{\rm L} = 28.9^{\circ}$$

Therefore the effect of earth rotation is to increase the initial inertial velocity by about 3.6%, to lower the flight path angle by 1.1°, and to increase the azimuth angle by 3.7°. These changes would have opposite signs if the vehicle had been fired in a westerly direction.

Using \overline{V}_L and γ_L for \overline{V}_0 and γ_0 , the equations given in Section 4.1 apply. However, note that the trajectory is not an MET since Eq. (4.1-11) is not satisfied. For the given \overline{V}_L a minimum energy trajectory would have $\gamma_L = 28.1$ so that the trajectory being considered is slightly high compared to an MET.

- 240 -

THE JOHNS HOPKINS UNIVERSITY APPLIED PHYSICS LABORATORY SILVER SPRING MARTLAND

The nonrotating earth range angle is (Eq. (4, 1-7)):

 $2\theta_0 = 67.4^\circ$.

The corresponding range, $r_0(2\theta_0) = 4020$ nmi, is the arc length traveled in inertial space (or the range on a non-rotating spherical earth for initial conditions V_{L} , γ_{L} , and ψ_{L}).

The impact time is (Eqs. (4, 1-5) and (4, 1-16)) t = 1660 seconds.

The impact latitude is unaffected by earth rotation and is $\theta_T = 45^\circ$.

The impact longitude is (Eq. (4.2-16)) $\Lambda_I = 49.4^\circ$. The impact range is (Eqs.(4.2-17) and (4.2-18)) R = 3780 nmi.

A summary of the results is given below:

Quantity	Non. otating Earth	Rotating <u>Earth</u>			
Flight time (seconds)	1480	1660			
Impact latitude (degrees)	45.0	45.0			
Impact longitude (degrees)	45.0	49,4			
Impact range (nmi)	3610	3780			

For the initial conditions used, the effect of earth rotation is to increase the range by 170 nmi and to increase the flight time by 180 seconds.

4.6.3 Example 3

The typical reentry body, having mass characteristics defined in Table 2-1, is separated from its booster in the exoatmosphere. At separation, the body is spun about its longitudinal axis at a rate of 360°/sec and,

- 241 -

We the Merseller of the

unintentionally, is given a transverse rate of 60°/sec. Determine the characteristics of the body motion during its travel in the exoatmosphere.

The only forces acting on the missile are weight and the forces resulting from the body rates. Under the influence of these forces, the body is gyroscopically stabilized, and the transverse rate imparted at separation results in the conical motion described in Section 4.4.1. The period of the motion is (Eq. (4, 4-25)):

 $\tau_s = 5.12$ seconds.

The signal strength of a tracking radar varies with the attitude of the reentry vehicle. Therefore, the radar signal strength would oscillate with a period $\tau_{g} = 5.12$ seconds.

The semicone angle of the space cone is (Eq. $(4, 4-23)) \emptyset = 50^{\circ}$. The body rates measured by on-board rate gyros have the following characteristics (assuming a symmetrical vehicle):

1. The roll rate is constant and equal to 360°/sec.

2. The pitch and yaw rates very sinusoidally about zero with peak values of $\pm 60^{\circ}$ /sec. The pitch and yaw rates are 90° out of phase. The period shown on the traces is (Eq. (4, 4-24)):

 $\tau_{\rm m}$ = 1.11 seconds,

If the body were spin stabilized so that the total angular momentum vector II were along the flight path_ direction at reentry, during the early phase of reentry α would be constant and equal to 59°. However, the pitch and yaw (inertial) components of α would oscillate with a half-amplitude of 59° and a period of 5.12 seconds.

- 242 -

4.6.4 Example 4

Determine the effect of the atmosphere on the motion of the reentry body using the following reentry conditions:

$$h_{E} = 400\ 000\ \text{feet}$$

$$V_{E} = 20\ 000\ \text{ft/see}$$

$$\gamma_{E} = -30^{\circ}$$

$$p_{E} = 0^{\circ}/\text{sec and } 360^{\circ}/\text{sec} \quad (6, 28\ \text{rad/sec})$$

$$\overline{\alpha}_{E} = 10^{\circ}$$

$$\omega_{m} = 0$$

Use the 1962 Standard Atmosphere given in Table 3-1 and the mass and aerodynamic characteristics given in Table 2-1.

Since $\omega_{\rm TP} = 0$, the vehicle is not in a conical motion as was described in the previous problem. The angle of attack α remains constant and in the original plane (inertial pitch and yaw angles of attack remain constant) until aerodynamic forces begin to affect the motion by causing a decay in α . Using $\alpha/\alpha_{\rm E} = 0.99$ as the beginning of the α decay, for $p_{\rm E} = 360^{\circ}/{\rm sec}$:

$$\frac{\rho}{\rho_0} (Eq. (4.4-55)) = 3.21 \times 10^{-7}.$$

From Table 3-1, this density ratio corresponds to an altitude between 300 000 and 350 000 feet. A more complete table shows the altitude to be h = 332 000 feet. If the initial roll rate were zero, Eq. (4.4-41) could be used to obtain the altitude at which the $\overline{\alpha}$ decay begins

$$\frac{P}{P_{p}} H = \frac{0.02 (300) (32, 2) (0, 25)}{9.65 (2) (0, 4) (2116)} = 2.96 \times 10^{-3} \text{ feet,}$$

- 243 -

Using the characteristics of the 1962 Standard Atmosphere, this value of $\frac{P}{P_0}$ II corresponds to an altitude of about 350 000 feet. Therefore, the effect of the 360°/sec spin rate was to delay the $\overline{\alpha}$ decay by about 18 000 feet.

The $\overline{\alpha}$ decay histories may be computed using Eqs. (4, 4-44) and (4, 4-53). In order to evaluate these equations, the variations of ω_A and K with altitude are required. These characteristics, obtained using Eqs. (4, 3-6) and (4, 3-23) and the atmosphere defined in Table 3-1, are listed in Table 4-1. From Eq. (4, 4-45) and Table 2-1, the damping constant is:

$C_3 = 7.6$

About 60% of C $_3$ comes from the $\mathrm{C}_{N_{ov}}$ term and 40% from the $\mathbf{C}_{\mathbf{m}_{\mathbf{n}}}$ term. The $\overline{\alpha}$ decay history for the rolling body was computed using Eq. (4, 4-53), and the results are shown in Fig. 4-20. Also shown is the case of the nonrolling missile and the case of a rolling missile with $C_3 = 0$. For the rolling vehicle, the effect of the C_3 damping term is negligible for altitudes above about 150 000 feet. For altitudes below about 60 000 feet, the α history for the rolling case (with damping) is nearly the same as that for the nonrolling case. However, at altitudes above 60 000 feet, the angle of attack at a given altitude for the rolling body is significantly higher than that for the nonrolling body. This fact could be important for some missiles in which the peak lateral load occurs at these altitudes. For the body studied, the peak interal load resulting from $\overline{\alpha}_{\rm E}$ occurs at an altitude (Eq. (4, 4-59)) for which:

$$K = \frac{3}{4} \frac{1}{(7,6+0,75)} = 0.090$$

From Table 4-1, this value of K corresponds to an altitude of about 85 000 feet. The altitude for maximum g_L is the same whether or not the missile is rolling. However, as a result of the $\overline{\alpha}$ histories, g_{Lmax} for the

- 244 -

rolling missile is about 45% greater than that for a nonrolling vehicle (Eqs. (4, 4-60) and (4, 4-63)). For $\alpha_{\rm E} = 10^{\circ}$, $g_{\rm Lmax}$ for a rolling vehicle is 2,75g compared to 1,9g for a nonrolling vehicle.

The reentry velocity, Mach number, and Reynolds number are important environmental factors that affect body loads and temperature. The time history of these characteristics estimated using Eqs. (4, 3-5), (4, 3-7), and (4, 3-10) and the atmospheric characteristics given in Table 3-1 are listed in Table 4-1.

The Mach number history shows that the vehicle is in the hypersonic regime ($M \ge 6$) down to an altitude of about 15 000 feet and is supersonic from 15 000 feet to impact. The Reynolds number history shows that transition on the body would be expected at an altitude of about 110 000 feet ($R_{e} \approx 10^{-7}$) with laminar flow at higher altitudes and turbulent flow at lower altitudes. The heating rate to the conical surface of the body would increase by a factor of 2 to 3 when the flow becomes turbulent, and therefore asymmetries that result from heating would be expected to increase rapidly at the transition altitude. The aerodynamic frequency data show that roll resonance

 $(\lambda = p \sqrt{1 - \frac{l_X}{l_Y}} / \omega_A = 1)$ occurs at an altitude of about

135 000 feet. It should be noted that the vehicle is approaching a second resonance condition at impact, and any asymmetries that exist near altitudes of 135 000 feet and sea level are likely to be amplified. The computed values of M/R_c show that free molecular flow regime begins at altitudes well above 400 000 feet ($M/R_e \approx 10$), transition flow exists down to an altitude of about 250 000 feet ($M/\sqrt{R_e} \approx 0.1$), slip flow exists at altitudes from 250 000 to 140 000 feet ($M/\sqrt{R_e} \approx 0.01$), and continuum flow exists at altitudes from 140 000 feet to sea level. The time from 400 000 feet to impact is about 49 seconds compared with about 39 seconds if the atmosphere were absent. Therefore, the atmosphere increased the flight time by about 10 seconds.

- 245 -

Although not shown in the table, several other important considerations may be obtained from the equations given in Section 4.3. Using the ratio $q_g/q_{B_{\rm MBX}} \approx$ 0,1 as an indication of conditions where heating is significant, the altitudes are computed (Eqs. (4, 3-11) and (4, 3-16)) to be about 190 000 feet for the beginning of the heating period and about 13 000 feet for the end of heating with the maximum heating rate at an altitude of about 58 000 feet. The longitudinal deceleration(Eqs. (4, 3-9) and (4, 3-14)) is maximum at an altitude of about 35 000 feet where $g_{\rm Xmax} \approx -54g$. The inflection points in the $g_{\rm X}$ versus time curve (Eq. (4, 3-19)) occur at altitudes of 58 000 and 19 000 feet.

The results obtained in this problem are summarized in Fig. 4-21. Although the altitude bands shown vary slightly depending upon vehicle geometry and trajectory characteristics, the results are typical of high β reentry bodies. Note that practically all of the quantities of interest occur at altitudes below about 150 000 feet and therefore are in the continuum flew regime. Both laminar and turbulent flow conditions exist in this altitude region, all three Mach number regimes occur (although only two occurred for this trajectory), and both oscillatory and trim types of body motion may be important.

4, 6, 5 Example 5

Suppose the reentry vehicle has a center of gravity offset from the body centerline of 0,042 inch. Furthermore, assume that as a result of exoatmospheric blast effects the body is distorted so that the body centerline is an arc of a circle and that the tangent to the centerline changes by 0,25° from base to nose tip. Use the same initial conditions as were used for Example 4 except use $p_{\rm E} = 1000^{\circ}/{\rm sec}$. Assess the probability that the missile will encounter excessive lateral loads. For this problem we will define $g_{\rm L} \ge 20g$ as excessive.

THE JOHNS HOPKING UNIVERSITY APPLIED PHYSICS LABORATORY BILVER SPRING MARYLAND

> From Eq. (2-35) the \overline{C}_{m_0} resulting from the bent body is 0.002. We will assume that the lateral location

of the center of pressure and center of gravity are unaffected by the bent body.

From Eq. (4.5-65) the trim angle of attack resulting from $\overline{C}_{\rm m_{\odot}}$ is:

$$\overline{\alpha}_{0_{M}} = \frac{0.002}{2 \times 0.114} \times 57.3 = 0.5^{\circ}.$$

The trim angle resulting from the center of gravity offset is (Eq. (4, 5-64)):

$$\overline{\alpha}_{0, 2} = \frac{0.104 \frac{0.042}{42}}{2 \times 0.114} \times 57.3 = 0.0261^{\circ}.$$

Therefore, the total trim angle $\overline{\alpha}_0$ is very nearly the angle resulting from the bent body and we will use $\overline{\alpha}_0 = 0, 5^{\circ}$. The equation for $g_{1,}$ is:

$$g_{L} = \frac{C_N S}{W} \overline{q} A \overline{\alpha}_0$$

For the moment assume that no amplification in $\tilde{\alpha}_0$ occurs, so that A = 1. For this case, the largest possible value of g_L occurs at \bar{q}_{max} . From Eq. (4.3-13) \bar{q}_{max} occurs at $P/P_0 = \frac{1000}{2116} \left(\frac{1}{2}\right) = 0.237$, corresponding to an altitude of 35 000 feet and a temperature of 394°R. The magnitude of \bar{q}_{max} is:

$$\overline{q}_{max} = \frac{1000 (20 000)^2 (0.5)}{9330 (394)} = 54 400 \text{ W/ft}^2$$

and

- 247 -

ar د به به به به در به مه در به مما د بر به مما

Excessive values of g_L will not be obtained unless values of A in excess of 2 are obtained. Amplifications of this magnitude will be encountered if large excursions in roll rate occur. The possibilities that must be considered are:

- 1. Roll lock-in at first resonance,
- 2. Roll lock-in at second resonance,
- 3. Spin up with lock-in at some intermediate altitude,
- 4. Spin down with lock-in at a negative roli rate.

Assuming that lock-in is attained, from Eq. (4.5-16) A = 1/D and D may be evaluated using Eq. (4.5-60). Substituting these equations and the equation for q (Eq. (4.3-8)) into the equation for g_{L} , excessive loads will occur when the following inequality is satisfied.

$$\overline{\alpha}_{0} \geq \frac{57.3 \times 20 (k_{1} - C_{m_{q}}) \text{gd}}{C_{N_{\alpha}} V_{E}} \left[\frac{-2C_{A} \frac{H}{d}}{\frac{\Delta X}{d} k_{1} \sin \gamma} \right]^{1/2} \frac{e^{k_{1}}}{V_{K}}$$

where $\overline{\alpha}_{0}$ is expressed in degrees.

Substituting for the mass, aerodynamic, and trajectory parameters, and assuming W = constant = 22,000feet:

$$\overline{\alpha}_{0} \geq 0,04 \Im \frac{\sqrt{K}}{\sqrt{K}}$$

This inequality is plotted in Fig. 4-22. It is observed that $\overline{\alpha}_0 = 0, 5^\circ$ is large enough to result in excessive loads at some altitudes. The first and second resonance eltimates are the altitudes for which $\omega_A = p_{\rm PE} = p_{\rm E} \sqrt{1 - \frac{\Gamma_2}{\Gamma_2}} = 0$ /sec.

· 248 -

Values of ω_A were computed using Eq. (4, 3-23) and the results are shown in Fig. 4-23. First resonance occurs at an altitude of about 89 000 feet, and second resonance occurs at an altitude of about 4000 feet. Therefore roll lock-in at first resonance would result in excessive loads, but lock-in at second resonance would not.

Depending upon the value of ϕ_{α} , the roll rate may not lock-in at first resonance but may either spin up or spin down. The magnitude of the decrease in roll rate is given by Eq. (4.5-93). This equation may also be used to obtain an approximate value of the spin up characteristics. However, the magnitude of Δp in this case will be underestimated slightly since the amplification in $\alpha_{\rm o}$ and the change in the angle \emptyset are neglected. Using Eq. (4.5-93) and assuming values of $\beta_c = 90^\circ$ and 270° (which give the largest values of Δp) values of p_r were computed and are shown in Fig. 4-23. From this consideration it is found that the asymmetries are not large enough to result in roll reversal. Therefore, resonance at negative roll rates may be eliminated as a possible condition that would result in excessive loads. However, if the vehicle does not lock in at first resonance but spins up, resonance may occur at an altitude of about 13 000 feet. From Fig. 4-22, resonance at this altitude would result in excessive loads. Therefore, the problem is reduced to an evaluation of whether or not lock-in will occur at altitudes of 89 000 and 13 000 feet.

The criterion for lock-in is (Eq. (4.5-61)):

$$\overline{\alpha}_{o} \frac{r}{d} \ge F_{1} \frac{(1-K_{H})}{k}$$

Using the data given in Tables 2-1 and 3-1 and the trajectory characteristics, the value of F_1 (Eq. (4.5-62)) is 5.4×10^{-6} for h = 89 000 feet (H = 21 700 feet) and is 4.7×10^{-6} for h = 13 000 feet (H = 25 000 feet). Using r/d = 0.042/12(3.5) = 0.001 and expressing $\overline{\alpha}_0$ in degrees:

- 249 -

THE JOHNS MOPKINS UNIVERSITY APPLIED PHYSICS LABORATORY SILVER SPRING MARYLAND

$$\overline{n}_{0} \approx 0.31 \frac{(1 - K_{R})}{k}$$
 for h = 89 000 feet
> 0.27 $\frac{(1 - K_{R})}{k}$ for h = 13 000 feet.

Using the data of Table 4-1 a plot of K versus altitude shows that $K_R = 0.077$ at 89 000 feet and $K_R = 2.58$ at 13 000 feet. Therefore, the criteria for lock-in are:

$$\overline{\alpha}_{0} \ge \frac{0.286}{k}$$
 for h = 89 000 feet
 $\ge \frac{-0.427}{k}$ for h = 13 000 feet.

The value of k depends upon the value of ϕ_0 and, for lock-in to occur, k must be positive for $K_R < 1$ and negative for $K_R > 1$. For this problem we must assume that ϕ_0 may have any value. Therefore, we must consider the following possibilities:

Resonant Altitude	<u>ø</u> o	k	<u>a_≥</u>
89 000 feet	. 0°	1	0,286
	9 0°	0.5	0,572
	2 70 °	0,5	0,572
13 000 feet	270°	-0.5	0 854

The other two values of \emptyset_0 having negative values for k for the 13 000 foot case are ruled out since the $\emptyset_0 = 270^\circ$ value is required to provide spin up during reentry (see Eq. (4.5-33)). Of the four possibilities only the $\emptyset_0 = 0^\circ$ case with resonance at 89 000 feet satisfies the α_0 criterion.

In the preceding discussion we considered only inplane and out-of-plane conditions. For intermediate values

- 250 -

t tenaka dalah dapataka alipopol Mala Aliberta dalamata of ϕ_0 , $\overline{\alpha}_0$ may be divided into in-plane and out-of-plane components. Any value of ϕ_0 for which the component of $\overline{\alpha}_0$ in the $\phi_0 = 0^\circ$ direction is ≥ 0.286 will result in excessive values of g_L . The possibilities are shown in Fig. 4-24. Any value of ϕ_0 defined by points from A to B (shorter arc length) will result in excessive values of g_L . Since any value of ϕ_0 is equally likely to occur, the probability of encountering excessive g_L is:

$$\overline{\mathbf{P}} = \frac{2 \cos^{-1} \frac{0.286}{0.500}}{360} = 0.31 .$$

4,6,6 Example 6

For the trajectory used in Example 4, estimate as a function of p_E the maximum allowable α_0 combined with r = 0.042 inch so that neither lock-in at first resonance nor spin through zero occurs. In view of the discussion given in Section 4.5.10, assume that the criterion for lock-in is the trim equation (Eq. (4.5-61)) or G = 20, whichever gives the larger value of α_0 .

The critical conditions to be investigated are the $\phi_0 = 0^\circ$ case for lock-in and the $\phi_0 = 90^\circ$ case for spin through zero.

For the lock-in case, assume that H = constant = 21 700 feet. Substituting the mass, aerodynamic, and trajectory terms into Eq. (4.5-58), the criterion for lock-in becomes:

$$\overline{\alpha}_{0} \geq \frac{2.65}{G} \frac{(1 - K_{R})}{\sqrt{K_{R}}} .$$

If trim conditions are attained, the criterion for lock-in is (see Example 5):

$$\overline{\alpha}_{0} \geq 0, 31 (1 - K_{R}),$$

- 251 -

THE JOHNS HOPKINS UNIVERSITY APPLIED PH15'CS LABORATORY BILVER BRAINS MARYLAND

Setting G = 20 in the first equation, the two equations give the same solution to $\overline{\alpha}_0$ at:

$$\frac{2.65}{20} \frac{1}{\sqrt{K_R}} = 0.31$$

or

alah **kanalahi**n aktion na shira samin sa sa ji sa sa sa sa sa sa sa

$$K_{R} = 0.183$$
,

corresponding to an altitude of about 70 000 feet (see Table 4-1). At altitudes lower than 70 000 feet (or $K_R > 0.183$) the trim equation is used; at altitudes higher than 70 000 feet the nontrim equation is used with G = 20.

For each resonant altitude the corresponding initial roll rate may be computed from the equation:

$$P_{E} = \frac{\omega_{A}}{\sqrt{1 - \frac{I_{X}}{I_{Y}}}}$$

where ω_A is given in Table 4-1. The values of $\overline{\sigma}_0$ required for lock-in are shown as a function of p_E in Fig. 4-25.

The change in roll rate resulting from asymmetries α_0 and r/d is given by Eq. (4, 5-93). Spin through zero roll rate will occur during reentry if Δp is negative and $|\Delta p| \ge p_E$. Using the most restrictive condition $(\phi_0 = 90^\circ \text{ and } K = K_{SL})$ and expressing α_0 and p_E in units, of degrees and deg/sec, respectively, the criterion for spin through zero roll rate is:

$$\overline{\alpha}_{0} \ge 8,17 \times 10^{-4} p_{E}$$
.

The values of $\overline{\alpha}_0$ required for spin through zero roll rate are shown in Fig. 4-25.

- 252 -

Second We Prairie Steer & W

, 🗰 ...

THE JOHNS HOPKING UNIVERSITY APPLIED PHYSICS LABORATORY SILVER SPRING MARYLAND

> This problem illustrates some of the trade-offs which must be considered by the designer in selecting a roll rate for the reentry body. If the roll rate is very low, first resonance occurs at a very high altitude, where large values of α_0 may be tolerated without encountering lock-in. However, vehicles with low values of p_E are susceptible to gpin through zero roll rate. On the other hand, for moderately high values of p_E , large values of a may be tolerated without spinning through zero roll rate, but $\overline{\alpha}_{0}$ must be very low if lock-in is to be avoided. For this particular problem the best roll rate is $p_{\rm H}$ = 760°/sec where the missile could tolerate up to 0, 6° in $\overline{\alpha}_{0}$ without encountering either adverse condition. A second possibility would be to select a value of p_E much greater than 2240°/sec. In this case spin through zero roll rate would be unlikely, and lock-in is avoided by using a roll rate that is much greater than the maximum aerodynamic frequency. This technique is sometimes referred to as "overspin."

4.6.7 Example 7

Using the trajectory of Example 4 with $p_E = 1000^{\circ}/\text{sec}$, first resonance occurs at an altitude of 89 000 feet (K_R = 0.077). Determine the values of $\overline{\alpha}_{\delta}$ and r for which lock-in at first resonance may occur. Assume that the trim equation (Eq. (4.5-125)) is valid. Compare the results with $\overline{\alpha}_{0}$ and r given by Eq. (4.5-61).

From Eq. (4.5-62) $F_1 = 5.4 \times 10^{-6}$ (H = 21 700 feet). Using the mass and aerodynamic characteristics given in Table 2-1, $F_3 = 4.87$ and $F_4 = 5.26 \times 10^{-5}$. For α_{δ} in degree units, Eq. (4.5-125) becomes:

$$\bar{\alpha}_{\delta} = 279 \frac{r}{d} \left[1 \pm \sqrt{1 - \frac{2.05 \times 10^{-6}}{(r/d)^2}} \right]$$

The α_0 boundary is shown in Fig. 4-26. At a given value of r, α_0 must be greater than α_0 min but less than α_0 max. Lock-in cannot occur for any value of α_0 if r < 0.06 inch.

- 253 -

By comparison the trim equation for $\overline{\alpha}_0$ and r becomes:

$$\overline{\alpha}_{0} = \frac{0.286 \times 10^{-3}}{r/d}$$

The $\overline{\alpha}_{0}$ boundary is also shown. It is noted that $\overline{\alpha}_{0}$ has no $\overline{\alpha}_{0}_{\max}$ boundary, and the $\overline{\alpha}_{0}$ boundary is less than $\overline{\alpha}_{0}_{\min}$.

Single points on each boundary obtained from sixdegree-of-freedom simulation results (see Section 4.5.10) are also shown. The $(\overline{\alpha_0}, r)$ point was obtained by holding r constant and increasing $\overline{\alpha_0}$ until lock-in occurred. The $(\overline{\alpha_0}, r)$ point was obtained by holding $\overline{\alpha_0}$ constant and varying r until lock-in occurred. Note that the real boundaries are somewhat higher than the boundaries based on the trim equations as discussed in Section 4.5.10,

For other values of p_E , the $\overline{\alpha}_{\delta}$ boundary must remain within the $\overline{\alpha}_{\delta \max}$ and $\overline{\alpha}_{\delta \min}$ asymptotes shown in Fig. 4-26, and r_{\min} is shifted right or left depending upon the value of K_R . Using data given in Table 4-1 and the relationship between p_E and ω_A at resonance,

$$p_E = \frac{\omega_A}{\sqrt{1 - \frac{I_X}{I_Y}}}$$
, r_{min} is shown in Fig. 4-27 as a function

of p_E . For r = 0.042 inch used in the previous problem, lock-in could exist only for values of p_E that result in resonance very near to the altitude of maximum dynamic pressure (35 000 feet).

For current reentry vehicles, values of ρ_E are selected such that first resonance occurs at altitudes above 100 000 feet. In this altitude range r_{\min} is insensitive to p_E , and the $\overline{\alpha_0}$ boundary shown in Fig. 4-26 is applicable to first resonance lock-in for all values of p_E

- 254 -

THE JOHNS HOPKING UNIVERSITY APPLIED PHYSICS LABORATORY SILVER BRING MARTLAND

> less than about 1000°/sec. For $p_E = 1000°/sec$ considered in this problem, second resonance occurs at an altitude of about 4000 feet (neglecting a change in roll rate during reentry) where r_{min} is nearly a factor of 2 greater than r_{min} at first resonance.

 A 2 3 1

anna an Sannaise Rait ne bh<mark>annaiseach s</mark>annaiseach Is an Sannaiseach an Sannaiseach san an Sannaiseach THE JOHNS HOPKINS UNIVERSITY APPLIED PHYSICS LABORATORY BILVER SPRING MARYLAND

REFERENCES

- 4-1 L. S. Glover, "Approximate Re-Entry Velocity and Heating Equations," <u>J. Spacecraft and Rockets</u>, January 1966.
- 4-2 D. T. Greenwood, "Flight Mechanics of Space and Re-Entry Vehicles," Michigan University Engineering Summer Conferences, 1964.
- 4-3 H. J. Allen, Motion of a Ballistic Missile Angularly Misaligned with the Flight Path upon Entering the Atmosphere and Its Effect upon Aerodynamic Heating, Aerodynamic Loads, and Miss Distance, NACA TN-4048, 1965.
- 4-4 R. B. Powell and R. L. Smith, "Dynamics of Spinning Re-Entry Bodies," <u>AIAA Preprint No. 64-</u> 470, 1964.
- 4-5 H. I. Leon, "Angle of Attack Convergence of a Spinning Missile Descending Through the Atmosphere," J. Aerospace Sci., August 1958.
- 4-6 L. S. Glover, Analytical Expressions for the Effect on Roll Rate of Mass and Aerodynamic Asymmetries for Ballistic-Type Bodies, APL/ JHU TG 560, March 1964.
- 4-7 J. D. Nicolaides, On the Free Flight Motion of Missiles Having Slight Configurational Asymmetries, Ballistic Research Laboratories, Aberdeen Proving Ground, Report 858, June 1953.
- 4-8 R. L. Nelson, <u>Measurement of Aerodynamic Characteristics of Re-Entry Configurations in Free</u> Flight at Hypersonic and Near Orbital Speeds, Lockheed Missiles and Space Division, TR 6-90-61-37, July 1961.

Preceding page blank

- 257 -

THE JOHNS HOPKINS UNIVERSITY APPLIED PHYSICS LABORATORY SILVER SPRING MARYLAND

s fi

REFERENCES (cont'd)

- 4-9 A. E. Hodapp, Jr. and E. L. Clark, Jr., "The Effects of Products of Inertia on the Rull Behavior of Ballistic Re-Entry Vehicles," <u>AIAA Paper No.</u> 70-204, January 1970.
- 4-10 L. S. Glover, "Effects on Roll Rate of Mass and Aerodynamic Asymmetries for Ballistic Re-Entry Bodies," J. Spacecraft and Rockets, April 1965.
- 4-11 K. A. Ehricke, Principles of Guided Missile Design, Space Flight, Vol. II, "Dynamics," D. Van Nostrand Company, Inc., 1962.

Ì

All any statements an

THE JOHNS HOPKINS UNIVERSITY APPLIED PHYSICS LABORATORY Silver Spring, Maryland

1

ł

Ē

•••





- 259 -

.

۲.)

at 1.

 $\mathcal{L}^{(n)}$

4

 THE JOHNS HOPKING UNIVERSITY APPLIED PHYSICS LABORATORY SILVER BPRING MARYLAND



大阪市内国

現山田を

n in the second se

「日本の一般の日本のない」のない

Fig. 4-21 DIGNIFICANT EVENTS



THE JOHNS HOPKINS UNIVERSITY APPLIED PHYSICS LABORATORY SILVER SPRINS MARYLAND

3



Barran and

'n

.

.

Lingun (Marine et a)

- 261 -

đ.

S. 1. 1.

THE JOHNS HOPKING UNIVERSITY APPLIED PHYSICS LABORATORY BILVER BRUND MARTLAND

STORY -

ł



- 262 -

and the second sec

.

THE JOHNE HOPKINE UNIVERSITY AFFLIED FHYSICS LABORATORY BILVER SPRINC MAI YLAND

ł

1

1

i

.

ì

. . . .



Fig. 4-24 CONDITIONS FOR ROLL RESONANCE AT h = 89 000 FEET

- 263 -

.. ..

(a) Some Antonia and a standard between an Antonia (Appl. 2019). (a) Some Antonia and Appl. (a) Some Antonia and Appl. (b) Some Antonia and Appl. (c) Some Antonia a THE JOHNS HOPKINS UNIVERSITY APPLIED PHYSICS LABORATORY Silver Bring Maryland







1

.

Charles Mary mary tory

THE JOHNS MOPKINS UNIVERSITY APPLIED PHYSICS LABORATORY SILVER SPRING MARVLAND



Fig. 4.26 BOUNDARIES OF \vec{a}_{o} and $\vec{a}_{\vec{b}}$ for lock-in at first resonance

- 265 -

THE JOHNS HOPKINS UNIVERSITY APPLIED PHYSICS LABORATORY BILVER SPRING MANYLAND

2



Fig. 4-27 VARIATION OF rmin WITH INITIAL ROLL RATE

122.5

.

. . .

Table 4-1

ļ

Reentry Characteristics

t (sec)	0					25.0	29.9	31.0	32.1	33. 2	34. 6	35.8	37.3	1	39.2	5.It	11.3	10 10
M/NR _e	3.9	+	0. 38	130 0	0. 034	0.013	0. 0044											
R _{e × 10} -6	0. 0000156	0. 000195	0. 0036	0, 0525	0.315	2.0	22	34	0 .0	87	:30	200	5 70	1	316	<u></u>	240	160
W	15.6	19.7	22 6	22.0	19.4	13.9	20.3	19.9	16.7	19.2	18.0	16.6	14,5	1	10.9	01 +1	4	ะง ก่
V (ft/sec)	20 000	20 000	20 000	20 000	20 000	20 000	20 000	19 500	19 200	18 600	17 500	16 CO0	13 900	1	008 C I	627 T	+ 760	. 2 470
نعd/sec) (£ad/sec)			0.135	0. 597	1.65	4.23	12.9	16.0	19.8	24.2	29.4	33. 8	37.0	37.2	36.1	30.8	22.1	13.4
×					0. 00083	0. 0057	0.047	0.074	0.116	9.182	0.301	0.486	0. 785	5.00	1.26	1.94	2.90	4. 23
Altitude (tiousands of foet)	400	350	300	250	200	150	100	06	80	70	60	in. M	10	10 C3	30	20	10	s. L.

THE JOHNS HOPKINS UNIVERSITY APPLIED PHYSICS LABORATORY BILVER SPHING MARYLAND

ينجل من

ì

Sec. Sec. Sec.

THE JOHNS HOPKING UNIVERSITY APPLIED PHYSICS LABORATORY SILVER SPRING, MARYLAND

L

L

I

I

Ţ

ар на 11 11 11 м. на

1

I

I

I

I

5. DISPERSION

.

「「「「「「「」」」」」」

2.42.25.20

Ple.

1.45

49

÷.,
THE JOHNS HOPKINS UNIVERSITY APPLIED PHYSICS LABORATORY SILVEN SPRING MARYLAND

I

SYMBOLS FOR SECTION 5

Symbol	Definition	Typical Units
Α	axial force	pounds
AR	component of A in downrange direction	pounds
CA	axial force coefficient = A/qS	-
c _D	drag coefficient = D/qS	-
c.g.	center-of-gravity location	inches
C _L	lift coefficient = L/qS	
	dC _L /dax	per radian
C _{lo}	asymmetric rolling moment coefficient l _o /qSd	-
с _{то}	moment coefficient about body Y axis at $\overline{\alpha} = 0$ (see Section 2.4)	-
¯ _{m₀}	$\sqrt{\frac{C_{m_0}^2 + C_{n_0}^2}{C_{m_0}}}$	-
C _N	normal force coefficient = N/qS	-
C _N	dC _N /da	pe r r adian
с _{ло}	moment coefficient about body Z axis at $\overline{\alpha} = 0$ (see Section 2.4)	-
c.p.	center-of-pressure location	inches
D	drag, aerodynamic force parallel to V	pounds
d	vehicle reference length	feet
Е	factor defined by Eq. (5-21)	ſeet
е	Napierian base	-
$\mathbf{F}_{\mathbf{w}}$	weighting factor for wind	-
^F р	Weighting factor for density variation	-

Preceding page blank

- 271 -

Symbol	Definition	Typical Units
g	acceleration of gravity	ft/sec ²
g	$\left[g_{Y}^{2}+g_{Z}^{2}\right]^{1/2}$	g
^g x' ^g y' ^g z	acceleration along body axes X , Y , and Z	g
Н	scale height RT/g	fect
h	altitude	feet
1 ₃ , 1 ₅	functions of K _{S.L.}	-
I _x	moment of inertia about body axis ${f X}$	$slug-ft^2$
j	index number	-
к	$-\frac{P}{\beta \sin \gamma}_{E}$	-
L	lift, aerodynamic force normal to V	pounds
l	asymmetric rolling moment	ft-lb
N	normal force, component of aero- dynamic force normal to the body centerline	pounds
	or	
	number of altitude bands	-
Р	ambient pressure	lb/ft ⁻
р	vehicle roll rate	rad/sec
q	dynamic pressure, $\rho V^2/2$	lb/ft ⁻
R	range	nmi
	or	ŋ ')
	gas constant = 1716	it"/ sec " - °R
Ro	radius of the earth	feet

Contraction of the local distribution of the

1 1 1

.

A.S.

A

and the star inter-

- 272 -

1.1.4.15

THE JOHNS HOPKINS UNIVERSITY APPLIED PHYSICS LABORATORY SILVER SPRING MARTLAND

المتعادية

١.

. .

đ

Ĵ,

Symbol	Definition	Typical Units
r	radial c.g. offset	inches
S	reference area	ft ²
S. L.	sea level	-
Т	ambient temperature	۰R
t	time	seconds
v	vehicle velocity	ft/sec
v _R	drift velocity resulting from winds	ft/sec
v _w	windspeed	ft/sec
V _{A/M}	velocity of the air relative to the missile	ft/sec
W	vehicle weight	pounds
X _{c.g.}	distance parallel to the body center- line from the nose to the $c \cdot g$,	inches
Х _{с. р.}	distance parallel to the body center- line from the nose to the c.p.	inches
ΔX	X _{c.p.} - X _{c.g.}	inches
ā	angle between the body centerline and $\vec{\nabla}_{A/M}$	radians
α _w	angle between the body centerline and the reentry flight path direction	radians
β	ballistic coefficient, W/C_AS	lb/ft ²
γ	flight path angle, negative for a reentry body	degrees
$\Delta \gamma$	change in flight path angle	degrees
ð	dispersion measured along the earths surface	feet
Δ	refers to an increment except as noted	-
ø	angle defined in Section 5, 4, 2	degrees

- 273 -

THE JOHNE HOPKING UNIVERSITY APPLIED PHYSICS LABORATORY BILVER SPRING MARYLAND

Symbol	Definition	Typical Units	
^ψ c.g.	angle defined in Section 5, 4, 2	degrees	
,o	ambient density	slug/ft ³	
Subscripts			
с	refers to cross-track component		
Е	refers to conditions at reentry		
0	refers to conditions at zero roll rate except as noted		
R	refers to along-track component		
S. L.	sea level		
w	refers to quantity resulting from wind effe	cts	
ρ	refers to quantity resulting from density e	ffect	

A dot over a symbol refers to a first derivative with respect to time. An arrow over a symbol means a vector quantity.

- 274 -

COLUMN AND

1

10014

ad'xcy

THE JOHNS HOPKINS UNIVERSITY APPLIED PHYSICS LABORATORY SILVER SPRING MARYLAND

SUMMARY AND CONCLUSIONS

The factors most likely to cause impact dispersion are identified, and simple equations are given for estimating the magnitude of each dispersion. These factors include atmospheric wind, deviations from a standard atmospheric density profile, errors in mass, and aerodynamic asymmetries. The following examples are given to illustrate this discussion.

- Example 1 Dispersion resulting from wind and deviation in density at low and moderate altitudes.
- Example 2 Dispersion resulting from wind and deviation in density at high altitude.
- Example 3 Dispersion resulting from a trim angle of attack for a missile rolling at a constant roll rate.
- Example 4 Dispersion resulting from roll lock-in at low altitude.
- Example 5 Dispersion resulting from a combined asymmetry for which spin through zero roll rate occurs.

The major conclusions are:

1. Winds and departures from the assumed density are the chief sources of impact dispersion resulting from meteorological characteristics at the impact site. This dispersion increases rapidly as the flight path becomes shallow, as the missile velocity is decreased, and as the ballistic coefficient of the missile is decreased. The density effect is more sensitive to these missile and trajectory parameters than is the wind effect. Equations are presented for evaluating the dispersion for any wind and density profile.

- 275 -

and See

and the state of the

1.111

ことの作用ではたちにはおいておうでして利用者のがたられたれたがでいいですようという。

ないのなどの中に自己にはずたの日本でしたいと

2. The equations derived for evaluating the dispersion resulting from deviations in density may also be used to evaluate the dispersion resulting from perturbations in missile weight and drag.

3. The dispersion resulting from a vehicle that encounters roll resonance (but remains intact) is likely to be small. On the other hand, the dispersion resulting from a vehicle having a combined asymmetry that results in roll rate reversal is likely to be extremely large.

AND THE AREA AND A THE AREA AND A MARKED AND A REAL AND A REAL AND A REAL AND A REAL AND A REAL AND A REAL AND A

weather a part of the start of the start

THE JOHNS HOPKINS UNIVERSITY APPLIED PHYSICS LABORATORY BILVER SPRING MARYLAND

> Aiming instructions are required in order to deliver accurately a ballistic vehicle to a predetermined impact point. To obtain launch orders, a mathematical model containing all factors that influence the trajectory is constructed. However, even with the most elaborate trajectory simulation that can be devised, the model is not perfect and the vehicle departs from the intended trajectory. The magnitude of the miss distance from the intended impact point will be referred to as the impact dispersion. In Section 4.1 some of the important factors which affect dispersion and which occur at the end of the powered phase were discussed (error in terminal velocity and flight path angle). In this section we will discuss only those perturbations that occur during reentry.

> Accurate evaluation of dispersion requires use of a three- or a six-degree-of-freedom simulation. Simpler solutions, although lacking in accuracy, make it possible to isolate the important factors and provide insight into the general behavior of the various types of disturbances. These simpler solutions will be discussed here.

5.1 Meteorological Characteristics

The meteorological characteristics that may produce dispersion are:

- 1. Errors in wind speed and direction,
- 2. Errors in atmospheric density,
- 3. Errors in atmospheric temperature.

Temperature is included since, in addition to its effect on density, it affects also the speed of sound and, thus, the aerodynamic coefficients that depend upon Mach number. However, the effect of the speed of sound is usually small (see Section 3.1) and only the wind and density effects will be considered further.

5.1.1 Effect of Wind

Atmospheric winds are a major contributor to the dispersion of uncontrolled reentry vehicles. A simplified analysis is given in Ref. 5-1. The important portions of this theory are summarized below.

We make the following assumptions:

- 1. The effect of gravity is negligible.
- 2. The earth is nonrotating and flat.
- 3. The flight path angle is a constant = $\gamma_{\rm E}$.
- 4. The angle of attack is zero.
- 5. The wind is a horizontal head or tail wind (except as indicated),
- 6. The reentry vehicle is aerodynamically stable (the center of pressure is aft of the center of gravity).
- 7. The weight and C_A are constant.

With these assumptions the flight geometry in the absence of wind is shown in sketch (a).



an an the star hand the head to be

THE JOHNS HOPKINS UNIVERSITY APPLIED PHYSICS LABORATORY BILVER SPEING MARYLAND

> The velocity vector is along the body centerline. The only force acting is the axial force, A, and it also acts along the body centerline (by definition). Suppose now the body encounters a tail wind. The body forces that act immediately are shown in sketch (b). The missile velocity



vector and A are still along the missile centerline. However the velocity vector that produces aerodynamic forces on the missile is $\vec{V}_{A/M} = \vec{V}_w - \vec{V}$. This velocity vector acts at an angle, α_w , to the body centerline and causes a normal force, N, which acts at the vehicle center of pressure. This force causes the body to pitch (weathercock) until the body centerline and $\vec{V}_{A/M}$ are aligned. The conditions that exist when the weathercock motion has damped to negligible magnitude are shown in sketch (c).

ليلي

THE JOHNS HOPKING UNIVERSITY APPLIED PHYSICS LABORATORY SILVER SPRING MARYLAND



The normal force is now zero and $\vec{v}_{A/M}$ is along the body centerline. At all later times the body centerline remains aligned with $\vec{v}_{A/M}$; however, since A acts at a slight angle to \vec{v} the missile acquires a drift velocity relative to the no-wind trajectory. The force and velocity diagram including the drift velocity is shown in sketch (d). We will assume that the geometry shown in sketch (d) is representative of the reentry trajectory. At any given time (or altitude) the missile velocity is assumed to be the vector sum of the no-wind velocity for that time plus the drift velocity \vec{v}_R . Furthermore, it is assumed that $V_R \ll V$. Using these assumptions it is shown in Ref. 5-1 that the dispersion resulting from winds is given by the equation:

$$\delta_{w} = -\frac{H}{V_{E} \sin \gamma_{E}} I_{3} \sum_{j=1}^{N} V_{w_{j}} F_{w}$$
(5-1)

- 280 -

THE JOHNS HOPKING UNIVERSITY APPLIED PHYSICS LABORATORY SILVER BPRING MARVLAND

1

-

24





and the construction of the second second second second second second second second second second second second

Station of the

の一大日間と

1

THE JOHNS HORKINS UNIVERSITY APPLIED PHYSICS LABORATORY BILVER SPRING MARYLAND

where

H = RT/gI_g is given in Table 5-1

 F_w is given in Table 5-2,

The index number j refers to a particular altitude band. To account for a variable wind with altitude, the reentry altitude is divided into N bands where the bandwidth is small enough that V_w may be considered constant in each band.



The F_w term is a weighting factor that depends upon the altitude, altitude bandwidth, and $K_{S. L.}$. Since $K_{S. L.} = -\frac{P_{S. L.}}{\beta \sin \gamma_E}$, F_w is a function of the atmosphere considered ($P_{S. L.}$ - a minor variation), the body characteristic (β), and trajectory characteristic (γ_E). For a given body

- 282 -

design and type of atmosphere, F_W for a given altitude and altitude bandwidth is a function of γ_E only. For most applications this variation may be neglected. When this is done the values of F_W should be evaluated at the lowest value of $|\gamma_E|$ used in the missile design. The dispersion decreases rapidly with increasing $|\gamma_E|$. Therefore, although the percentage error in δ may be large when this procedure is followed, the magnitude of the error in δ will be small.

If the altitude bandwidth is constant, the magnitude of F_W is indicative of the sensitivity of δ_W to winds in a given altitude band. From Table 5-2 it is observed that the most sensitive band corresponds to the 20 000 to 25 000 foot band for $K_{S, L} = 1.5$, and the most sensitive altitude band decreases with increasing $K_{S, L}$ to the 10 000 to 15 000 foot band for $K_{S, L} = 6.0$. The calculation of the most sensitive band is treated mathematically in Ref. 5-1.

The ballistic coefficient β varies with altitude and Mach number. The value of β at hypersonic speeds should be used for this calculation (also for δ_{ρ} discussed in the next subsection).

The summation term in Eq. (5-1) is called the ballistic wind. By definition, a ballistic wind is a constant wind (with altitude) such that the dispersion is the same as that resulting from a variable wind. If V_w is constant at all altitudes, then Eq. (5-1) becomes:

$$\delta_{w} = -\frac{V_{w}H}{V_{E}\sin\gamma_{E}}I_{3}.$$
 (5-2)

The quantity preceding the summation sign in Eq. (5-1) is called the range partial for winds.

In Ref. 5-1, it is shown that Eq. (5-1) is valid for either head (or tail) winds or cross winds. If the wind

- 283 -

is neither a head (or tail) wind nor a cross wind, the total dispersion may be obtained by computing the head wind and cross wind components of δ_w and adding the results vectorially.

Equation (5-1) has been found to be in excellent agreement with results obtained using three- or sixdegree-of-freedom trajectory simulations provided KS. L. is not greater than about 6. For higher values of KS. L. the equation overestimates δ_w .

For applications where the body characteristics have been fully defined, the accuracy of Eq. (5-1) may be improved by the following procedure:

- 1. Using a three-degree-of-freedom trajectory program, evaluate δ_w for various values of V_E and γ_E and a constant wind.
- 2. Using these results and Eq. (5-1), compute I_3 and plot it as a function of γ_{E1} .
- 3. Using the trajectory program, evaluate δ_{W} resulting from a constant wind in each altitude band. These computations are made for one value of V_{E} and the lowest $|\gamma_{E}|$ to be used for the missile.
- 4. Using the results of step 3 and Eq. (5-1), compute F_w .

These new values of I_3 and F_w will differ slightly from those given in Tables 5-1 and 5-2. Using these new data, the accuracy of δ_w given by Eq. (5-1) may be improved, particularly for high values of K_{S_1,L_2} .

5.1.2 Effect of a Deviation in Density

The impact point of a reentry body is computed using some standard density profile (variation with altitude). If the actual profile is different from the standard

- 284 -

THE JOHNE HOPKINE UNIVERSITY APPLIED PHYSICS LABORATORY BILVER SPRING MARYLAND

profile, the missile velocity history is affected. Therefore, deviations in density from the standard profile are a source of impact dispersion.

Using an analysis similar to that used for δ_w , it is shown (Ref. 5-1) that:

$$\delta_{\rho} = \left(\frac{H}{V_{E} \sin \gamma_{E}}\right)^{2} \frac{g}{\tan \gamma_{E}} I_{5} \frac{\Sigma}{j=1} \frac{\Delta \rho}{\rho} F_{\rho} \quad (5-3)$$

where

 I_{g} is given in Table 5-3

F is given in Table 5-4

 $\Delta \rho$ is the deviation in density from the standard value (ρ) at a given altitude.

All other terms are as defined in the previous subsection on the effect of wind.

Again, F_0 is a weighting factor that depends upon the altitude, altitude bandwidth, and $K_{S, L}$. The most sensitive altitude band is the 25 000 to 30 000 foot band (see Ref. 5-1 for fuller discussion) and is nearly independent of $K_{S, L}$.

The summation term is called the ballistic density. By definition ballistic density is the constant value of $\Delta \rho / \rho$ that results in the same dispersion as the variable $\Delta \rho / \rho$.

The quantity preceding the summation sign is called the range partial for density deviation. Equation (5-3) has been found to be in excellent agreement with results obtained using three- and six-degree-of-freedom simulations, provided $K_{S,L}$ is not too high. In Ref. 5-1 it was shown that the conditions required for the accuracy of Eq. (5-3) are those given in Figs. 5-1 and 5-2. When these conditions are exceeded, the dispersion given by Eq. (5-3) is too high. THE JOHNS HOPKINS UNIVERSITY APPLIED PHYSICS LABORATORY BILVER SPRING MARYLAND

Martin Ballamaren e.

そうしょうと思想を行うないですが、そうとうないないです。 ちゅうちょう しゅうかい あっちょう しゅうかい ひょうせい かんかい たいしょう しょうせい いたい しょうしょう しょうしょう しょうしょう

For applications where the body characteristics have been fully defined, the accuracy of Eq. (5-3) may be improved following the same procedures given for winds in the preceding subsection.

Since values of F_{ρ} and F_{w} become very low at high altitude, one is tempted to conclude that the dispersion resulting from winds and deviations in density at altitudes above, say, 100 000 feet is negligible. For a high precision vehicle this conclusion is not necessarily valid, since the magnitudes of V_{W} and $\Delta \rho / \rho$ at these altitudes are usually very large.

It should be noted that, except for the direct effects $(\Delta \rho / \rho \text{ and } V_w)$, the variables contributing to dispersion resulting from winds and deviations in density are the same. However δ_{ρ} is more sensitive to changes in the variables.

5.2 Errors in Aerodynamic Coefficients

Errors or variation in aerodynamic coefficients resulting from reentry environmental effects (such as heating) may contribute to dispersion. The most sensitive coefficient from the dispersion standpoint is the axial force coefficient, C_A .

Dispersion results when a perturbation affects the γ history during reentry. This change in γ was shown (Ref. 5-1) to result primarily from changes in the K history. In Section 5.1.2 we interpreted changes in K in terms of deviations in density. We could just as well have interpreted a variation in K as a variation in β

or $\sin \gamma_E$. Since $\beta = \frac{W}{C_A S}$, then $K = -\frac{PC_A S}{W \sin \gamma_E}$, and the dispersion characteristics computed in Section 5.1.2 are equally valid for $\Delta C_A/C_A$, $-\Delta W/W$, or $-\Delta \beta/\beta$. The minus sign means that a given percent increase in W or β results in the same dispersion as the same percentage decrease in pressure, density, or axial force coefficient.

- 286 -

and the state of the

THE JOHNS HOPKINS UNIVERSITY APPLIED PHYSICS LABORATORY SILVER SPRING MARYLAND

Errors in the normal force coefficient have a small effect on δ . Since in general the angle of attack is not zero, a lateral force results, and the magnitude of this force for a given trajectory depends upon α and $C_{N_{\alpha}}$.

The dispersion resulting from this source arises from erroneous evaluation of lift and drag forces (see Sections 5.4.1, 5.4.2, and 5.5). This increment in dispersion is very small unless the error in $C_{N_{\alpha}}$ or the angle of attack is large.

Errors in the damping derivative may result in dispersion since this characteristic determines to some degree the $\overline{\alpha}$ history and thus the lateral force history early in the reentry phase and during periods of roll resonance (see Sections 4.4 and 4.5.5). Again the dispersion resulting from these errors is usually small.

Errors in the center of pressure location may result in dispersion since this factor also determines to some degree the $\overline{\alpha}$ history early in the trajectory and the $\overline{\alpha}$ resulting from asymmetries. The dispersion contribution is usually small.

5.3 Errors in Mass Characteristics

Errors in mass characteristics that contribute to dispersion are errors in W, $X_{C, g}$, and the moment of inertia terms. Of these, the most significant is the error in W, which results from uncertainty in the reentry weight and uncertainty in weight loss caused by heating effects during reentry. The dispersion resulting from an error $\Delta W/W$ is the same as that from an error, $-\Delta \rho/\rho$, as discussed in Section 5.2.

The effects of $\mathcal{C}_{c,g}$ and moment of inertia may cause small dispersions since these factors affect the α history during reentry.

THE JOHNS HOPKING UNIVERSITY APPLIED PHYSICS LABORATORY BILVER BERING MARTLAND

5.4 Asymmetries

The problems associated with asymmetries were discussed in detail in Section 4.5. Asymmetries produce angles of attack; angles of attack result in lift and a change in drag. For cases where angles of attack develop and the missile is rolling, the major effect on δ is the change in drag since the direction of the lift vector (in an inertial axis system) is constantly changing so that, over a period of time, the net effect of lift is "averaged out." The effect of $\overline{\alpha}$ for a rolling vehicle is usually small unless resonance conditions occur. In this case α may become large and the vehicle drag will increase significantly, causing the missile to fall short of the intended target, This case can be treated as a change in CA so that the procedures given in Section 5, 1, 2 may be used. A more serious contributor to dispersion is the case where $\overline{\alpha} \neq 0$ and the roll rate becomes zero. Even for a very short period of zero roll rate, large dispersions may result from small angles of attack. Since, for this case, the lift vector remains fixed in space, the vehicle moves off course in the direction of the lift. We will consider these two cases separately.

5.4.1 Rolling Vehicle with Angle of Attack

If asymmetries result in a trim angle of attack and the missile roll rate is not zero, the major contributor to dispersion is the effect of the increased drag resulting from $\overline{\alpha}$. The equation for the drag is:

$$D = C_D qS$$
 (5-4)

where, from Eq. (2-9):

$$C_{D} = C_{A} \cos \overline{\alpha} + C_{N} \sin \overline{\alpha}$$
. (5-5)

Therefore the change in C_D resulting from $\overline{\alpha}$ is:

$$\Delta C_{D} = C_{A} \cos \overline{\alpha} + C_{N} \sin \overline{\alpha} - C_{A} \approx C_{N} \sin \overline{\alpha} , \qquad (5-6)$$

- 288 -

THE JOHNS HOPKINS UNIVERSITY APPLIED PHYSICS LABORATORY BILVER SPRING MARYLAND

If the aerodynamic characteristics are linear so that $C_N = C_{N_{\alpha}} \overline{\alpha}$ and if $\overline{\alpha}$ is small:

$$\Delta C_{\rm D} = C_{\rm N_{\alpha}} \tilde{\alpha}^2 . \qquad (5-7)$$

Therefore the effect of $\overline{\alpha}$ may be estimated using the procedures given in Section 5.1.2 where:

$$\frac{\Delta \rho}{\rho} = \frac{\Delta C_D}{C_A} = \frac{C_N}{C_A} \frac{\sigma^2}{\sigma^2} . \qquad (5-8)$$

When working with telemetry data, it is convenient to express $\Delta C_D/C_A$ in terms of body accelerations. Substituting for α using Eq. (7-11):

$$\frac{\Delta C_{\rm D}}{C_{\rm A}} = \frac{C_{\rm A}}{C_{\rm N_{\rm A}}} \left(\frac{\overline{g}}{g_{\rm X}}\right)^2 \qquad (5-9)$$

Note that $\Delta C_D/C_A$ varies as the square of $\overline{\alpha}$, and therefore the effect tends to be small at low values of $\overline{\alpha}$ but tends to increase rapidly with increasing α .

5.4.2 Nonrolling Vehicle with Angle of Attack

The dispersion resulting from angle of attack for a nonrolling vehicle was considered in detail in Ref. 5-2 and the results are summarized below. Consider a vehicle with a fixed angle of attack that results in a lift force L. Consider an <u>inertial</u> axis system YZ as shown in the sketch. If the roll rate is low compared to the aerodynamic frequency, the lift (or normal) force is in the plane containing the asymmetric force (see Section 4.5.1). Therefore L rolls with the body. As long as the roll rate is not close to zero, the effect of L on dispersion is small since the direction of L (in the inertial axis system) is constantly changing. However if the roll rate becomes

- 289 -

THE JOHNS HOPKINS UNIVERSITY APPLIED PHYSICS LABORATORY BILVER SPRING MARYLAND

1 BARRAN



zero, even for an instant, the body will move off course in the direction of L. We will consider in this subsection the case of a vehicle that is spinning at reentry but, under the influence of body asymmetries, the directior of roll rate is reversed during entry. The duration of "near zero" roll rate is very short.

Assume that at h_0 the roll rate passes through zero and that at this instant the lift is in the vertical plane. A change in flight path angle, $\Delta \gamma$, results.



- 290 -

Assuming a straight line flight path and a small value for $\Delta \gamma$:

the contract of the second second

$$\delta_{\rm R} = \frac{h_o \Delta \gamma}{\sin^2 \gamma_{\rm E}} . \tag{5-10}$$

If L is in a plane perpendicular to the vertical plane and if $\Delta \gamma$ is interpreted as the resultant change in the direction of the velocity vector:

$$|\delta_{c}| = \left| \frac{h_{o} \Delta \gamma}{\sin \gamma_{E}} \right| = |\delta_{R} \sin \gamma_{E}|$$
 (5-11)

The impact pattern is elliptical with the major axis along the trajectory track and the minor axis across track. The ratio of the minor to major axis is $|\sin \gamma_E|$. The change in γ is given by the equation:

$$\Delta \gamma = \int \dot{\gamma} \, dt \qquad (5-12)$$

where (see Ref. 5-3);

$$\dot{\gamma} = \frac{Lg}{WV} - \frac{g\cos\gamma}{V} + \frac{V\cos\gamma}{R_o}$$

The change in γ of interest is just that resulting from L. Therefore:

$$\Delta \gamma = \frac{g}{WV_o} \int Ldt . \qquad (5-13)$$

言語などの

"

Beer Stragger 10

Equation (5-13) is valid provided the direction of L is fixed. However, since the direction of L changes over the time interval of interest, an effective value of $\int Ldt$ must be used. Based on six-degree-of-freedom simulation results, it may be shown that:

- 291 -

THE JOHNS HOPKINS UNIVERSITY APPLIED PHYSICS LABORATORY BILVER SPRING MARYLAND

$$\int Ldt \approx \frac{2.116L}{\sqrt{|\dot{p}_0|}}$$
(5-14)

where the constant is dimensionless. Writing $L = C_{L_{\alpha}} \overline{\alpha}_{o} q_{o} S$ where $C_{L_{\alpha}}$ is given by Eq. (2-10):

$$\int Ldt = \frac{2.116}{\sqrt{|\dot{p}_{o}|}} (C_{N_{\alpha}} - C_{A}) \overline{\alpha}_{o} q_{o} S. \qquad (5-15)$$

a filmer and a second second

Substituting Eqs. (5-15) and (5-13) into (5-10):

$$\delta_{\mathrm{R}} = \frac{2.116 \, \mathrm{h_og}}{\mathrm{WV_o} \, \sin^2 \gamma_{\mathrm{E}}} \frac{(\mathrm{C_N_o} - \mathrm{C_A})}{\sqrt{|\dot{p}_o|}} \, \overline{\alpha}_o \mathrm{q_oS} \, . \qquad (5-16)$$

In terms of normal force $N_o = C_{N_{\alpha}} \overline{\alpha}_o q_o S$:

$$\delta_{\mathrm{R}} = \frac{2.116 \, \mathrm{h_o g}}{\mathrm{WV_o} \, \sin^2 \gamma_{\mathrm{E}} \sqrt{|\dot{\mathbf{p}}_0|}} \left(1 - \frac{\mathrm{C_A}}{\mathrm{C_N_a}}\right) \mathrm{N_o} \quad (5-17)$$

In terms of lateral acceleration $\overline{g}_0 = \sqrt{g_Y^2 + g_Z^2} = \frac{N_0}{W}$:

$$\delta_{\rm R} = \frac{2.116 \ h_{\rm o}g \left(1 - \frac{C_{\rm A}}{C_{\rm N_{\alpha}}}\right)\overline{g}_{\rm o}}{V_{\rm o} \sin^2 \gamma_{\rm E} \sqrt{|\dot{p}_{\rm o}|}} \quad . \tag{5-18}$$

Equation (5-18) is convenient for analysis of telemetry data since \dot{p}_0 and g_0 are obtained directly from the data and V_0 and h_0 may be estimated using Eqs. (4, 3-22) and

- 292 -

THE JOHNS HOPKINS UNIVERSITY APPLIED PHYSICS LABORATORY BILVER SPRING MARYLAND

W ADARTS PERSONAL STREET

(4.3-5). Equations (5-16), (5-17), and (5-18) are valid for any combination of asymmetries.

We consider next the dispersion that results from the particular types of mass and aerodynamic asymmetries (N_o and r) shown in the sketch.



We will assume that the normal force resulting from r is very small so that the N_o results from an asymmetric moment defined by:

$$\overline{C}_{m_o} = \sqrt{C_{m_o}^2 + C_{n_o}^2}$$

The trim angle of attack, $\overline{\alpha}_{0}$, is given by Eq. (2-27) where, in the notation of the present section, $\overline{\alpha}_{TR} = \overline{\alpha}_{0}$ and $C_{m_{0}} = \overline{C}_{m_{0}}$. The roll acceleration, \overline{p}_{0} , is given by Eq. (4.5-35) where $\overline{q} = q_{0}$, $\overline{\alpha} = \overline{\alpha}_{0}$, and $\emptyset = \emptyset_{0}$. Substituting these values into Eq. (5-16), writing p_{0} in terms of H_{0} and P_{0} , and noting that $q_{0} = \frac{1}{2} p_{0} V_{0}^{2}$:

- 293 -

THE JOHNS HOPKINS UNIVERSITY APPLIED PHYSICS LABORATORY SILVER MPRING MARYLAND

$$\delta_{\mathrm{R}} = \frac{1.49 \left(1 - \frac{C_{\mathrm{A}}}{C_{\mathrm{N_{O}}}}\right)}{\sqrt{H_{\mathrm{O}}/g} W \sin^{2} \gamma_{\mathrm{E}}} \left[\frac{\overline{C}_{\mathrm{m_{O}}} SI_{\mathrm{X}}}{\frac{\rho \Delta X}{d} \sin \phi_{\mathrm{O}}}\right]^{1/2} h_{\mathrm{O}} \sqrt{P_{\mathrm{O}}} (5-19)$$

The constant is $2.116/\sqrt{2}$ and is dimensionless.

Equation (4.5-94) may be used to evaluate K_0 and

thus $h_0 \sqrt{P_0}$. Using Eq. (4.5-94) and Eq. (2-27), δ_R may be expressed in terms of \bar{x}_0 rather than \bar{C}_{m_0} and $r \sin \phi_0$, giving the result:

$$\delta_{\mathrm{R}} = 1.49 \,\overline{\alpha}_{\mathrm{o}} \sqrt{\frac{\mathrm{V}_{\mathrm{E}} \mathrm{P}_{\mathrm{S}, \mathrm{L}}}{\beta \mathrm{H}_{\mathrm{o}} \mathrm{P}_{\mathrm{E}}}} \frac{\left(\frac{\mathrm{C}_{\mathrm{N}}}{\mathrm{C}_{\mathrm{A}}} - 1\right)}{\sin^{2} \gamma_{\mathrm{E}}} h_{\mathrm{o}} \sqrt{\frac{\mathrm{P}_{\mathrm{o}}}{\mathrm{P}_{\mathrm{S}, \mathrm{L}}}} \left(1 - \mathrm{e}^{\frac{\mathrm{P}_{\mathrm{o}}}{\mathrm{P}_{\mathrm{S}, \mathrm{L}}}} \frac{\mathrm{K}_{\mathrm{S}, \mathrm{L}}}{2}\right)$$

$$(5-20)$$

Equation (5-20) shows that for a given $\overline{\alpha}_0$ (given \overline{C}_{m_0} and $\Delta X/d$ primarily), the dispersion varies with h_0 (variations in r sin \emptyset_0 if other variables are held constant) in accordance with the factor:

$$E = h_{o} \sqrt{\frac{P_{o}}{P_{S, L.}}} \left(1 - e^{-\frac{P_{o}}{P_{S, L.}}} \frac{K_{S, L.}}{2}\right) = f(h_{o}, K_{S, L.}). (5-21)$$

At sea level, this factor is zero since $h_0 = 0$. The factor increases with increasing h_0 to a maximum and then decreases, becoming zero at very high altitudes. The critical altitude (altitude for thakimum E) depends upon the type of atmosphere (P_{S.L.}) and upon K_{S.L.} (or upon β and/or γ_E). For the exponential atmosphere (Eq. (3-15))

- 294 -

it may be shown that E is a maximum at $h_0 = H_0$ for very small values of K_S. L. and increases to $h_0 = 2H_0$ for very large values of K_S. L. For the 1962 Standard Atmosphere, using H = constant = 23 000 feet, the critical altitude varies from about 23 000 feet at low K_S. L. to about 46 000 feet at high K_S. L. A plot of E versus h_0 is shown for several values of K_S. L. in Fig. 5-3.

In summary, the following conclusions were reached regarding the dispersion resulting from reversal of roll rate:

1. The dispersion may be in any direction. The locus of possible impact points is an ellipse centered at the nominal impact point. The major axis of the ellipse is along the flight path. The ratio of minor to major axis is $|\sin \gamma_{\rm m}|$.

2. The magnitude of the dispersion is a function of the many factors listed below, making it impossible to draw generalized conclusions regarding the effect of specific variables.

¯⊂_{mo} с_А Asymmetry Aerodynamic Х_{с.р.} $\gamma_{\rm E}$ W X_{c.g.} ${}^{p}E$ Initial Conditions Mass v_E $\mathbf{I}_{\mathbf{X}}$ Type Atmosphere d Geometry S

- 295 -

Only one type of combined asymmetries capable of resulting in $\overline{\alpha}_{O}$ and zero roll rate is listed above. However, other combined asymmetries such as $C_{L_{O}}$ and $\overline{C}_{m_{O}}$ could also result in roll rate reversal and large dispersion.

3. For given aerodynamic characteristics, mass, geometry, initial conditions, type of atmosphere, and \overline{C}_{m_0} , the $\overline{\alpha}_0$ is fixed. However, h_0 (and $\overline{\alpha}_0$ to a very small extent) varies with the values of r and \emptyset_0 . The dispersion that results as r and/or \emptyset_0 is changed is maximum when h_0 has some specific value. For the 1962 Standard Atmosphere, the value of h_0 corresponding to δ_{max} is between 23 000 and 46 000 feet, depending upon the value of K_{S.L.}.

5.5 Errors in Initial Reentry Angle of Attack and Body Rates

In some cases the initial reentry angle of attack, α , and body rates, p, q, and r, are not known accurately. During the high altitude portion of the trajectory, these initial conditions affect the $\overline{\alpha}$ history as discussed in Section 4.4. If $\overline{\alpha}$, p, q, and r are not zero in the altitude region where the aerodynamic restoring force is small, the missile experiences coning motion such that, in general, the body centerline does not rotate about the velocity vector. Therefore, in inertial axes, the lift force does not rotate about the velocity vector as it does for a rolling vehicle at altitudes where aerodynamic forces are predominant. During the early phase of reentry, initial values of α , p, q, and r cause dispersion resulting from both lift and drag effects. The drag effect always tends to cause the vehicle to fall short of the intended impact point. However, the lift effect may cause dispersion in any direction depending upon the direction of the lift vector (determined by the location of the coning motion relative to the velocity vector). In general the locus of impact points resulting from these initial conditions is an ellipse whose major axis is approximately along the flight path. The ratio of the minor to major axis is approximately $|\sin \gamma_{\rm E}|$. The drag effect shifts the center of the ellipse toward the launch site from the nominal point (impact for nominal α , p, q, and r).

The vehicle motion is rather complex, and a six-degreeof-freedom simulation is required to evaluate the magnitude of this source of dispersion. A typical order of magnitude of the semimajor axis for δ resulting from α is 5 to 10 feet per degree error in α . Therefore, the error is not negligible unless α is known fairly accurately.

5.6 Examples

5.6.1 Example 1

Estimate as a function of γ_E the dispersion resulting from a constant 100 ft/sec headwind ($V_w = -100$) and a constant 10% increase in density ($\Delta \rho / \rho = 0.1$) from the 1962 Standard Density for the following conditions:

> $V_{E} = 20\ 000\ \text{ft/sec}$ $\beta = 1000,\ 2000,\ \text{and}\ 3000\ \text{lb/ft}^{2}.$

Since dispersion is most sensitive to the altitude region from 20 000 to 30 000 feet, a value of H = 23 000 feet (see Table 3-1) is used.

From Eq. (5-2): $\delta_{\rm w} = -\frac{(-100) \times 23\ 000}{20\ 000} \frac{{\rm I}_3}{\sin\gamma_{\rm E}} = 115 \frac{{\rm I}_3}{\sin\gamma_{\rm E}} \ .$

From Eq. (5-3):

$$\delta_{\rho} = \left(\frac{23\ 000}{20\ 000}\right)^2 \quad \frac{32.2 \times 0.1 \times I_5}{\sin^2 \gamma_E \tan \gamma_E} = 4.26 \frac{I_5}{\sin^2 \gamma_E \tan \gamma_E}$$

From Table 3-1, $P_{S, L} = 2116 \text{ lb/ft}^2$. Then $K_{S, L} = \frac{2116}{\beta \sin \gamma_E}$. For a given value of β and γ_E , δ_W and δ_p

may be computed using I_3 and I_5 obtained from Tables 5-1 and 5-3. The results are shown in Fig. 5-4. This

- 297 -

figure illustrates the high sensitivity of dispersion to winds and density for shallow reentry angles and low values of β . It also illustrates the fact that δ_{ρ} becomes insignificant at large values of $\gamma_{\rm E}$ whereas $\delta_{\rm W}$ may be significant even for $\gamma_{\rm E} = -90^{\circ}$.

5, 6, 2 Example 2

Assume that in targeting a missile no corrections were made for winds and deviations in density from the 1962 Standard Density profile for altitudes greater than 100 000 feet. If the average wind at altitudes from 100 000 to 160 000 feet were a 300 ft/sec headwind and if the density were 30% greater than the 1962 Standard profile, estimate as a function of β the impact error resulting from these high altitude meteorological characteristics for V_E = 20 000 ft/sec and $\gamma_E = -25^\circ$. Use H = 23 000 feet.

From Eq. (5-1):

$$b_{w} = \frac{23\ 000(-300)}{20\ 000(-0,\ 422)} I_{3} \sum_{j=2}^{4} F_{w}$$
$$= -817\ I_{3} \sum_{j=2}^{2} F_{w}.$$

From Eq. (5-3):

$$\delta_{\rho} = \left(\frac{23\ 000}{20\ 000\ \times\ 0.422}\right)^2 \frac{32.2\ \times\ 0.3}{(-0.465)} I_5 \frac{4}{j=2} F_{\rho}$$
$$= -155\ I_5 \frac{4}{j=2} F_{w}.$$

Using values of I_3 , F_w , I_5 , and F_p given in Tables 5-1 through 5-4, δ_w and δ_p were computed as a

- 298 -

function of $K_{S, L}$. The values of β corresponding to $K_{S, L}$, were computed using the equation:

$$\beta = -\frac{2116}{K_{S, L}} (-0.422) = \frac{5000}{K_{S, L}}$$

Values of δ_w and δ_{β} are plotted in Fig. 5-5. The accuracy limit as given in Fig. 5-1 is also shown. Values of $\beta \ge 1230 \text{ lb/ft}^2$ fall within the limits of accuracy; values of δ shown for β less than 1230 lb/ft² will be slightly high.

This problem illustrates that bodies having low values of β tend to be sensitive to meteorological characteristics even at very high altitudes.

5.6.3 Example 3

For the reentry body having the characteristics given in Table 2-1, estimate the dispersion resulting from a constant trim angle of attack $\overline{\alpha}$ if the missile has a constant roll rate during reentry. The trajectory characteristics are V_E = 20 000 ft/sec and γ_E = -30°. Use H = 23 000 feet.

For this case the dispersion results from the increase in drag, and the effect on dispersion is equivalent to a change in density.

$$\frac{\Delta \rho}{\rho} (Eq. (5-8)) = \frac{C_N \alpha}{C_A} \overline{\alpha_0}^2$$
$$= \frac{2}{0.104} \overline{\alpha_0}^2$$
$$= 19.2 \ \overline{\alpha_0}^2 \text{ for } \overline{\alpha_0} \text{ in radians}$$
$$= 0.00585 \ \overline{\alpha_0}^2 \text{ for } \overline{\alpha_0} \text{ in degrees.}$$
$$= 299 - 2$$

THE JOHNS HOPKINS UNIVERSITY APPLIED PHYSICS LABORATORY SILVER SPRING MARYLAND

. Martin

The dispersion is computed using Eq. (5-3).

$$K_{S.L.} = \frac{-2116}{1000(-0.50)} = 4.23$$

 $I_5 = 21$.

Since $\Delta \rho / \rho$ is constant for all altitudes:

$$\sum_{j=1}^{N} \frac{\Delta \rho}{\rho} F_{\rho} = \frac{\Delta \rho}{\rho} = 0.00585 \overline{\alpha}_{0}^{2}.$$

Therefore:

$$\delta_{\rho} = \left(\frac{23\ 000}{20\ 000\ (-0.\ 50)}\right)^2 \ \frac{32.\ 2\times\ 21}{(-0.\ 577)}\ (0.\ 00585)\ \overline{\alpha}_0^2$$
$$= -36.\ 2\ \overline{\alpha}_0^2.$$

The missile will fall short by 9 feet for $\overline{\alpha}_0 = 1/2^\circ$, 36 feet for $\overline{\alpha}_0 = 1^\circ$, and 145 feet for $\overline{\alpha}_0 = 2^\circ$.

5.6.4 Example 4

Suppose in Example 3 the reentry body has an $\overline{\alpha}_0 = 1/2^\circ$ at reentry, the body passes through first resonance without lock-in, but the asymmetries are such that lock-in occurs at an altitude of 10 000 feet and remains until impact. Determine the dispersion.

Again, for this problem, dispersion results from the drag effect. However the angle of attack is amplified by the resonance phenomena. From Eq. (4.5-16), $\overline{\alpha} = A\overline{\alpha}_0$ and, since $\lambda = 1$ at resonance, A = 1/D. Using Eq. (4.5-60):

$$D = 0.115 \sqrt{K}$$
.

- 300 -

A Barris Alam

THE JOHNS HOPKINS UNIVERSITY APPLIED PHYSICS LABORATORY BILVER BRING, MARYLAND

Therefore;

$$\overline{\alpha} = \frac{8.7(1/2)}{\sqrt{K}} = \frac{4.35}{\sqrt{K}}$$

Substituting for K = $-\frac{P}{\beta \sin \gamma_E} = \frac{P}{P_{S.L.}}$ KS.L. = 4.23 $\frac{P}{P_{S.L.}}$,

$$\overline{\alpha} = \frac{2.11}{\sqrt{P/P_{S, L_i}}}$$

At h = 2500 feet $P/P_{S. L.} = 0.913$ $\overline{\alpha} = 2.21^{\circ}$ h = 7500 feet $P/P_{S. L.} = 0.757$ $\overline{\alpha} = 2.42^{\circ}$

From Example 3, the equivalent value of $\Delta \rho / \rho$ is 0.00585 $\overline{\alpha_0}^2$. Therefore, for h = 2500 feet, $\Delta \rho / \rho = 0.0286$; for h = 7500 feet, $\Delta \rho / \rho = 0.0343$. Assuming that the value of $\Delta \rho / \rho$ at h = 2500 feet is representative of the altitude band from sea level to 5000 feet and that the value of $\Delta \rho / \rho$ at h = 7500 feet is representative of the 5000 to 10 000 foot band;

 $\frac{17}{\sum_{j=16}^{\infty} \frac{\Delta \rho}{\rho}} F_{\rho} = 0.0286(0.049) + 0.0343(0.055) = 0.00329.$

Using Eq. (5-3):

 $\delta_{0} = \left[\frac{23\ 000}{20\ 000\ \times\ 0.5}\right]^{2} \frac{32.2}{(-0.577)} \times 21\ \times\ 0.00329 = -20\ \text{feet} \ .$

5, 6, 5 Example 5

The body having characteristics given in Table 2-1 has the following asymmetries:

1. A center of gravity offset of 0.042 inch.

1

2. A body centerline distorted into an arc of a circle. The slope of the centerline at the body nose and base differ by the angle $\Delta \theta$.

As a function of $\Delta \theta$, assess the probability of encountering spin through zero roll rate, and estimate the dispersion that results if spin through zero roll rate occurs. Neglect the possibility of encountering roll lock-in at negative roll rate. Use the 1962 Standard Atmosphere with initial conditions, $p_E = 1000^\circ$ /sec, $V_E = 20\,000$ ft/sec, and $\gamma_E = -30^\circ$.

From Eq. (2-35), $C_{m_0} = 0.008 \Delta \theta$. From Eq. (2-27):

$$\overline{\alpha}_{0} = 57.3 \frac{(0.008 \Delta \theta)}{2(0.114)} = 2 \Delta \theta$$

where $\Delta \theta$ and $\overline{\alpha}_0$ are in degrees. The altitude (in terms of K) at which the roll rate becomes zero is given by Eq. (4.5-94). Substituting for the given terms:

$$e^{-K/2} = 1 - \frac{\frac{1000}{57.3} (0.104)(0.079)}{2(0.001) \left(\frac{20000}{3.5}\right) \frac{2\Delta\theta}{57.3} \sin \phi_0}$$
$$= 1 - \frac{0.36}{\Delta\theta \sin \phi_0} \quad .$$

At sea level K = $K_{S. L.}$ = 2116/1000 (0.5) = 4.23. For zero roll rate at sea level:

$$\Delta \theta \sin \phi_0 = \frac{0.36}{1 - e^{-2.115}} = 0.41^\circ$$
.

Since ϕ_0 may have any value from 0° to 360° and assuming no correlation between the two types of asymmetries, the probability of encountering zero roll rate during reentry is:

- 302 -

Sec. 17 14

 $\overline{\mathbf{P}} = \frac{\Delta \phi_{0}}{360}$

where $\Delta \phi_0$ is the range in values of ϕ_0 for a given $\Delta \theta$ such that

 $\Delta \theta \sin \phi_{O} \ge 0.41$.

<u> </u>	sin Ø	<u> </u>	
0.41	1	0	0
0. 50	0.82 to 1	70	0.195
0.60	0.683 to 1	94	0.261
0,70	0.585 to 1	1 08	0,300
0.80	0.512 to 1	118	0.328
1.0	0.41 to 1	1 3 2	0.367
2.0	0.205 to 1	156	0.434
3.0	0.137 to 1	164	0.456

The probability of encountering zero roll rate is shown as a function of $\Delta \theta$ in Fig. 5-6. The altitude, h_0 , at which zero roll rate occurs is an important factor in the calculation of dispersion. For $\Delta \theta \ge 0.41$, h_0 may be any value from sea level to some upper limit obtained when sin $\theta_0 = 1$ or

$$e^{-K/2} = 1 - \frac{0.36}{\Delta \theta}$$
.

The upper limit to h_0 is shown as a function of $\Delta \theta$ in Fig. 5-7.

The dispersion that results when spin through zero roll rate occurs may be computed using Eq. (5-20). For a given $\Delta \theta$ (or $\overline{\alpha_0}$), the dispersion may be any value from 0 to some maximum value depending upon h₀ which, in

- 303 -

Store to a Olie

. Y

turn, depends upon \emptyset_0 . For low values of $\Delta \theta$, the maximum value of E (Eq. (5-21) and Fig. 5-3) correspond to the highest value of h_c ($\emptyset_0 = 90^\circ$). However for the larger values of $\Delta \theta$, the maximum value of E for this problem (K_{S.L.} = 4.23) occurs at $h_0 \approx 27000$ feet. Using h_0 data given in Fig. 5-7 and the E values given in Fig. 5-3, E_{max} for various values of $\Delta \theta$ are tabulated below.

<u> </u>	h at E o max	Emax	Δθ E _{max}
0.41	0	0	υ
0.5	13 000	8 600	4 300
0.6	21 000	10 800	6 480
0,7	27 000	11 300	7 920
0.8	27 000	11 300	9 050
1.0	27 000	11 300	11 300
2.0	27 000	11 300	22 600
3 . 0	27 000	11 300	33 900

Using Eq. (5-20) with $\overline{\alpha}_{0} = 2 \Delta \theta$:

$$\delta_{R_{max}} = 1.49 \frac{(2\Delta\theta)}{57.3} \sqrt{\frac{20\ 000\ \times\ 2116}{1\ 000\ \times\ 23\ 000\ \times\ \frac{1000}{57.3}}} \frac{\left(\frac{2}{0.104}-1\right)}{\left(1/2\right)^2} E_{max}$$

 $\delta_{R_{\max}} = 1.22\Delta \theta E_{\max}$.

The values of $\delta_{R_{max}}$ may be computed using the listed values of E_{max} , and the results are shown in Fig. 5-8.

The impact may be anywhere on an ellipse having a semimajor axis of any value from zero up to the values shown in Fig. 5-8. The major axis is along the trajectory. The semiminor axis is $\delta_R/2$ (see Eq. (5-11)).

- 304 -

Sec. 5 19 March 10 10 10

THE JOHNS HOPKING UNIVERSITY APPLIED PHYSICS LABORATORY BILVER SPRING MARYLAND

:000

The magnitude of the dispersion is highly dependent upon \emptyset_0 . In the previous discussion we considered only the maximum effect. The probability of encountering a given magnitude for the major axis of the dispersion ellipse may be obtained by plotting $\delta_{\rm R}$ versus $\phi_{\rm O}$ for various values of $\Delta \theta$ as shown in Fig. 5-9. The dispersions for values of $\phi_0 > 90^\circ$ are not shown since δ_R is symmetrical about $\phi_0 = 90^\circ$. The probability of encountering or exceeding a given value of $\boldsymbol{\delta}_{\mathbf{R}}$ for a given $\Delta\boldsymbol{\theta}$ is obtained from the ratio $\Delta \phi_0/360$ where $\Delta \phi_0$ is the range in values of ϕ_0 for which δ_R equals or exceeds the specified value. The probability plot is shown in Fig. 5-10. The values of $\overline{\mathbf{P}}$ shown for $\delta_{\mathbf{R}} = 0$ are the probabilities of encountering zero roll rate shown in Fig. 5-6. As an example of the interpretation of this figure, for $\Delta \theta = 0.5^{\circ}$ there is a 19% probability that zero roll rate will be encountered; there is a 16% probability that $\delta_R \ge 2500$ feet; the maximum possible dispersion is 5300 feet (the values of $\delta_{\rm B}$ at $\mathbf{\tilde{P}}$ = 0 are the values shown in Fig. 5-8).

Note that the dispersion that may be encountered if the missile spins through zero roll rate may be at least an order of magnitude greater than the dispersions considered in the previous problems. For example, the $\Delta \theta = 0.5^{\circ}$ asymmetry corresponds to a 1° trim angle of attack. It was shown in Example 3 that if the missile does not encounter zero roll rate the resulting dispersion is about 36 feet for $\overline{\alpha}_0 = 1^{\circ}$. The spin through zero roll rate phenomenon is a serious problem to the designer of reentry vehicles that require good impact accuracy. THE JOHNS HOPKINS UNIVERSITY APPLIED PHYSICS LABORATORY SILVER SPRING. MARYLAND

REFERENCES

- 5-1 L. S. Glover, <u>Approximate Equations for Evaluat-</u> ing the Impact Dispersion Resulting from Reentry <u>Winds and Deviations in Density</u>, <u>APL/JHU TG</u> 1132, September 1970.
- 5-2 B. F. Fuess, "Impact Dispersion Due to Mass and Aerodynamic Asymmetries," J. Spacecraft and Rockets, Vol. 4, No. 10, October 1967, pp. 1402-1403.
- 5-3 W. H. T. Loh, <u>Dynamics and Thermodynamics of</u> <u>Planetary Entry</u>, <u>Prentice-Hall Space Technology</u> Series, 1963.

- 307 -
THE JOHNS HOPKING UNIVERSITY APPLIED PHYSICS LABORATORY Silver Spring, Maryland





Preceding page blank

. . .

- 309 -

بيرك جوافيته بدحص

en egel

THE JOHNS HOPKING UNIVERSITY APPLIED PHYSICS LABORATORY BILVER SPRING MARYLAND



Fig. 5-2 LIMITING VALUES OF $\gamma_{\rm E}$

- 310 -

بالمهاد ومؤثر ومعاد الحسوية الأرار الأرائي

THE JOHNS IN JPKINS UNIVERSITY APPLIED PHYSICS LABORATORY SILVER SPRING MARYLAND





- 311 -



1111

ł

No. of Lot of Lot

all as the setting of a cost of a

Fig. 5-4 DISPERSION RESULTING FROM WIND AND DEVIATION IN DENSITY

- 312 -

THE JOHNS HOPKINE UNIVERSITY APPLIED PHYSICS LABORATORY BILVER SPRING MARYLAND THE JOHNS HOPKING UNIVERSITY APPLIED PHYSICS LABORATORY BILVER BPRING MARYLAND



Fig. 5-5 DISPERSION RESULTING FROM METEOROLOGICAL CHARACTERISTICS AT HIGH ALTITUDE

- 313 -

en el 1937 a seria a catalène de l'hangel en 1965 - El Canada Marine, fin el catalène e



THE JOHNS HOPKINS UNIVERSITY APPLIED PHYSICS LABORATORY AILUPA Sphine MARYLAND

₿ i.





10.00

No. 1 Sec.

3.1

unitatioparticates (255 + 11 + 1944

. It is a new warma privilage applying the state of the s

in the market was not a

- 315 -

THE JOHNS HOPKINS UNIVERSITY APPLIED PHYSICS LABORATORY Silver Epring. Maryland

2005

ł ſ.

そうちにたいたななながのといれていたいというないないないないないないないないないというないとうない





Langer B. C. The Software In Lorent and

:

ŗ ÷

- 316 -

1.1.1

THE JOHNE HOPKINE UNIVERSITY APPLIED PHYSICS LABORATORY BILVER SPRIME, MARYLAND

おいたいというものでものの国家に行い

13 i.

ì

÷.,

1

í ÷.



- 317 -

.

and the accuracy operation of the second second second second second second second second second second second

aban fin se may gold have been and

THE JOHNS HOPKINS UNIVERSITY APPLIED PHYSICS LABORATORY SILVER SPRING MARYLAND





and the mathematic in

.

THE JOHNS HOPKINS UNIVERSITY APPLIED PHYSICS LABORATORY Bilver Spring Manyland

Table 5-1

Values of I₃

^K S.L.	13
0	0
0.5	0,25
1.0	0.57
1.5	0.92
2.0	1,33
2.5	1,84
3.0	2.44
3.5	3,20
4.0	4.13
4.5	5.25
5,0	6.60
5.3	8.17
6.0	9.90

- 319 -

فخريق بيسبوب

ويتعالم ويتعالد ويراك الم

. . .

.

and the state of the second second second second second second second second second second second second second

AND ALL AND A

· . .

THE JOHNS HOPKING UNIVERSITY APPLIED PHYSICS LABORATORY BILVER SPRING MARYLAND

Altitude

	Value	s of F w	
K _{S.1.}			
5	2.0	3.0	4,0
007	0.008	0,005	0.00
005	0,004	0,004	0,00

Table 5-2

.....

(thousands	ands S.L.						
of feet)	J	1.5	2.0	3, 0	4,0	6,0	6, 0
160-0	1	0,007	0.008	0,005	0,004	0,003	0,003
140-160	2	0,005	0,004	0,004	0,002	0,002	0, 001
120-140	3	0.010	0,009	0, 007	0,006	0,005	0.004
100-120	4	0.025	0,023	0,018	0,013	0, 011	0, 01 0
90-100	5	0,021	0,019	0,015	0,011	0, 009	0, 008
80-90	6	0,030	0, 027	0, 021	0,017	0,015	0, 01 3
70-80	7	0,045	0,040	0, 030	0,024	0, 020	0,016
60-70	8	0, 064	0,058	0, 047	0.035	0,029	0, 025
50-60	9	0.085	0,079	0,085	0,053	0.042	0, 036
40+50	10	0.109	0,106	0, 092	0.077	0.066	0.063
30-40	11	0.147	0,143	0,126	0,110	0.100	0, 097
25-30	12	0, 080	0,086	0, 078	0.071	0.068	0.066
20-25	13	0.103	0,100	0,092	0,086	0.085	0,086
15-20	14	0,086	0.090	0, 097	0.105	0.114	0.117
10-15	15	0, 089	0.080	0,110	0.142	0,158	0,168
5-10	16	0,058	0,070	0,101	0,130	0,146	0.155
S. L5	17	0, 047	0,060	0.092	0.114	0,127	0,132

Note. These data are applicable to a 1962 Standard Atmosphere. However, they may be used for any atmosphere by changing the altitude limits for each band. For example, the j = 8 data apply to the altitude band from an altitude corresponding to $P/P_{S,L}$ = 0.0714 (60 000 feet for the 1902 Standard Atmosphere) to an altitude corresponding to $P/P_{S,L}$ = 0.0714 (60 000 feet for the 1902 Standard Atmosphere) to an altitude corresponding to $P/P_{S,L}$ = 0.0443 (70 000 feet for the 1962 Standard Atmosphere) to a standard Atmosphere (the standard Atmosphere) to a standard Atmosphere (the standard Atmosphere) to a standard Atmosphere (the standard Atmosphere (the standard Atmosphere (the standard Atmosphere (the standard Atmosphere (the standard Atmosphere (the standard Atmosphere (the standard Atmosphere (the standard Atmosphere (the standard Atmosphere (the standard Atmosphere (the standard Atmosphere (the standard Atmosphere (the standard Atmosphere (the standard Atmosphere (the standard Atmos aphere).

gety page they beaution

THE JOHNS HOPKINS UNIVERSITY APPLIED PHYSICS LABORATORY BILVER BRING MARYLAND

.

b

Table 5-3

Values of I_5

^K S.L.	I ₅
0	0
Ú.5	0,50
1.0	1.05
1.5	1,85
2.0	3.00
2.5	4.90
3.0	7.55
3.5	11.30
4.0	17.30
4.5	27,00
5.0	41.00
5,5	61.00
6.0	92.50

A. S. S. S. March 1991 .

THE JOHNS HOPKINS UNIVERSITY APPLIED PHYSICS LABORATORY BILVER SPRING MARYLAND

وخانجاته عدوكان أسالمي

State of the second second second second second second second second second second second second second second s

Altitude		^K s. L.					
(feet x 1000)	j	1.5	2.0	3,0	4.0	5,0	6.0
160	1	0, 004	0.012	0,009	0,006	0.004	0.004
140-160	2	0.014	0, 01 4	0,009	0.006	0.005	0.004
120-140	3	0.027	0.023	0.017	0.012	0.008	0.007
100-120	4	0,053	0,046	0,035	0,025	0,019	0.017
90-100	5	0, 045	0,039	0,028	0.021	0,015	0.013
80-90	6	0.055	0,050	0 , 039	0,029	0,025	0.022
70-80	7	U. 069	0,067	0.058	0.045	0,034	0.030
60-70	8	0, 088	0.085	0,077	0 .065	0.056	0,051
30-60	9	0.118	0,111	0.099	0.09 0	0,087	0,086
40-50	10	0.142	0.138	0,130	0.123	0,120	0,118
30-40	11	0.140	0,146	0.154	0.157	0.158	0,156
25-30	12	0.072	0,076	0.083	9.088	0.092	0.095
20-25	13	0.068	0.074	0.082	0,090	0.098	0,102
15-20	14	0,044	0,048	0.064	0.078	0.087	0,091
10-15	15	0. 030	0,032	0,051	0,066	0.076	0,080
5-10	16	0.019	0.022	0,037	0.053	0.061	0.065
0-5	17	0.012	0.017	0,028	0.046	0,055	0,059

Table 5-4

Values of F_{ρ}

Note: See note to Table 5-2.

- 322 -

計画に、

A Second Second

÷.

THE JOHNS HOPKINS UNIVERSITY APPLIED PHYSICS LABORATORY BILVER SPRING, MARYLAND

.

6. TRAJECTORY SIMULATION

S TRAJECTORY SMULATION

i

in the states we describe a state of the

¢,

3

. ...

2.67 (22) 121 111 1

• • • •

THE JOHNS HOPKINS UNIVERSITY APPLIED PHYSICS LABORATORY SILVER SPRINS MARTLAND

SUMMARY

The discussion of the motion of ballistic missiles would be incomplete without some comment on the "work horse" of all trajectory analysts - trajectory simulations. It is beyond the scope of this report to discuss any simulation in detail. However, for the uninitiated, a few brief comments are presented regarding the basic ingredients of all simulations, the relative operational cost of several types of simulations, and some characteristics of a sixdegree-of-freedom simulation currently in use at APL.

Preceding page blank

- 325 -

والمحافية والمحافظ والمتحاف والمحافظ

A trajectory simulation is a computer program by which the time history of the vehicle motion may be determined. Given a set of initial conditions and the particular quantities for which a time history is desired (outputs), the reentry body simulation consists of five major parts (see sketch, page 328):

- 1. Equations of motion,
- 2. Earth characteristics,
- 3. Atmospheric characteristics,
- 4. Mass characteristics,
- 5. Aerodynamic characteristics,

Each of the major parts may be very simple or very complex. The equations given in Section 4.3 provide the basis for a very simple simulation in which the equations of motion are reduced to a single degree of freedom (motion along a constant flight path angle), the earth is flat and nonrotating, the acceleration of gravity is zero, the atmosphere obeys the hydrostatic equation, the wind is zero, and the vehicle mass and aerodynamic characteristics are defined by a single parameter (β). However, the usefulness of this simulation is limited to those studies for which approximate time histories of a very few trajectory parameters are sufficient. For many types of work this simple simulation is not adequate.

Increasing the complexity (and cost) over the one-degree-of-freedom (1 DOF) problem, a threedegree-of-freedom (3DOF) trajectory simulation may be used. Usually the 3DOF designation refers to degrees of freedom in the three orthogonal linear dimensions. In this case the simulation is also called a "point mass" simulation. The only mass and aerodynamic characteristics used are the body weight and axial force characteristics. Any atmosphere and earth model may be used. This simulation is reasonably uncomplicated but has sufficient accuracy for most types of flight test analyses, for defining targeting data, and for evaluating most types of range partials (sensitivity of impact dispersion to various anomalies).

- 327 -

Preceding page blank

THE JOHNS HOPKINS UNIVERSITY APPLIED PHYSICS LABORATORY BILVER SPRING MARYLAND

:

And the second second

ì

i.

ł ĺ.

> i

0 3 and the second second



i,

÷

ľ

.

. . . .

ļ

.

.

1 -1 -

}

. . $e^{i\frac{1}{2}} f_{i,j} e_{ij}$

and the

٠,

- 328 -

For evaluating the detailed dynamic behavior of missile motion and for calculating precise impact locations, a six-degree-of-freedom (6DOF, three linear and three rotational) program is required. In this simulation, the body mass characteristics are defined by weight, center of gravity location (specified by three coordinates), moments of inertia, and products of inertia. The aerodynamic terms include the axial force, normal force, center of pressure location, pitch (or yaw) damping, roll damping, and asymmetric force and moment terms as required for a specific study. This type of simulation is the only one (of the ones discussed) that accurately accounts for angle of attack effects. Any atmosphere and earth model may be used.

There are many modifications to these three basic simulations. For a given study the analyst will select the type of simulation based on a trade-off between accuracy and cost. For a typical reentry trajectory, the approximate machine cost of running the three simulations discussed above is as follows:

DOF	Negligible,
NOF	\$1 per trajectory,
BDOF	\$30 per trajectory.

As an illustration of a typical 6DOF reentry body simulation, one of the simulations currently in use at Al·L is described briefly.

6.1 Initial Conditions

1. Geometric altitude,

- 2. Missile velocity and flight path angle relative to rotating earth,
- 3. α and β components of $\overline{\alpha}$ (see Fig. 4-11),
- 4. Initial roll orientation of the body axes with respect to the inertial axes,
- 5. Body rates (roll, pitch, yaw).

- 329 -

6.2 Missile Mass Characteristics /

- 1. Weight,
- 2. X, Y, Z coordinates of the center of gravity,
- 3. Moments of inertia (about X, Y, Z axes),
- 4. Products of inertia (about X, Y, Z axes).

Provision is made for two sets of characteristics so that the trajectory for a two-stage reentry body may be computed. Provision is also made for varying the weight and the two lateral coordinates of the center of gravity as a function of altitude. This option is used for studies in which body shape changes as a result of aerodynamic heating.

6.3 Aerodynamic Characteristics

Some of the aerodynamic data are in an aerodynamic subroutine of the program and some are inputs. The subroutine provides the axial force coefficient, normal force coefficient, center of pressure location, pitch (or yaw) moment damping coefficient, and roll moment damping coefficient. The axial force is a function of Mach number, angle of attack, altitude, and velocity. The center of pressure, pitch damping, and normal force data are functions of Mach number and angle of attack. The roll damping coefficient is a function of velocity and altitude. Provision is made for using two sets of data corresponding to the two sets of mass data.

The aerodynamic inputs include the asymmetric pitching, yawing, and rolling moment coefficients as a function of altitude and multiplying factors as a function of altitude for varying the aerodynamic data included in the aerodynamic subroutine.

- 330 -

6.4 Atmosphere Characteristics

The 1962 Standard Atmosphere is built into the simulation but any variation of density, temperature, wind, and wind direction with altitude may be used.

6.5 Earth Model

The WGS/60 ellipsoidal rotating earth model is used.

6.6 Equations of Motion

The six equations of motion (in body axes) used are the three rotational degrees of freedom given in Eqs. (4.5-1), (4.5-2), and (4.5-3), and the three linear degrees of freedom given in Eqs. (4.4-4), (4.4-5), and (4.4-6). The aerodynamic forces and moments are computed in the body axis system. Accelerations are then computed in the inertial axes, and velocities and translations are computed by integration.

6.7 Output

Various output quantities are available. The output format currently programmed provides the following time histories:

- 1. Three components of acceleration (body axes),
- 2. Total angle of attack,
- 3. Pitch, yaw, and roll rates (body axes),
- 4. Mach number,
- 5. Velocity (relative to rotating earth),
- 6. Dynamic pressure,
- 7. Flight path angle (relative to rotating earth),
- 8. Altitude (geometric),
- 9. Downrange and crossrange coordinates of the trajectory.

- 331 -

THE JOHNS HOPKINS UNIVERSITY APPLIED PHYSICS LABORATORY BILVER SPRING MARYLAND

The final latitude and longitude are also listed.

Several options are available regarding the printout time interval. The integration time step is variable (with a minimum of 0.0001 second), and readouts may be obtained at every time step, any multiple of the time step, every second, or any multiple of a second.

For a 3DOF simulation only the first and second initial confidence, the first mass characteristic, the axial force coefficient (as a function of Mach number and altitude only), and the three linear degrees of freedom are used. Hence the computer cost is lowered by an order of magnitude compared with the cost of using the 6DOF program. THE JOHNS HOPKING UNIVERSITY APPLIED PHYSICS LABORATORY SILVER SPRING, MARYLANS

I

1

Ι

5

÷ .

Ì.

ويتبر والأل والديل

n

7. FLIGHT DYNAMICS DATA ANALYSIS

 \overline{x}^{\pm}

の時代の見たりのためというというです。これです。

「おおいまでののないないないとうとう」

ł

and the second second second

NY MARANA SA DARA SA D

••

THE JOHNE HOPKING UNIVERSITY APPLIED PHYSICS LABORATORY BILVER SPRING MARYLAND

が設置においれたとなったというというというというと

į,

行行

Į

SYMBOLS FOR SECTION 7

Symbol	Definition	Typical Units
А	axial force	pounds
А	amplification factor	-
a	semimajor axis of the dispersion ellipse	feet
b	semiminor axis of the dispersion ellipse	feet
CA	axial force coefficient, A/qS	-
C _N	normal force coefficient, N/qS	-
C _N α	slope of the normal force coefficient, $\partial C_N / \partial \alpha$	per radian
c _{mq}	pitch damping coefficient, $\partial C_m / \partial (qd/V)$	-
D	damping parameter (see Fig. 7-11)	-
F	resultant transverse force	pounds
g	resultant transverse acceleration,	
	$\sqrt{g_Y^2 + g_Z^2}$	g
g _X	longitudinal acceleration	g
g _Y	lateral acceleration	g
gz	normal acceleration	g
h	altitude	feet
^I x	roll inertia	slug-ft ²
I _Y	pitch inertia	slug-ft ²
ĸ	parameter defined in Eq. (4.3-6)	-

Preceding page blank

- 335 -

- interest

a go Chrontein

···*

THE JOHNS HOPKING UNIVERSITY APPLIED PHYSICS LABORATORY SILVER BPRING MARYLAND

Bes Min Draw and a second

Symbol	Definition	Typical Units
m	mass	slugs
N	normal force	pounds
Р	atmospheric pressure	1b/ft ²
þ	roll rate	deg/sec
q	pitch rate	or
r	yaw rate	rad/sec
q or q	dynamic pressure	$1b/ft^2$
S	reference area	ft ²
v	velocity	ft/sec
W	weight	pounds
Χ,Υ,Ζ	body coordinates with origin at center of gravity. X coordinate is parallel to the body centerline, positive for- ward; Y coordinate is perpendicular to body centerline, positive to the right looking forward; Z coordinate forms a right-hand system with X and Y, positive direction is down when looking forward.	. · · ·
x _{c.g.}	distance from nose to center of gravity	feet
х _{с. р.}	distance from nose to center of pressure	feet
α	angle of attack in X-Z plane	degrees
β	angle of yaw in X-Y plane	degrees
β	ballistic coefficient, W/C_AS	$1b/ft^2$
ā	total trim angle of attack	degrees
γ	flight path angle	degrees
$\Delta \mathbf{X}$ $\Delta \psi$	Xc. p. Xc. g. change in trim maneuver plane due	feet
	to roll rate	degrees
λ	resonance index (see Fig. 7-11)	-

- 336 -

THE JOHNE HOPKING UNIVERSITY APPLIED PHYSICS LABORATORY BILVER BPRING, MARYLAND

σ

د. برد ۱۹ ماه و بروه امیرود رو <mark>بالاسی</mark>د از میگرد.

Î

Symbol	Definition	Typical Units
ψ	orientation of trim maneuver plane, positive measured clock- wise from Y axis looking forward	degrees
ωΑ	aerodynamic frequency (see Fig. 7-10) deg/sec
Subscripts		
e. p.	center of pressure	
c. g.	center of gravity	
E	conditions at beginning of reentry (400	000 feet)
0	conditions when $p = 0$	i.

component in the X, Y, or Z direction, respectively X, Y, Z

A dot over a symbol means the first derivative with respect to time. Two dots mean the second derivative with respect to time.

- 337 -

.....

THE JOHNS HOPKING UNIVERSITY APPLIED PHYSICS LABORATORY BILVER BPRING MARYLAND

SUMMARY

A detailed discussion is presented of the procedures currently used at APL in the analysis of vehicle motion during reentry. This discussion includes comments on:

- 1. The basic data available,
- 2. Basic data processing,
- 3. Conversion to engineering quantities,
- 4. Special analyses for dispersion and "roll lock-in."

Examples are given of:

- 1. Typical flight test records,
- 2. A method of extracting range time from these records,
- 3. Typical body rate and acceleration traces,
- 4. Work sheets for computing angle of attack, aerodynamic pitch frequency, and several quantities related to roll resonance.

Preceding page blank

part 1811 and 1818 and the

- 339 -

AT ALL THE

THE JOHNS HOPKING UNIVERSITY APPLIED PHYSICS LABORATORY BUYER BRIDG MAR 1430

7.1 Available Data

The flight data most commonly available for dvnamics analysis are the pitch, vaw, and roll rates and the normal, lateral, and longitudinal accelerations at the reentry body center of gravity. In addition, the weather data in the reentry area are required for a detailed reentry dispersion analysis. These data include the atmospheric pressure and temperature, ar² the wind speed and direction as a function of altitude. Of course, the mass characteristics of the reentry body are required. These include the weight, center of gravity location (three coordinates), and the moments and products of inertia. The initial conditions of the reentry trajectory are also required. These conditions are frequently specified at 400 000 feet altitude, which is an altitude well above the sensible atmosphere, and include the velocity, flight path angle, latitude and longitude, flight path azimuth, and roll rate. Additional initial conditions must be specified if the reentry body motion at very high altitudes is being examined or if the precise impact point must be determined. These quantities include the initial angle of attack, side slip angle, roll angle, pitch rate, and yaw rate.

Although in general all of these data are required for a precise analysis of the reentry dynamics, it occasionally happens that some part of the data is not available. When this situation arises, the analysis sometimes may be continued without seriously compromising its acculacy by substituting nominal values for the missing data. As an example, the weather data may be unavailable, but it may be possible to substitute weather data (except for winds) that were determined during previous tests in that part of the world (and during that time of the year) and still obtain satisfactory results. Or it may be possible to choose one of the available standard atmospheres and obtain a satisfactory result.

Should the acceleration at the reentry body center of gravity be unavailable, accelerations measured at other locations may be used to calculate the acceleration at the

Preceding page blank

- 341 -

center of gravity. An accelerometer positioned at a point away from the center of gravity will be influenced by the body angular rates and accelerations; the data from such an accelerometer may be corrected to yield the acceleration at the center of gravity by means of the following equations:

$$g_{X} = \ddot{X}_{\text{measured g}} - \frac{Z}{32, 2} (\dot{q} + rp) - \frac{Y}{32, 2} (pq - \dot{r}) + \frac{X}{32, 2} (q^{2} + r^{2}) (7-1)$$

$$g_{Y} = \ddot{Y}_{\text{measured g}} - \frac{Z}{32, 2} (rq - \dot{p}) + \frac{Y}{32, 2} (r^{2} + p^{2}) - \frac{X}{32, 2} (\dot{r} + pq) (7-2)$$

$$g_{Z} = \ddot{Z}_{\text{measured g}} + \frac{Z}{32, 2} (p^{2} + q^{2}) - \frac{Y}{32, 2} (\dot{p} + qr) - \frac{X}{32, 2} (pr - \dot{q}) (7-3)$$

7.2 Data Sources

Redundant receiving stations are frequently used to ensure reception of telemetry from a reentry body. Ground stations, shipboard stations, and airborne stations may be used. The quality of the data received at any one of these stations may vary widely depending upon the geometry of the trajectory relative to the station location. As is now commonly known, a body reentering the atmosphere develops a plasma sheath that impedes RF transmission. The quality of the signal received through this sheath varies greatly with the aspect of the body relative to the station. Reentry bodies with a low ballistic coefficient (essentially a low weight to drag ratio) may have a plasma sheath of such impedance that transmission of the reentry data acquired during this "plasma blackout" must be delayed.

It may be necessary to examine the signal received by several stations in order to acquire the data needed for a reentry analysis.

7.3 Data Format

Flight test data, as currently processed at APL, are in the form of either DITAR records or DITAP records.

- 342 -

In both cases the flight functions are digitized by dividing the calibrated range of the sensing instrument into 255 intervals. A value of the function may then be represented by a number of counts or bits ranging from 0 to 255. In the case of the DITAP records, the count level is plotted as a function of time, and the record is an analog display of the function. An example of the data format is shown in Fig. 7-1. Timing marks are applied to the DITAP records according to the scheme in Fig. 7-2. In the case of the DITAR records, the bits are plotted in a condensed form that permits many telemetry functions to be displayed on one record. Two traces are required for each function, one trace to count from 0 to 15 and the second trace to count in increments of 16 from 0 to 240. By adding the counts from both traces, any number from 0 to 255 can be represented. An example of the data format for DITAR records is shown in Fig. 7-3. An example of the timing marks on the DITAR records is shown in Fig. 7-4. The series of marks on the margin of the record may be interpreted as time of day.

The DITAR data format presents the data very compactly, and although a two point calibration is attached, it is sometimes difficult to establish the zero point from which one should begin counting the bits. From certain physical considerations, a value for one point on the flight records can sometimes be established, and this point can be used to confirm that the correct zero bit level has been chosen.

The following is an example of these techniques for confirming the bit level. The pitch and yaw rate gyro data, before reentry, will have a sinusoidal shape. Except for the effect of principal axis misalignment, which is usually small, the sine wave will be centered about zero rate. From the calibration tables used to convert to engineering units, the number of counts for zero rate can be determined, and this then fixes one point on the rate gyro data curve. Subsequent values can be determined simply by counting bits. Furthermore, the frequency of the pitch and yaw rate gyro sine wave is related

- 343 -

to the roll rate by means of Eq. (4.4-21) when the body is unaffected by the atmosphere. Therefore the exoatmospheric roll rate can be fixed by determining the frequency of the pitch rate gyro output, and subsequent values can be determined by counting. The count level for the accelerations may be established by noting that the exoatmospheric accelerations are zero. The bit level, therefore, must be the one that corresponds to zero acceleration. Subsequent bit levels can be established by counting.

In the case of the longitudinal acceleration data, advantage can also be taken of the two ranges of instruments that are frequently used. One accelerometer is sensitive enough to detect the low level of acceleration at the beginning of reentry, and the second is sturdy enough to measure the peak g. The less sensitive instrument may be expected to go off scale at some point in the reentry. The count level is 255 at the point where the range of the instrument was just exceeded. The acceleration at this point may be determined easily from the calibration and, of course, the less sensitive instrument should also be recording this level of acceleration so that, by means of the calibration, the count level for the less sensitive instrument is established.

Future plans call for a computerized conversion of the data from counts to engineering units as well as for machine plotting of the data in engineering units.

The flight data most commonly used for dynamics analysis are the roll rate and the three accelerations. Typical examples of these functions (as well as the pitch rate) are shown in Figs. 7-5 to 7-8.

7.4 Data Processing

Further data processing depends upon a knowledge of the flight Mach number history. The flight Mach number could be calculated based on the longitudinal deceleration if the altitude were known; however, the altitude history generally is not available. The recourse, then,

- 344 -

is to determine the flight Mach number from a computer simulation. The results of the simulation must be checked against the flight data to ensure an adequate simulation of the reentry trajectory. The easiest check that can be made is a comparison of the longitudinal acceleration history as predicted by the simulation with the flight data results. The velocity history of the body is obtained by integrating the acceleration history. If the simulated longitudinal acceleration history is verified by the flight data. and the transverse accelerations are small enough that the longitudinal acceleration represents the total acceleration, then the simulated velocity history (hence Mach number history) will be correct. For the purpose of matching the simulation results to the flight acceleration, it is convenient to compare the results based on the time after the occurrence of the peak g. The time from peak g to impact is a good check on the calculated Mach number history.

If the acceleration from a nominal simulation does not agree with flight results, the factors in the flight that are not nominal (such as the aerodynamic asymmetries) must be included in the simulation in order to produce a match between the flight results and the simulation. Occasionally it is necessary to modify some of the aerodynamic coefficients in order to produce an acceptable match.

At points where the altitude is known (generally the only point is at impact) the Mach number may be calculated based on the following equations.

By definition of the axial force coefficient:

$$C_{A}\overline{q}S = A$$
 (7-4)

or

$$\frac{C_A \overline{qs}}{W} = \frac{A}{W} = -g_X . \qquad (7-5)$$

- 345 -

We may introduce the definition of dynamic pressure to give:

$$\frac{C_A(\gamma/2)PM^2S}{W} = -g_X \qquad (7-6)$$

so that

$$C_A M^2 = \frac{-g_X W}{(\gamma/2) PS} . \qquad (7-7)$$

States Subhard

The axial force coefficient, C_A , is principally a function of Mach number (and also a slowly varying function of angle of attack and altitude). For a particular angle of attack and altitude, we can plot $C_A M^2$ versus Mach number based on predicted aerodynamic coefficients. We may then use flight data to calculate a value of $C_A M^2$ from Eq. (7-7), enter a plot of $C_A M^2$ versus Mach number, and determine the Mach number. As mentioned previously, this scheme can usually be applied only at the impact point since the altitude (which determines P and has some effect on C_A) is not known at other points along the trajectory.

With the Mach number history established by matching the simulation longitudinal acceleration to the flight results, the angle of attack may be computed as follows.

The maneuver acceleration is related to the transverse force by Newton's second law:

$$\frac{F}{W} = \overline{g} \qquad \text{where } \overline{g} = \sqrt{g_Z^2 + g_Y^2} . \tag{7-8}$$

Expressing the transverse force in terms of the normal force coefficient, we have:

$$\frac{C_{N} \overline{\alpha} qS}{W} = \overline{g} \quad . \tag{7-9}$$

- 346 -

From a similar development considering the longitudinal acceleration we have:

$$\frac{c_A \overline{q}s}{W} = -g_X \quad (7-10)$$

The above equations are combined to give:

$$\overline{\alpha} = \frac{\overline{g}}{-g_X} \frac{C_A}{C_N}$$
(7-11)

The use of Eq. (7-11) with the flight values for the accelerations and the established (wind tunnel test, etc.) values for the aerodynamic coefficients permits determination of trim angle of attack as a function of time.

For routine flight test analysis, the envelopes of the normal and lateral accelerations are usually all that is extracted from the flight records, and the mean value of the envelope at a particular point in time is used to calculate the trim angle of attack. A sample work sheet that shows a convenient technique for calculating trim angle of attack is shown in Fig. 7-9.

An alternate technique for computing angle of attack is based on the rate gyro data. For small angles of attack:

$$\alpha = \frac{r + \dot{\beta}}{p} \tag{7-12}$$

$$\beta = \frac{q - \dot{\alpha}}{p} \tag{7-13}$$

$$\overline{\alpha} = \sqrt{\alpha^2 + \beta^2} . \qquad (7-14)$$

For the times of interest, $\dot{\alpha}$ and $\dot{\beta}$ are frequently negligible so that α and β may be obtained directly from rate gyro data. The significant advantage of this technique – that neither the Mach number history nor the

- 347 -

aerodynamic coefficients must be known — is obvious. The drawback is the required accuracy of the rate gyro data. Rate gyros have a tendency to drift so that the rates indicated may be somewhat in error. At low angles of attack in particular this bias may introduce a considerable error into results obtained with the above equation.

The trim maneuver plane (i. e., the plane of the transverse force acting on the body) is simply calculated from:

$$\psi = \tan^{-1} \frac{g_Z}{g_Y} \quad . \tag{7-15}$$

The aerodynamic frequency is computed from Eq. (4, 4-36):

$$\omega_{A} = \sqrt{\frac{C_{N_{\alpha}} \Delta X \overline{q} S}{I_{Y}}}$$
 rad/sec. (7-16)

Note that $C_{N_{\alpha}}$ must be in "per radian" units. We may rewrite this expression in terms of the flight data by substituting for \overline{qS} from Eq. (7-10):

$$\omega_{A} = \sqrt{\frac{-C_{N_{\alpha}}}{C_{A}}} \frac{\Delta X g_{X}}{I_{Y}} w \qquad (7-17)$$

Flight values of g_X and the established values for the aerodynamic coefficients permit calculation of the aerodynamic frequency. A sample work sheet which describes a convenient format for calculating ω_A is shown in Fig. 7-10,

Depending upon the closeness of the aerodynamic frequency to the roll rate, the trim angle of attack may be amplified to a value different from the result that would be attained by a nonrolling body, as has been discussed in

- 348 -

THE JOHNS HOPKING UNIVERSITY APPLIED PHYSICS LABORATORY BILVER BPRING MARYLAND

> Section 4.5. For some purposes in reentry flight analysis, it is convenient to have a knowledge of the amplification factor. A sample work sheet that indicates the data necessary to calculate the amplification factor is shown in Fig. 7-11. The ratio of roll rate to aerodynamic frequency also affects the direction of the trim transverse force. A technique for calculating the change in direction of this force $\Delta \psi$ is described in Fig. 7-12.

7.5 Data Analysis

A convenient technique for demonstrating the closeness of the reentry body to resonance is to plot both the roll rate and the aerodynamic frequency as a function of time on the same graph. As discussed in Section 4.5, when the roll rate is close to the aerodynamic frequency, the body is near resonance. Near resonance the previously mentioned amplification factors may be large, and large values of trim angle of attack may develop. It is possible that the large transverse forces associated with resonance (in combination with a center of pressure/center of gravity offset) may cause the roll rate to follow the aerodynamic frequency. This effect, known as "roll lock-in," may be easily detected from the plot of roll rate and aerodynamic frequency versus time.

Roll lock-in is usually considered to be an undesirable phenomenon because of the large lateral loads, localized heating, and high roll rate associated with it (which may be sufficient to cause structural damage and which will cause some impact dispersion due to the abnormally high drag encountered).

Of course, the roll rate may vary so that the rate reaches zero (a condition that usually leads to large impact dispersion), and the aforementioned plot of roll rate versus aerodynamic frequency displays this condition also.

A technique for estimating the dispersion that results when a reentry body spins through zero roll rate

- 349 -

an and a fair in the statement of the statement of the statement of a sector of the statement
THE JOHNS HOPKING UNIVERSITY APPLIED PHYSICS LABORATORY Silver Spring Maryland

1.011

is described in Ref. 7-1. The semimajor axis of the dispersion ellipse can be estimated from:

$$a = \frac{59.1}{\sin^2 \gamma_0} \left(\frac{h_o}{V_o}\right) \frac{\overline{g}_o}{\sqrt{-\dot{p}_o}} \quad \text{feet} \quad (7-18)$$

and the semiminor axis is:

$$b = -a \sin \gamma$$
 feet.

If a simulation for the basic trajectory is available, γ_0 , h_0 , and V_0 are obtained for the time when the flight record indicates zero roll rate. (However, it usually is not necessary to simulate the roll or transverse acceleration behavior in this simulation.) Values of g_0 and \dot{p}_0 are obtained directly from the telemetry records. If a simulation is not available, γ_0 , h_0 and V_0 may be estimated as follows:

1. Assume $\gamma_{o} = \gamma_{E}$.

2. Compute $K_0 = \frac{-P_0}{\beta \sin \gamma_0}$ at the time of p = 0

using Eq. (4, 3-22) and the g_X telemetry trace.

3. For $\gamma_0 = \gamma_E$ and $\beta = \frac{W}{C_A S}$, compute P_0 and thus h_0 .

4. Compute V_0 using Eq. (4.3-5).

THE JOHNE HOPKING UNIVERSITY APPLIED PHYSIC3 LABORATORY BILVER SPRING MARYLAND

REFERENCE

7-1

Ì

「市民におき」・別には、市民に

B. F. Fuess, "Impact Dispersion Due to Mass and Aerodynamic Asymmetries," <u>J. Spacecraft and Rockets</u>, Vol. 2, No. 10, October 1967, pp. 1402-1403.

ing the second other printers

THE JOHNS HOPKING UNIVERSITY APPLIED PHYSICS LABORATORY BILVER EPRING MARYLAND

「「「「「「「「「「「」」」」」

4 1 . 4.

1

. . .



THE NUMBER OF COUNTS (BITS) INCREASES FROM ZERO TO 50.

Fig. 7-1 BIT PRESENTATION SCHEME, DITAP FLIGHT RECORDS

a state of the sta

. Inda go Brow to a Ca

Preceding page blank - 353 -

THE JOHNS HOPKING UNIVERSITY APPLIED PHYSICS LABORATORY BILVER BPRING MANYLAND



THE TIMING TRACES HAVE THE POTENTIAL VALUES INDICATED BY THE CALIBRATION ON THE LEFT. A MARK INDICATES A "ONE" AND A BLANK INDICATES A "ZERO". TO READ TIME, THE "ONE" PULSES ARE SUMMED, AND THE GREENWICH MEAN TIME IS THEN 1 SECOND PLUS THE SUM SEE EXAMPLE ABOVE.

ı

.

TIMING TRACES, DITAP FLIGHT RECORDS Fig. 7-2 e an an an an ann ann ann an an an an ann. Ann an aidd seoraid dearantaidean saobh air an air ann an an 18 Bheann an an an an air air air air air air air

٩

;

.

.,

•

1. M. A.

v

.

THE JOING NOPHING UNIVERSITY APPLIED PHYSICS LABORATORY Silver Spring Maryland

l

1

i



Fig. 7.3 BIT PRESENTATION SCHEME, DITAR FLIGHT RECORDS

THE JOHNS HOPKINS LINIVERSITY APELIED PHYSICS LABORATORY BILVER SPRING MARYLAND



Fig. 7.4 TIMING TRACES, DITAR FLIGHT RECORDS

.

.

1 1 1

1

ι

ţ

·. , 21.



з,

1

ł





Contraction of the providence

the second provide the second provide the

Specific martine without the

....

.

THE JOHNS HOPKINS UNIVERSITY APPLIED PHYSICS LABORATORY SILVER SPRING MARYLAND

---- ENVELOPE TIME(seconds) (9) NOITARBJEDOA JAIRETAJ

SAMPLE LATERAL ACCELEFATION HISTORY

Fig. 7-6

2012

5

ĩ

 ∂f

6.0

4

- 358 -

THE JOHNS MOPRINS UNIVERSITY APPLIED PHYSICS LABORATORY SILVER SPRING MARYLAND

ł,

Į,





State of the state

NESS STR

.

- 359 -

THE JOHNS MOPKING UNIVERGITY APPLIED PHYSICS LABORATORY BILVER SPRING MARYLAND



- 360 -

Contraction the second second

4

THE JOHNE HOPAINE UNIVERSITY APPLIED PHYSICS LABORATORY BILVER SPRING MATLAND

· 4.4.1 . 1

يديد ور

11.1

シンションである

LANN MARK WAL



$$\bar{a} = \frac{\sqrt{B_{\gamma}^{2} + g_{Z}^{2}}}{-g_{\chi}} \frac{C_{A}}{C_{N_{d}}} \text{ (degrees)}$$

Fig. 7-9 ANGLE OF ATTACK WORK SHEET

- 361 -

. .

the second second second second second second second second second second second second second second second se

THE JOHNS HOPKINS UNIVERSITY APPLIED PHYSICS LABORATORY SILVER EPRING MARYLAND

14	12 × 13 ^ω _A	AERODYNA
13	к′	$K' = 57.3\sqrt{\frac{57.3W}{12 l_Y}}$
12	$\sqrt{0}$	
1	10 × 7	
10	B - B	
9	× _{c,g,}	CENTER OF GRAVITY OBTAINED FROM PREFLIGHT DATA (INCHES)
8	Х _{е,р,}	CENTER OF PRESSURE OBTAINED FROM LISTING OF AERODYNAMIC CHARACTERISTICS (INCHES)
$\textcircled{\baselinetic}$	6×3	
働	4 ÷ 5	
(5)	CA	AXIAL FORCE COEFFICIENT OBTAINED FROM LISTING OF AERODYNAMIC CHARACTERISTICS (DIMENSIONLESS)
٩	C _{N(1}	NORMAL FORCE COEFFICIENT SLOPE OBTAINED FROM LISTING OF AERODYNAMIC CHARACTERISTICS (PER DEGREE)
3	- a ^X	FLIGHT TEST. LONGITUDINAL DECELERATION (g)
2	MACH NUMBER	OBTAINED FROM COMPUTER SIMULATION. USE TIME OF PEAK LONGITUDINAL DECELERATION TO MATCH FLIGHT DATA TO SIMULATION (DIMENSIONLESS)
0	TIME BEFORE IMPACT	

 $\omega_{A} = 57.3 \sqrt{\frac{W}{L_{y}}} = \frac{57.3}{12} \sqrt{\frac{C_{N_{L}}}{C_{A}}} = (X_{c,p} - X_{c,p})(-\theta_{X})$

Fig. 7-10 AERODYNAMIC PITCH FREQUENCY (WEATHERCOCK FREQUENCY) WORK SHEET

- 362 -

the formation of the

2.88 Mar 2012 8 14

THE JOHNS HOPKINS UNIVERSITY APPLIED PHYSICS LABORATORY NILVER BPRING MARYLAND

2		AMPLIFICATION FACTOR
0	18+19	
19	1) ² (8)	
18	[1 · (B)] ²	
17		
16	13 - 18	
15	$\textcircled{1} \frac{d^2}{i_Y}$	
1	C _{imq}	PITCH DAMPING COEFFICIENT OBTAINED FROM LISTING OF AERODYNAMIC CHARACTERISTICS (DIMENSIONLESS)
13	$\underbrace{\textcircled{12}}_{m}(1-\frac{1}{1}\frac{x}{y})$	
12	с _{N и}	NOTIMAL FORCE COEFFICIENT SLOPE OBTAINED FROM LISTING OF AERODYNAMIC CHARACTERISTICS (PER RADIAN)
0	1 <u>0</u> ≈ 30	
1	¢ x	LONGITUDINAL DECELERATION FROM FLIGHT DATA (g)
9	C A	AXIAL FORCE COEFFICIENT OBTAINED FROM LISTING OF AERODYNAMIC CHARACTERISTICS (DIMENSIONLESS)
٩	$\lambda^2 = \frac{5}{7}$	
0	²	
6	AERODYNAMIC FREQUENCY	OBTAINED FROM PREVIOUS CALCULATION
(6)	$\textcircled{1}^2 (1 - \frac{i_X}{i_Y})$	IX (FOLL INERTIA) AND IX (PITCH INERTIA) OBTAINED FROM PREFLIGHT DATA (degrees/second)
٩	ROLL NATE	FLIGHT DATA (degrees/second)
3	VELOCITY	OBTAINED FROM COMPUTER SIMULATION (feet/second)
2	MACH NUMBER	OBTAINED FROM COMPUTER SIMULATION, TIME OF PEAK LONGITUDINAL DECELERATION TO MATCH FLIGHT DATA TO SIMULATION (DIMENSIONLESS)
0	TIME BEFORE IMPACT	

 $A = \frac{1}{\sqrt{(1-\lambda^2)^2 + p^2 \lambda^2}} \qquad \lambda^2 \approx \frac{p^2}{\omega_A^2} (1-\frac{l_X}{l_Y}) \qquad D = \frac{(-g_X)W}{VC_A} \qquad \frac{\left[\frac{C_{N_d}}{m}(1-\frac{l_X}{l_Y}) - \frac{C_m q d^2}{l_Y}\right]}{\omega_A \sqrt{1-\frac{l_X}{l_Y}}}$

Constraint of the local distance

1114

Fig. 7-11 AMPLIFICATION FACTOR WORK SHEET

- 363 -

THE JONNS HOPKING UNIVERSITY APPLIED PHYSICS LABORATORY SILVER SPRING MARYLAND

Ī	۸Ψ	SET SIGN OF ${\scriptstyle\Delta}\psi$ to be opposite to that of the holl hate
(6)	[∆¥]	= SIN-* (B)
\$	SIN AV	· (2)3)4
٩	A	- (COL. (1) - FIG. 7 11)
3	λ	= (VCOL. (8) FIG. 7-11)
2	D	- (CUL 17) FIG. 7-11)
0	TIME BEFORE IMPACT	

 $\sin \Delta \psi = 1 \pm D \lambda A$

Fig. 7-12 ROTATION OF MANEUVER PLANE WOPK SHEET

- 364 -

.

. ...

Section Deside & State Section States

THE JOHNS HOPKING UNIVERSITY APPLIED PHYSICS LABORATORY BILVER SPRING, MARYLAND

TUT

F

「生産」というない。 Ţ

1

8. CONSTANTS AND CONVERSION

BOULWIDE

EQUATIONS

365 -

...

a and a subscription of the second statement

Conversion Equations

Degrees Centigrade = $\frac{5}{9}$ (degrees Fahrenheit - 32) Degrees Rankine = degrees Fahrenheit + 460 Degrees Kelvin = degrees Centigrade + 273 Time (in hours) at zero meridian = local time (in hours) + $\frac{local longitude}{15}$ (in degrees)

where

West longitude is positive East longitude is negative

Conversion Constants

- 1 radinn = 57, 3°
- 1 meter = 3.28 feet
- 1 nmi = 6080 feet
- 1 gram/cubic centimeter = 1,94 slugs/ft³
- 1 kilogram/cubic meter = 1.94×10^{-3} slugs/ft³
- 1 millibar = 2,09 lb/ft^2
- 1 knot = 1,69 ft/sec

Proceding page blank

- 367 -

Constants

Acceleration of gravity at sea level = $g = 32.17 \text{ ft/sec}^2$ Ratio of specific heats = $\gamma = 1.4$ Gas constant for dry air = $1736 \text{ ft}^2/\text{sec}^2$ - degrees Rankine Gas constant for water vapor = $2759 \text{ ft}^2/\text{sec}^2$ - degrees Rankine Radius of earth = $R_e = 3441 \text{ nmi} = 2.09 \times 10^7 \text{ feet}$ Rotational velocity of earth = $\omega_e = 0.729 \times 10^{-4} \text{ rad/sec}$ Napierian base = e = 2.718

 $\pi = 3,142$

વર પ્રોપ્ટ અંદ

· . 🍋 .

With the Partainer +

THE JOHNS HOPKING UNIVERSITY APPLIED PHYSICS LABORATORY BILVER SPRING, MARYLAND

I

мун Мин Мин

1

.

I

. . .

i.

.

Ъ. н

INDEX

.....

HIDEX

「「「

÷

¦∧ ₁

1

ł

- ----

*

THE JOHNS HOPKING UNIVERSITY APPLIED PHYSICS LABORATORY BILVER SPRING MARYLAND

1

Α

Acceleration, Lateral	
α convergence – nonrolling missile	152
- rolling missile	154
Aerodynamic Coefficients	
Asymmetric moment for flap on a cone	28
Asymmetric moment for a distorted cone	2 9
Definitions	16,25
Equations for a cone at high Mach	
numbers	23
Aerodynamic Forces	
Asymmetric	24
Axial	13
Drag	21
Lift	21
Normal	13
Aerodynamic Moments	
Pitch	25
Pitch damping	15
Roll damping	16
Aerodynamic Pitch Frequency	147
Altitude	
Geometric	53
Geopotential	53
Amplification Factor	
Angle of attack	171
Roll acceleration	181

Preceding page blank

- 371 -

Same Werner Hong Stranger

À

Angle of Attack Amplification factor Convergence – nonrolling vehicle – rolling vehicle Definition Trim	171 148 151 13,155 26
Angular Momentum, Conservation of	81
Apogee	86
Asymmetry Center of gravity or center of pressure offset Effect on roll rute In plane Moment (couple) Moments of inertia, unequal Out of plane Products of inertia Types Atmosphere Exponential Polar Standard Tropical Standard 1962 Standard	174 180 177 168 217 177 207 167 55 66 66 66 68, 69
Axial Force	13
Azimuth, Inertial	103
В	
Ballistic Coefficient	117
Ballistic Density	285
Ballistic Wind	283
Barosphere	49

- 372 -

1

÷

1 1 1

glad for the detail of the ford day

Å

THE JOHNS HOPKINS UNIVERSITY APPLIED PHYSICS LABORATORY SILVER SPRING MARYLAND

ť

Ċ,

 $A_{\rm e}^2 \sim - \Lambda^{-1}$

1

Boaj' Rate	
Effect of product of inertia in vacuum	208
Transverse	140
Effect of unequal pitch and yaw moment	
of inertia	217
Bound ar y Layer	18
С	
Center of Pressure	13
Cone	
Aerodynamic coefficients	23, 24, 29
Space	140
	0.0 5
Constants and Conversion Equations	367
Continuum Flow	20 280
Continuum Flow	20, 200
Coriolis Force	59
	- •
D	
Damping	
Pitch	15, 24
Roll	16, 24
Deceleration, Reentry History	119, 120
Density	
Ballistic	285
Polar Standard Atmognhere	66
Range nartial for	285
Tropical Standard Atmosphere	66
Weighting factor for	285
1962 Standard Atmosphere	68, 69
The second	

BRAN HINE BALLER

1

ł

- 373 -

. . .

Dispersion	
Axial force coefficient errors	286
Density deviation	285
Missile weight error	286
Trim angle of attack — rolling missile	288
– nonrolling missile	28 9
Wind	283
Drag	21
Dynamic Pressure	
Definition	17
Reentry history	119
Dynamic Stability	146
\mathbf{E}	
Energy Equation, Conservation of	81
Equations of Motion Six-degree-of-freedom	
- symmetrical missile	137
- asymmetrical missile	168
Exosphere	49
F	
Flight Path Angle	81
Flight Path Angle, Inertial	103
Flight Test Analysis	341
Elight Time	
Ballistic trajectory in vacuum	84
Reentry history	116
	· · ·
Free-Molecular Flow	20, 260

199

China the second water and the second second

ų V

i 1

The second second second second second second second second second second second second second second second s

* : :

••

The base the glood start and a second

te:

- 374 -

المراجعة المعام والمتحافظ والوور والمطاطع في

THE JOHNE HOPKING UNIVERSITY APPLIED PHYSICS LABORATORY BILVER SPRING MARYLAND

;

1.

3

1

119

1

÷

ł,

1

í

ł

Frequency	
Aerodynamic or weathercock	147
Body pitch — in vacuum	1 39
G	
Gravity	
Effect on ballistic trajectory	81
Law of	81
Variation with altitude	68
н	
Heat	
Absorbed at stagnation point	119
Rate at stagnation point	119, 120
Homosphere	50
Humidity, Effect on Density	56, 67
Hydrostatic Equilibrium Equation	51
I	
Trontin	
Moment of	217
Product of	207
In-Plane Asymmetry	177
Isobar	59
K	
Knudsen Number	19
L	
Lateral Acceleration	152, 154

and the second second second second second second second second second second second second second second second

and a state of the

and the local data

「「「「「「」」

は、「「「「「「」」」」」」

Ŋ

御上にはないの しょうちん 御代の 日本

;

. .

a and the second second second second second second second second second second second second second second se

. ...

- 375 -

٠..

......

THE JOHNE HOPKINE UNIVERSITY APPLIED PHYSICS LABORATORY SILVER SPRING MARYLAND

> Lock In Conditions required for 188 Definition 186 198 Effect of roll damping Effect of roll torque 198 Longitudinal Deceleration 119, 120 M Mach Number 17 Mean Free Path, 1962 Standard 68 Atmosphere Mesopause 65 65 Mesosphere 83 Minimum Energy Trajectory Moisture, Effect on Density 56, 67 Moment Asymmetric 28, 29 Pitch 25 15 Pitch damping Roll damping 16 217 Moment of Inertia, Unequal 81 Momentum, Conservation of Angular N Normal Force 13 Ρ 52 Perfect Gas Law

> > - 376 -

and a star and a star and a star a

THE JOHNE HOPKING UNIVERSITY APPLIED PHYSICS LABORATORY BILVER SPRING, MARYLAND

I

Pitch Frequency, Aerodynamic	147	
Polar Standard Day	66	
Pressure, Dynamic	17	
Pressure Force, Horizontal	59	
Pressure, 1962 Standard Day	68 215 207 208 211	
Product of Inertia Criterion for lock in Definition Effect on body rate history, vacuum Effect on trim angle of attack		
R		
Radiosonde Observation (RAOB)	55	
Range Partial Density Wind	285 283	
Range Sensitivity Factors	87, 93, 9	
Reentry Dynamic pressure Heat absorbed Heating rate Longitudinal deceleration Reynolds number Time Velocity	119, 120 119 119, 120 119, 120 119, 120 119, 120 118 118	
Resonance, Roll-Yaw Definition Effect on roll acceleration Effect on trim angle of attack Effect on windward meridian First and second	171 180 171 173 187	

94

19

ないのないであ

THE JOHNS HOPKINS UNIVERSITY APPLIED PHYSICS LABORATORY SILVER SPRING MARYLAND

> Reynolds Number 18 Definition 119, 120 Reentry history 56 Rocket Observation (ROCOB) 16 **Roll Damping Roll Rate** 178 Effect of aerodynamic asymmetry 175 Effect of center-of-gravity offset 213 Effect of product of inertia 181 Effect of resonance Effect of unequal moments of inertia 219 181 Equilibrium S Scale Height 55 Definition 68 1962 Standard Atmosphere 187 Second Resonance 327 Simulation 11, 260 Slip Flow 140 Space Cone Speed of Sound 51 Definition 66 **Polar** Atmosphere 66 Tropical Atmosphere 68 1962 Standard Atmosphere 187 Spin Up, Spin Down Standard Day 6**6** Polar 62 Tropical 68, 69 1962

> > - 378 -

.

3.42.5 T

1

「「「」」、「」」」、「」」、「」、「」、

÷, ₩

1

ľ

. ŀ

ľ.

ł

THE JOHNS HOPKING UNIVERSITY APPLIED PHYSICS LABORATORY BILVER SPRING MARYLAND

Ì.

2

1146 1

		1.4
Static	Margin	1 **

Static Stability 14

Stratophuse 65 Stratosphere 65

Sutherland's Viscosity Equation

Т

35

140

66

65

65

۰,

and the second of the stand of a second second and the second

Temperature Polar Standard Tropical Standard 1962 Standard	66 66 68, 69
Thermopause	6 5
Thermosphere	65
Time History, Reentry	118

Trajectory	
Minimum Energy	83
	327
Simulation	

Transvorse Body Rate Tropical Day Tropopause

Troposphere

V

Velocity, Missile	
Inertial	103
Minimum Energy Trajectory	B3, 90
	118
Reentry history	

- 379 -

. .

W. Caldinson

ar 1

THE JOHNE HOPKINE UNIVERSITY APPLIED PHYSICS LABORATORY SILVER SPRINE MARYLAND

Viscosity	
Definition	18
Sutherland's equation	35
1962 Standard Atmosphere	68

W

and Training

i i

「日本のない」の「日本のない」」

'N

10 40 - CAR

1

「「「「「」」のの方で

726

.....

į

.

ł

i ...

. |

and the same & Parts Station of a 199

ľ

. Anter

Wind	
Ballistic	283
Geostrophic	6 0
Range partial for	283
Weighting factor for	282